UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT Pursuant to Section 13 OR 15(d) of The Securities Exchange Act of 1934

Date of Report (Date of earliest event reported) December 20, 2019

TRIMAS CORPORATION

(Exact name of registrant as specified in its charter)

001-10716

38-2687639

Delaware

(State or other jurisdiction	n	(Commission	(IRS I	Employer
of incorporation)		File Number)	Identific	cation No.)
20505 147-5-4	S	Dl	40204	
38505 Woodward Avenue, Mi	Suite 200, chigan	Bloomfield Hills,	48304	
(Address of princ	· ·	fices)	(Zip Code)	
	Regis	trant's telephone number, including area	code (248) 631-5450	
	(Fo	Not Applicable ormer name or former address, if changed	d since last report.)	
Check the appropriate box below if provisions (see General Instruction		iling is intended to simultaneously satisfy	y the filing obligation of the registrant ur	nder any of the following
☐ Written communications pursu	ant to Rule 425	under the Securities Act (17 CFR 230.42	25)	
☐ Soliciting material pursuant to	Rule 14a-12 un	der the Exchange Act (17 CFR 240.14a-	12)	
☐ Pre-commencement communi	cations pursuant	to Rule 14d-2(b) under the Exchange A	ct (17 CFR 240.14d-2(b))	
☐ Pre-commencement communi	cations pursuant	to Rule 13e-4(c) under the Exchange Ad	et (17 CFR 240.13e-4(c))	
Securities registered pursuant to Se	ction 12(b) of th	e Act:		
Title of each	class	<u>Trading symbol(s)</u>	Name of exchange on w	hich registered
Common stock, \$0.	01 par value	TRS	The NASDAQ Global	Market LLC
ndicate by check mark whether the Rule 12b-2 of the Securities Excha		emerging growth company as defined in (17 CFR §240.12b-2).	Rule 405 of the Securities Act of 1933	(17 CFR §230.405) or
			Emerging growth company	
		mark if the registrant has elected not to rsuant to Section 13(a) of the Exchange		omplying with any new or

Item 2.01 Completion of Acquisition or Disposition of Assets.

On December 20, 2019, TriMas Company LLC, a Delaware limited liability company and wholly-owned subsidiary of TriMas Corporation (the "Company"), completed the previously announced sale of its Lamons division (the "Lamons Division") to two wholly-owned subsidiaries of an investment fund sponsored by First Reserve, LGC US Holdings, LLC, a Delaware limited liability company, and LGC US Finco, LLC, a Delaware limited liability company, pursuant to an Asset and Stock Purchase Agreement dated as of November 1, 2019 (the "Purchase Agreement"), for a purchase price of \$135 million, subject to certain adjustments as set forth in the Purchase Agreement. The foregoing description of the Purchase Agreement does not purport to be complete, and is qualified in its entirety by reference to the full text of the Purchase Agreement, a copy of which is filed hereto as Exhibit 2.1 and is incorporated by reference herein.

Item 7.01 Regulation FD.

On December 20, 2019, the Company issued a press release announcing the completion of the sale of the Lamons Division. A copy of the press release is attached hereto as Exhibit 99.1 and is incorporated herein by reference. The press release is also available on the Company's website at www.trimascorp.com.

The information furnished pursuant to Item 7.01, including Exhibit 99.1, shall not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934 (the "Exchange Act") or otherwise subject to the liabilities under that Section and shall not be deemed to be incorporated by reference into any filing of the Company under the Securities Act of 1933 or the Exchange Act.

Item 9.01 Financial Statements and Exhibits.

(b) Pro Forma Financial Information.

The unaudited pro forma consolidated balance sheet of the Company as of September 30, 2019, the unaudited pro forma consolidated statements of income of the Company for the nine months ended September 30, 2019 and for the years ended December 31, 2018 and 2017 and the unaudited pro forma consolidated statement of operations of the Company for the year ended December 31, 2016 giving pro forma effect to the disposition of the Lamons Division are included as Exhibit 99.2 to this Current Report on Form 8-K and are incorporated into this Item 9.01 by reference.

(d) Exhibits. The following exhibits are filed or furnished, as applicable, herewith:

Exhibit No.	Description
2.1*	Asset and Stock Purchase Agreement, dated as of November 1, 2019, by and between TriMas Company LLC and LGC US Holdings, LLC and LGC US Finco, LLC.
99.1	Press Release, dated December 20, 2019.
99.2	Unaudited Pro Forma Consolidated Financial Information of TriMas Corporation.
104	Cover Page Interactive File (embedded within the Inline XBRL document).

^{*} Certain exhibits and schedules have been omitted and the Company agrees to furnish supplementally to the Securities and Exchange Commission a copy of any omitted exhibits and schedules upon request.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

TRIMAS CORPORATION

Date: December 20, 2019 By: /s/ Joshua A. Sherbin

Name: Joshua A. Sherbin

Title: Senior Vice President, General Counsel and Secretary

ASSET AND STOCK PURCHASE AGREEMENT

AMONG

TRIMAS COMPANY LLC

AND

LGC US HOLDINGS, LLC AND LGC US FINCO, LLC

DATED

NOVEMBER 1, 2019

TABLE OF CONTENTS

	<u>Page</u>		
ARTICLE I DEFINITIONS			
Tormo 2	1.1Certain Defined		
Terms 12	1.20ther Defined		
Terms 13 Matters 16	1.3Certain Interpretive		
Matters 16			
ARTICLE II PURCHASE AND SALE			
the Sold Assets 18	2.1Purchase and Sale of		
the Shares 19	2.2Purchase and Sale of		
Liabilities; Excluded Liabilities 21	2.3Excluded Assets 19 2.4Assumption of		
	2.5Purchase Price 22 2.6Purchase Price		
Adjustment 22	2.7Allocation of Total		
Consideration 26	2.8The Closing 27 2.9Deliveries at the		
Closing. 27	2.10Further		
Assurances 28	2.11Withholding 29		
ARTICLE III REPRESENTATIONS AND WARRANTIES OF THE COMPANY			
	3.1Organization 29 3.2Authorization;		
Enforceability 30	3.3Capital Stock of the		
Divested Companies 30	3.4Financial Information;		
No Undisclosed Liabilities; Solvency. 30	3.5Sufficiency of the		
Assets 31	3.6No Approvals or		
Conflicts 32	3.7Compliance with Law;		
Permits 32	3.8Proceedings 33 3.9Absence of Certain		
Changes 33	3.10Tax Matters 33 3.11Employee		
Benefits 36	3.12Active Employees;		
Labor Relations 37	3.13Intellectual		
Property 38	2.14Contracts 40		

3.14Contracts. 40

Matters. 42

Assets. 43

3.15Environmental

3.16Insurance 433.17Personal Property

3.18Real Property. 43

i

TABLE OF CONTENTS (continued)

<u>Page</u>

5.5Tax Indemnity 61

	Suppliers. 44	3.19Customers and
	Anti-Corruption Laws 45	3.20Compliance with
	Economic Sanctions. 45	3.21Export Controls and
		3.22Bankruptcy 46 3.23Accounts
	Receivable 46	3.24Inventory 46 3.25Operating
	Areas 47	3.26Affiliate
	Transactions 47	3.27No Brokers' or Other
	Fees 47	3.28Transaction
	Expenses 47	3.29Credit Support
١	Obligations 47 Shutdown Costs 47	
	Representations or Warranties 47	3.31No Other
	ARTICLE IV REPRESENTATIONS AND WARRANTIES OF THE BUYERS	
		4.1Organization 48 4.2Authorization;
	Enforceability 48	4.3No Approvals or
	Conflicts 49	4.4Proceedings 49 4.5Compliance with
Laws	Laws; Permits 49	4.6Financing 50 4.7Solvency 51 4.8Limited
	Guaranty 51	4.9No Brokers' or Other
	Fees 51	4.10Condition of the
	Business 52	4.11No Other
	Representations or Warranties 53	
ARTICLE V COVENANTS AND AGREEMENTS		
	Prior to the Closing 53	5.1Conduct of Business
	of Books and Records 56	5.2Access; Preservation
		5.3Regulatory Filings;
	Cooperation; Preparation of Returns; Tax Elections 60	5.4Tax Matters;
	Soperation, Freparation of Notaline, Tax Electronic 00	E ETay Indomnity 61

30

Employment Matters 65

Customers and Suppliers 72

Non-Solicitation and Confidentiality 72

ii

to Indemnity of Tax Claims 63

Laws 75

Treatment 64

Actions 64

5.6Procedures Relating

5.7Refunds and Tax

5.8Post-Closing Tax

5.9Employees;

5.10Labor Matters 715.11Contact With

5.12Non-Competition;

5.13Further Actions 745.14Bulk Transfer

TABLE OF CONTENTS (continued)

<u>Page</u>

	5.15Confidentiality 75 5.16Exclusivity 77		
Closing 77	5.17Names Following		
Certain Matters 78	5.18Notification of		
Obligations 78	5.19Intercompany		
Conduct 78	5.20Post-Closing		
	5.21Financing 79 5.22R&W Insurance		
Policy 84	5.23Books and		
Records 85	5.24Releases 85 5.25Seller		
Guarantee 86	5.26Wrong Pocket 87 5.27Release of Credit		
Support Obligations 88	5.28Certain Matters 88		
ARTICLE VI CONDITIONS TO THE COMPANY'S OBLIGATIONS			
	6.1Representations and		
Warranties 89	6.2Performance 89 6.3Officer's		
Certificates 89	6.4Competition/Foreign		
Investment Laws 89	6.5Governmental		
Orders 89	6.6Deliveries 89		
ARTICLE VII CONDITIONS TO THE BUYERS' OBLIGATIONS			
	7.1Representations and		
Warranties 89	7.2Performance 90 7.3Officer's		
Certificate 90	7.4Competition/Foreign		
Investment Law 90	7.5Governmental		
Orders 90	7.6Required		
Consents 90	7.7Deliveries 90		
ARTICLE VIII			

ARTICLE IX INDEMNIFICATION

iii

TABLE OF CONTENTS (continued)

<u>Page</u>

Commons, 02	9.1Indemnification by the
Company 92	9.2Indemnification by the
Buyers 93	9.3Exclusive
Remedy 94	9.4Indemnification
Calculations 94	9.5Survival 95 9.6Notice and
Opportunity to Defend 96	9.7Additional
Limitations 97	9.8Subrogation 98
ARTICLE X MISCELLANEOUS	
F. manage	10.1Fees and
Expenses 98	10.2Governing Law 98 10.3Projections 99 10.4Amendment 99 10.5No Assignment 99 10.6Waiver 100 10.7Notices 100 10.8Complete
Agreement 101	10.9Counterparts 101 10.10Publicity 102 10.11Severability 102 10.12Third Parties 102 10.13Non-Recourse 102 10.14Dispute
Resolution 103	10.15Specific
Performance 103	· ·
Trial 104	10.16Waiver of Jury
Privilege and Conflict Waiver 104	10.17Attorney-Client
	10.18Frustration of

Closing Conditions 105

EXHIBITS

Exhibit A – Form of Assignment and Assumption of Leases

Exhibit B – Form of Bill of Sale and Assignment and Assumption Agreement

Exhibit C – Binder Agreement for R&W Insurance Policy

Exhibit D – Form of Transition Services Agreement

Exhibit E – Working Capital Example

Exhibit F – Closing Statement Example

ANNEXES

Annex A - Company Deliverables

Annex B - Buyer Deliverables

ASSET AND STOCK PURCHASE AGREEMENT

THIS ASSET AND STOCK PURCHASE AGREEMENT (this "Agreement"), dated November 1, 2019 (the "Execution Date"), is entered into by and among TRIMAS COMPANY LLC, a Delaware limited liability company (the "Company"), LGC US Holdings, LLC, a Delaware limited liability company (the "Asset Buyer"), and LGC US Finco, LLC, a Delaware limited liability company (the "Equity Buyer"), and solely with respect to Section 5.25, TRIMAS CORPORATION, a Delaware corporation ("Seller Guarantor"). The Asset Buyer and the Equity Buyer are each a "Buyer" and collectively, the "Buyers." The Company and the Buyers are each a "Party" and collectively, the "Parties."

RECITALS

WHEREAS, Rieke-Lamons Nederland Holdings B.V., a private company with limited liability (besloten vennootschap met beperkte aansprakelijkheid) incorporated under the laws of the Netherlands and an indirect wholly owned subsidiary of the Company ("R-L Netherlands Holdings"), and Rieke-Lamons International Holdings LLC, a Delaware limited liability company and an indirect wholly owned subsidiary of the Company ("R-L International Holdings"), collectively own all of the issued and outstanding Equity Interests of Lamons Belgium BVBA, a private company with limited liability (besloten vennootschap met beperkte aansprakelijkheid) incorporated under the laws of Belgium ("Lamons Belgium" and such Equity Interests, the "Lamons Belgium Shares");

WHEREAS, R-L Netherlands Holdings, R-L International Holdings and Rieke-Lamons UK Holdings B.V., a private company with limited liability (*besloten vennootschap met beperkte aansprakelijkheid*) incorporated under the laws of the Netherlands and an indirect wholly owned subsidiary of the Company ("R-L UK Holdings"), collectively own all of the issued and outstanding Equity Interests of Lamons (Thailand) Co. Ltd., a private limited company incorporated under the laws of Thailand ("Lamons Thailand" and such Equity Interests, the "Lamons Thailand Shares");

WHEREAS, R-L Netherlands Holdings owns all of the issued and outstanding Equity Interests of (a) Lamons Singapore Pte. Ltd., a private limited company incorporated under the laws of Singapore ("Lamons Singapore" and such Equity Interests, the "Lamons Singapore Shares") and (b) Lamons Canada Limited, a corporation incorporated under the laws of Canada ("Lamons Canada" and such Equity Interests, the "Lamons Canada Shares");

WHEREAS, Lamons Nederland B.V., a private company with limited liability (besloten vennootschap met beperkte aansprakelijkheid) incorporated under the laws of the Netherlands and an indirect wholly owned subsidiary of the Company ("Lamons Netherlands" and together with R-L Netherlands Holdings, R-L International Holdings and R-L UK Holdings, the "Equity Sellers") owns all of the issued and outstanding Equity Interests of Lamons Gasket and Bolt Ibérica, S.L., a company organized under the laws of Spain ("Lamons Spain" and such Equity Interests, the "Lamons Spain Shares");

WHEREAS, the Lamons Belgium Shares, the Lamons Thailand Shares, the Lamons Spain Shares, the Lamons Singapore Shares and the Lamons Canada Shares are collectively the "Shares";

WHEREAS, Lamons Belgium, Lamons Thailand, Lamons Spain, Lamons Singapore and Lamons Canada are collectively the "Divested Companies";

WHEREAS, the "Lamons" division of the Company, through the Divested Companies and Lamons Gasket Company, a Delaware corporation and a direct wholly owned subsidiary of the Company (the "Asset Seller" and together with the Equity Sellers, the "Sellers"), is engaged in the design, manufacture and distribution of industrial sealing, fastener and specialty products for the petrochemical, petroleum refining, oil field, water treatment, wastewater treatment and other industrial markets (such business as is conducted by or through the Asset Seller and the Divested Companies as presently conducted (excluding, for the avoidance of doubt, any Disposed Lamons Business), the "Business"); and

WHEREAS, the Company desires to sell the Business (by causing the Sellers to sell the Shares and the Sold Assets (as defined below) and transfer the Assumed Liabilities (as defined below) to the Buyers), and the Buyers desire to purchase the Business from the Company (by purchasing all of the Shares and the Sold Assets and assuming the Assumed Liabilities from the Sellers), in each case, upon the terms and conditions contained in this Agreement and the Ancillary Agreements, as applicable.

NOW, THEREFORE, in consideration of the foregoing and the representations, warranties, covenants and agreements herein contained, and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, and intending to be legally bound hereby, the Parties hereby agree as follows:

Article I DEFINITIONS

1.1 Certain Defined Terms. As used in this Agreement, the following terms shall have the following meanings:

"Accounting Methods" means, the accounting principles, methods and practices utilized in preparing the Financial Information. For the avoidance of doubt, the calculation of Working Capital will be calculated in accordance with terms set forth in the definition thereof and taking into account certain accounting methodologies applied, all of which are in accordance with GAAP, as presented on Schedule 1.1(a).

"Acquired Intellectual Property" means all Intellectual Property owned by the Divested Companies, and all Intellectual Property owned by the Asset Seller, and the right to sue, and collect damages for, past, present or future, infringement, dilution or other violation of the foregoing.

"Active Employee" means (a) any employee of a Divested Company, (b) those employees of the Asset Seller who are listed on the Active Employee Schedule, and (c) any employee of an Affiliate of the Asset Seller set forth on Schedule 1.1(b), including, in each case of clauses (a), (b), and (c), those employees (i) who are absent due to vacation, jury duty, military, FMLA, pregnancy, parental and bereavement leave, scheduled time off, long service leave, or illness or injury leave in compliance with the applicable written policies of the Divested Companies, the Asset Seller or their respective Affiliates or the applicable collective bargaining agreement or (ii) who are receiving workers' compensation payments as required by Law and have the right to re-employment in accordance with applicable Law or the applicable collective bargaining agreement.

"Active Employee Schedule" means that schedule set forth on Schedule 1.1(c).

"Affiliate" means, with respect to any Person, any other Person controlling, controlled by or under common control with such Person as of the date on which, or at any time during the period for which, the determination of affiliation is being made. For the avoidance of doubt, for purposes of this Agreement, (a) the Sellers shall be considered Affiliates of the Company and of each other, (b) the Divested Companies shall be considered Affiliates of the Company and the Sellers prior to the Closing and (c) the Divested Companies shall be considered Affiliates of the Buyers after the Closing. Notwithstanding the foregoing, except with respect to Section 8.2(c) and Section 10.12, no portfolio company or investment fund organized by First Reserve XIII Advisors, L.L.C., First Reserve XIV Advisors, L.L.C. or any of their respective Affiliates shall be deemed an Affiliate of any Buyers for purposes of this Agreement.

"Ancillary Agreements" means (a) the Bill of Sale and Assignment and Assumption Agreement (and such other documents or instruments contemplated therein to give effect to a transfer of the Sold Assets and Assumed Liabilities to the Asset Buyer), (b) the Assignment and Assumption of Leases, (c) the transfer Contracts contemplated by Annex A and Annex B (and such other documents or instruments contemplated therein to give effect to a transfer of the Shares to the Equity Buyer), and (d) the Transition Services Agreement.

"Anti-Corruption Laws" means the United States Foreign Corrupt Practices Act of 1977, as amended, the Canadian Corruption of Foreign Public Officials Act, the UK Bribery Act, the anti-bribery or anti-corruption of laws of the Netherlands, Singapore, Spain, and Thailand, and all other similar or equivalent anticorruption or anti-bribery laws of any jurisdiction applicable to the Company,

Asset Seller, the Divested Companies and it's or their respective officers, directors, employees, agents or representatives.

"Asset Seller Predecessor" means any predecessor (to the extent not subsequently merged, consolidated or liquidated into the Asset Seller) or Affiliate (that is not a Divested Company) of the Asset Seller.

"Asset Taxes" means ad valorem, property, excise, severance, production, sales, use and similar Taxes based upon the acquisition, operation or ownership of the Sold Assets, but excluding, for the avoidance of doubt, Income Taxes and Transfer Taxes.

"Assignment and Assumption of Leases" means the Assignment and Assumption of Leases substantially in the form of Exhibit A.

"Bill of Sale and Assignment and Assumption Agreement" means the Bill of Sale and Assignment and Assumption Agreement substantially in the form of Exhibit B.

"Business Day" means any day that is not a Saturday, a Sunday or other day on which banks are required or authorized by Law to be closed in the city of Bloomfield Hills, Michigan or the city of New York, New York.

"Business Material Adverse Effect" means any change, effect, event, occurrence, state of facts or development ("Effect") that, individually or in the aggregate, has or would reasonably be expected to have, a material adverse effect on the business, results of operations or financial or operational condition of the Sold Assets, the Divested Companies and the Business, taken as a whole, but none of the following shall be deemed either alone or in combination to constitute, and none of the following shall be taken into account in determining whether there has been a Business Material Adverse Effect: any adverse Effect resulting from (a) general economic conditions, (b) political conditions, including acts of war (whether declared or undeclared), armed hostilities and terrorism, or developments or changes therein, (c) any conditions resulting from natural disasters or other acts of God, (d) compliance by the Company with its covenants and agreements contained in this Agreement, (e) the failure of the financial or operating performance of the Business to meet internal projections or budgets for any period prior to, on or after the Execution Date (but not, for the avoidance of doubt, the underlying changes, events, facts, circumstances, conditions, developments or occurrences relating to such failure), (f) changes or developments in the business or regulatory conditions affecting the industries in which the Business operates, (g) the announcement of this Agreement or the transactions contemplated hereby, including any losses of employees or any disruption in supplier, customer, distributor and similar relationships, in each case solely to the extent specifically resulting from such announcement (h) currency exchange rates, including any fluctuations or other changes therein (i) any changes in applicable Laws or accounting rules, (j) any actions or omissions by the Company prior to the Closing that are expressly required under this Agreement or are taken or not taken with the Buyers' prior written consent (but not, for the avoidance of doubt, the underlying changes, events, facts, circumstances, conditions, developments or occurrences relating to such act or omission), in each case, except to the extent any of the changes, events, circumstances, developments or occurrences described in clauses (a), (b), (c), (f), (h), and (i) disproportionately impact the Business as compared to other participants in the industry in which the Business is operated.

"Cash" means cash and cash equivalents (determined in accordance with GAAP), but excluding restricted cash.

"Closing Working Capital" means the Working Capital as of the Closing.

"Code" means the Internal Revenue Code of 1986, as amended.

"Commercial Tax Agreement" means customary commercial agreements not primarily related to Taxes that contain agreements or arrangements relating to the apportionment, sharing, assignment or allocation of Taxes (such as financing agreements with Tax gross-up obligations or leases with Tax escalation provisions).

"Confidentiality Agreement" means the confidentiality agreement dated June 5, 2019 among First Reserve XIII Advisors, L.L.C., First Reserve XIV Advisors, L.L.C. and the Company.

"Consolidated Group" means any affiliated, combined, consolidated, unitary or similar group with respect to any Taxes, including any affiliated group within the meaning of Section 1504 of the Code electing to file consolidated federal Income Tax Returns and any similar group under foreign, state or local Law.

"Contract" means any contract, agreement, lease, mortgage, license, purchase order, bid, commitment, or any other legally binding instrument, arrangement, obligation or understanding of any kind, whether oral or written.

"control" (including the terms "controlled by" and "under common control with"), with respect to the relationship between or among two or more Persons, means the possession, directly or indirectly, of the power to direct or cause the direction of the affairs or management of a Person, whether through the ownership of voting securities, by contract or otherwise, including the ownership, directly or indirectly, of securities having the power to elect a majority of the board of directors or similar body governing the affairs of such Person.

"Controlled Group Liabilities" means any and all Liabilities of the Asset Seller, the Divested Companies or any of their respective ERISA Affiliates (a) under Title IV of ERISA, (b) under Sections 206(g), 302 or 303 of ERISA, (c) under Sections 412, 430, 431, 436 or 4971 of the Code and (d) as a result of the failure to comply with the continuation of coverage requirements of Section 601 et seq. of ERISA and Section 4980B of the Code.

"Credit Agreement" means that certain Credit Agreement, dated October 16, 2013, by and among TriMas Corporation, TriMas Company LLC and JPMorgan Chase bank, N.A., as Administrative Agent and Collateral Agent, and the various lenders from time to time thereto, as amended September 20, 2017.

"Debt Financing Sources" means the Persons (including the Lenders) that have committed to provide or have otherwise entered into agreements in connection with the Debt Financing or alternative debt financings in connection with the transactions contemplated hereby, including any joinder agreements or credit agreements entered into pursuant thereto or relating thereto, together with their respective Affiliates, officers, directors, employees, agents, advisors, counsels, consultants and other representatives involved in the Debt Financing, and their respective successors and assigns.

"Debt Obligations" means, with respect to any Person as of any date without duplication, (a) the principal of and accreted value and accrued but unpaid interest, and any prepayment premiums and penalties, redemption costs, related expenses, commitment and other fees, sale or liquidity

participation amounts, reimbursements, indemnities and other payment obligations payable, in respect of (i) indebtedness for borrowed money of such Person and (ii) indebtedness evidenced by notes, bonds, debentures or other similar instruments, the payment of which is such Person's responsibility or liability, (b) all obligations of such Person for the deferred purchase price of goods or services (other than trade payables incurred in the ordinary course of business), (c) all liabilities under or in connection with any letter of credit, bankers' acceptances, performance bonds, securities or similar obligations (in each case, to the extent drawn or a payment obligation is otherwise outstanding thereunder), (d) all liabilities arising out of interest rate and currency swap arrangements and any other hedging or similar arrangements, (e) all liabilities under capitalized leases or other leases recorded or required to be recorded as capitalized leases, (f) all unfunded obligations for deferred compensation (including arising from any Seller Benefit Plan) for any officer, director or employee of the Company or any of its subsidiaries that are required to be accrued for as a pension liability under GAAP, (g) all liabilities of any Divested Company resulting from underfunding of or failure to timely remit payments due with respect to any pension or other retirement obligation covering any present or former officer, director or employee of any Divested Company (for the avoidance of doubt, excluding any liabilities with respect to the Thailand Statutory Severance and Retirement Programs), (h) all indebtedness or obligations of any other Person of the types referred to in the preceding clauses (a) through (g) secured by any Encumbrance on any assets of such Person and (i) guarantees of obligations of any other Person of the types described in clauses (a) through (g) above by such Person.

"Disclosure Schedules" means the Schedules delivered by the Company concurrently with the execution and delivery of this Agreement.

"Disposed Lamons Businesses" means any business or line of business formerly owned or operated by as a part of the Business that has been disposed of, discontinued or liquidated by or on behalf of the Business prior to the Execution Date, including under any Contract providing for the sale of any such business or line of business (which businesses, for the avoidance of doubt, include the businesses operated by the Persons set forth on Schedule 1.1(j)).

"Duty" means any stamp, transaction or registration duty or similar charge imposed by any Governmental Authority, including any interest, fine, penalty, charge or other amount imposed in respect thereof.

"Encumbrance" means any security interest, pledge, mortgage, lien, transfer restriction, charge, Option, easement, claim, right of first refusal or similar property interest or any other encumbrance of any kind or nature.

"Environment" means soil, surface water, groundwater, stream sediment, surface or subsurface strata and ambient air.

"Environmental Claim" means any unresolved written notice, request for information, claim, demand, action, suit, complaint or proceeding by any Person alleging any actual or potential liability or violation under any Environmental Law or with respect to Hazardous Materials.

"Environmental Law" means any applicable international, foreign, federal, state, provincial or local Law, statute, ordinance, regulation, or valid and legally-binding Governmental Order relating to (a) the protection or restoration of the Environment or natural resources, (b) occupational health or safety, (c) the regulation, investigation, monitoring or remediation of Hazardous Materials, or (d) the exposure to, or the release, threatened release, or disposal of Hazardous Materials. For

purposes of this definition, "Environmental Law" includes, but is not limited to, the Comprehensive Environmental Response, Compensation, and Liability Act (42 U.S.C. § 9601 et seq.), the Resource Conservation and Recovery Act, the Federal Water Pollution Control Act, the Federal Clean Air Act, the Emergency Planning and Community Right to Know Act and any analogous Law.

"Equity Interests" means, with respect to a Person, any (a) capital stock, shares, membership interest, partnership interest (whether general or limited), unit or other equity interest or participation or any other right to receive a share of the profits or losses of, or distribution of assets of, such Person, and (b) any subscriptions, warrants, Contracts or other rights or Options or commitments of any kind or character relating to any of the foregoing securities of such Person.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and the rules and regulations promulgated thereunder.

"ERISA Affiliate" means, with respect to an entity, any trade or business (whether or not incorporated) (a) under common control (within the meaning of Section 414(b), (c), (m) or (o) of the Code or Section 4001(b)(1) of ERISA) with such entity, (b) which, together with such entity, is treated as a single employer under Section 414(t) of the Code or (c) that is a member of the same "controlled group" as such entity, trade or business pursuant to Section 4001(a)(14) of ERISA.

"Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.

"Excluded Liabilities" means (a) all Excluded Asset Seller Liabilities, (b) all Specified Excluded Liabilities, (c) all Transaction Expenses and Debt Obligations, (d) all Liabilities associated with any Disposed Lamons Business, (e) any Liabilities resulting from the failure to comply with the requirements of clause (21) of Annex 3 of the Foreign Business Act B.E. 2542 (1999) (and the failure to obtain any foreign business license required thereunder) as it relates to any charges received by Lamons Thailand in connection with the advanced payment of salary and other allowances to any employees of the Company or any of its Affiliates (including Rieke Packaging Australia), and (f) all Liability suffered or incurred by Buyer arising from or in connection with any question of R-L Netherlands Holdings', R-L International Holdings' and R-L UK Holdings' legal title to and ownership of the Lamons Thailand Shares and the validity and enforceability of the transfer of the Lamons Thailand Shares (i) from Miss Mattaga Sudthangtum, Miss Isara Lovanich and Miss Daosiree Chayasirisobhon to R-L Netherlands Holdings, (ii) from Miss Mattaga Sudthangtum to R-L International Holdings and (iii) from Miss Mattaga Sudthangtum to R-L UK Holdings.

"FMLA" means the Family and Medical Leave Act of 1993, as amended, and the rules and regulations promulgated thereunder.

"Fraud" means an intentional misrepresentation of a material fact by a Party with intent to deceive or mislead another Party and upon which such other Party justifiably relied.

"GAAP" means United States generally accepted accounting principles and practices.

"Government Contract" means any (a) prime contract, grant agreement, cooperative agreement or other type of Contract with a Governmental Authority to which the Asset Seller is a party and which is related to the operation of the Business or the Sold Assets or to which any Divested Company is a party or (b) any subcontract under any such Contract as described in clause

(a) to which the Asset Seller is a party and which is related to the operation of the Business or the Sold Assets or to which any Divested Company is a party.

"Governmental Authority" means any federal, state, local, tribal or foreign government, governmental, regulatory or administrative authority, agency or commission or any court, tribunal, judicial body or arbitrator (public or private).

"Governmental Order" means any order, writ, injunction, decree, judgment, assessment or award of a Governmental Authority.

"Hazardous Material" means any substance or material that is listed, defined, classified or regulated as a "hazardous substance," "hazardous waste," "toxic substance" or any other term of similar import under any Environmental Law, including petroleum, friable asbestos, asbestos containing materials, and polychlorinated biphenyls.

"HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder.

"Income Taxes" means any income, capital gains, franchise and similar Taxes.

"Intellectual Property" means any and all proprietary, industrial and intellectual property rights, under the law of any jurisdiction or rights under international treaties, both statutory and common law rights, including (a) patents, patent applications, and invention disclosures, worldwide, together with reissuances, continuations, continuations-in-part, divisions, supplementary protection certificates, extensions and re-examinations thereof, (b) registered and unregistered trademarks, trade names, trade dress, service marks, brand names, logos, Internet domain names and other indicia of source, worldwide, and registrations and applications for registration thereof, and their associated goodwill, (c) copyrights and copyrightable works (including software, databases and related items) and all other rights of authorship recognized by statute or otherwise, and all applications, registrations and renewals in connection therewith, (d) trade secrets and rights in other proprietary or confidential information, including inventions, improvements thereto, technical information, processes, formulae, technology, discoveries, know-how, specifications, designs, plans, manuals, drawings, research, computer programs, marketing studies, customer lists and (e) all other intellectual property and moral rights, and all registrations and applications for any of the foregoing.

"Intercompany Obligations" means (a) all intercompany notes, cash advances and payables between the Company or any of its Affiliates (other than the Asset Seller or the Divested Companies), on the one hand, and any of the Divested Companies, on the other hand, and (b) all intercompany notes, cash advances and payables between the Company or any of its Affiliates (other than the Asset Seller or the Divested Companies), on the one hand, and the Asset Seller, on the other hand, including those set forth on Schedule 1.1(d).

"IT Systems" means (a) all computing or communications systems and equipment, including any internet, intranet, extranet, e-mail, or voice mail systems and the hardware associated with such systems, (b) all software, the tangible media on which it is recorded (in any form) and all supporting documentation, data and databases and (c) all peripheral equipment related to the foregoing, including printers, scanners, switches, routers, network equipment and removable media.

"Knowledge Group" means, collectively, the individuals listed on Schedule 1.1(e).

"Knowledge of the Buyers" means the actual knowledge (after due inquiry of their respective direct reports) of the individuals listed on Schedule 1.1(f).

"Knowledge of the Company" means the actual knowledge (after due inquiry of their respective direct reports) of the individuals comprising the Knowledge Group.

"Law" means any statute, law, common law, treaty, judgment, ordinance, regulation or rule of any Governmental Authority.

"Liabilities" means any debt, liability, loss, damage or obligation (whether direct or indirect, absolute or contingent, accrued or unaccrued, liquidated or unliquidated, or due or to become due), including all costs and expenses relating thereto.

"Master Guaranty" means that certain Trimas Master Guaranty, dated June 5, 2002, by and between Seller Guarantor and the Company.

"Option" means any option, warrant, subscription, commitment, call, right or Contract requiring, and any securities which upon conversion, exercise or exchange would require, the issuance, sale or transfer of any Equity Interests or other securities convertible into, exchangeable for or evidencing the right to subscribe for or purchase shares, capital stock or other Equity Interests.

"Permits" means any permits, concessions, licenses, certificates, approvals, entitlements, registrations, waivers, exemptions, variances, rights of use, Orders and authorizations required under applicable Law.

"Permitted Encumbrances" means (a) Encumbrances for Taxes that are not yet due and payable or are being contested in good faith by appropriate Proceedings and for which adequate reserves have been established in the Financial Information in accordance with GAAP, (b) carriers', warehousemen's, materialmen's, mechanics', construction, labor, and landlord's liens and other similar liens in respect of property or assets imposed by applicable Law that were incurred in the ordinary course of business for obligations that are not past due or are being contested in good faith by appropriate Proceedings and for which adequate reserves have been established in the Financial Information in accordance with GAAP, (c) pledges or deposits made in the ordinary course of business to secure obligations under employment insurance, statutory pension plans, workers' compensation laws or similar legislation, (d) any easements, restrictions, covenants, servitudes, rights of way (including, without limiting the generality of the foregoing, easements, rights of way and agreements for sewers, drains, gas and water mains or electric light and power or telephone, telecommunications or cable conduits, poles, wires and cables) or similar matters of record relating to real property, if any, that do not, individually or in the aggregate, materially impair the value or continued use and operation of any property to which they relate in the Business, (e) zoning, building codes and other land use applicable Law regulating use or occupancy of any real property or activities that are conducted thereon that are imposed by any Governmental Authority which are not violated by the present use and operation of any property to which they relate. and which do not, individually or in the aggregate, materially impair the continued use and operation of any property to which they relate in the Business, and (f) restrictions on the transfer of securities arising under applicable Law.

"Person" means any individual, partnership, firm, corporation, association, trust, unincorporated organization, joint venture, limited liability company, Governmental Authority or other entity.

"Post-Closing Tax Period" means a taxable period that begins after the Closing Date.

"Pre-Closing Tax Period" means a taxable period that ends on or prior to the Closing Date.

"Proceeding" means any judicial, regulatory, administrative or arbitral actions, suits, investigations or proceedings (public or private) by or before any Governmental Authority.

"R&W Insurance Costs" means the costs of the R&W Insurance Policy, including the premium, surplus lines Taxes and fees, and any related broker compensation and underwriting fees.

"Required Consents" means the consents set forth on Schedule 1.1(h).

"R&W Insurance Policy" means the buyer-side representation and warranty insurance policy issued by the R&W Insurer to the Buyers with respect to this Agreement and conditionally bound as of the Execution Date, as evidenced by the binder agreement attached hereto as Exhibit C.

"R&W Insurer" means Tokio Marine HCC.

"Release" shall have the meaning provided in 42 U.S.C. Section 9601(22).

"Seller Material Adverse Effect" means a material adverse effect on the ability of the Company or any of its Affiliates to consummate the transactions contemplated by this Agreement or the Ancillary Agreements to which the Company or such Affiliate is a party.

"Seller Taxes" means any Taxes (i) imposed on the Asset Seller or any of its direct or indirect owners for any Tax period, (ii) that are Asset Taxes or unclaimed property or escheat obligations attributable to any Pre-Closing Tax Period or the portion of any Straddle Period ending on the Closing Date (determined in accordance with Section 5.5(b)), (iii) imposed on any Divested Company or any subsidiary of a Divested Company or for which any Divested Company or any subsidiary of any Divested Company may otherwise be liable for any Pre-Closing Tax Periods or for the portion of any Straddle Period ending on the Closing Date (determined in accordance with Section 5.5(b)), (iv) imposed on the Buyers or any of their Affiliates or the direct or indirect owners or shareholders of Buyers or their Affiliates (not taking into account any reduction in such Taxes resulting from the utilization of any loss carryforwards, credits or any other offsets or reductions by the Buyers or any of their Affiliates) as a result of a gross income inclusion under Section 951 or Section 951A of the Code with respect to income of any Divested Company or any subsidiary of a Divested Company for any U.S. federal Income Tax Pre-Closing Tax Period or the portion of any U.S. federal Income Tax Straddle Period ending on the Closing Date (determined in accordance with Section 5.5(b)) of such Divested Company or of any subsidiary of a Divested Company, as the case may be, (v) of any Consolidated Group (or any member thereof, other than the Divested Companies or any subsidiary of any Divested Company) of which any Divested Company or any subsidiary of any Divested Company (or any predecessor of any Divested Company or any subsidiary of any Divested Company) is or was a member on or prior to the Closing Date by reason of Treasury Regulations Section 1.1502-6(a) or any analogous or similar foreign, state or local Law, (vi) of any other Person for which any Divested Company or any subsidiary of any Divested Company is or has been liable as a transferee or successor, or by Contract other than any Commercial Tax Agreement, resulting from events, transactions or relationships occurring or existing prior to the Closing or (vii) that are Transfer Taxes for which the Sellers are liable pursuant to Section 5.4(d);

provided, that no such Tax will constitute a Seller Tax to the extent such Tax was included as a current liability in the determination of Final Working Capital.

"Specified Employees" means the Active Employees set forth on Schedule 1.1(b).

"Specified Excluded Liabilities" means all Liabilities set forth on Schedule 1.1(i) whether arising on, prior to or after the Closing.

"Straddle Period" means a taxable period beginning on or before and ending after the Closing Date.

"subsidiaries" means, with respect to any Person, any other Person 50% or more of the voting equity of which is owned, directly or indirectly, by such first Person or by one or more subsidiaries of such first Person or a combination thereof.

"Tax" or "Taxes" means (i) any and all taxes, assessments, fees and other governmental charges, in each case, in the nature of a tax, imposed by any Governmental Authority, including those on or measured by or referred to as income, gross receipts, capital, sales, harmonized sales, provincial or territorial sales, use, ad valorem, franchise, profits, license, goods and services, withholding, payroll, employment, excise, severance, stamp, occupation, premium, transfer, real property transfer, real property transfer gains, value added, property or windfall profits taxes, customs, duties or similar fees, similar assessments or similar charges, together with any related interest, penalties, and additions to tax, whether disputed or not, (ii) any unclaimed property or escheat obligations, together with any related interest, penalties, and additions, whether disputed or not, (iii) any liability for the payment of any amounts of the type described in clauses (i) or (ii) as a result of being a member of a Consolidate Group for any period and (iv) any liability for the payment of any amounts of the type described in clauses (i), (ii), or (iii) as a result of the operation of Law or any express or implied obligation to indemnify any other Person.

"Tax Refund" means any refund or credit of Taxes.

"Tax Return" means any return, form, report or statement required to be prepared and filed with any Governmental Authority with respect to Taxes, including any schedule or attachment thereto or amendment thereof.

"Taxing Authority" means, with respect to any Tax, the Governmental Authority that imposes such Tax and the agency (if any) charged with the collection of such Tax for such Governmental Authority.

"Transaction Expenses" shall mean, collectively, (i) all of the fees and expenses incurred by the Company or its Affiliates, or with respect to which the Company, Buyer or any of their respective Affiliates have any obligation to third parties as a result of Contracts by the Company or any of its Affiliates entered into prior to the Closing, in connection with the negotiation, documentation and consummation of this Agreement or the Ancillary Agreements and the transactions contemplated hereby and thereby, including all fees, expenses, disbursements and other similar amounts paid or payable to attorneys, financial advisors or accountants and all obligations under any engagement letter or other agreement or understanding with Bank of America Merrill Lynch or any Affiliate thereof and (ii) all stay, retention, change of control, severance, bonus, equity appreciation, phantom equity or similar payments due or payable by the Company or its Affiliates to any Person or the Buyer or any of its Affiliates to any Person as a result of Contracts entered into by the Company or any of

its Affiliates prior to the Closing, and any other accelerations or increases in rights or benefits provided by the Company or its Affiliates, whether payable or occurring prior to, on or after the Closing Date, under any Contract, plan or other arrangement, which obligation, in each case, arises on or before the Closing Date or in whole or in part as a result of the execution of this Agreement or the Ancillary Agreements or the consummation of the transactions contemplated hereby or thereby (including the employer portion of all Taxes that are payable in connection with such obligations).

"Transition Services Agreement" means the transition services agreement, dated as of the Closing Date, to be entered into by the Company and the Buyers (or their respective designees), substantially in the form of Exhibit D.

"Treasury Regulations" means the Treasury Regulations promulgated under the Code.

"Working Capital" means, as of the date of computation, an amount, which may be positive or negative, equal to (i) the total "current assets" of the Business as defined in accordance with GAAP (but excluding, without duplication, any Cash, the Excluded Assets, any operating lease right-of-use assets, current and deferred Tax assets, prepaid Taxes in excess of Tax Liabilities, any receivables from Company or any of its Affiliates, any receivables of the Company or any of its Affiliates (other than the Divested Companies) that do not constitute Sold Assets), minus (ii) the total "current liabilities" of the Business as defined in accordance with GAAP (including, without duplication, any current Tax liabilities, and excluding, without duplication, the Excluded Liabilities, Debt Obligations, Transaction Expenses and deferred Tax Liabilities). Solely for illustrative purposes, set forth on Exhibit E is a sample calculation of Working Capital as if the Closing had occurred on the Balance Sheet Date. For the avoidance of doubt, the accounts specified in Exhibit E shall be the sole accounts used in the calculation of Working Capital for purposes of this Agreement.

1.2 Other Defined Terms. The following terms shall have the meanings defined for such terms in the Sections set forth below:

<u>Term</u>	Section
Acquisition Transaction	5.16
Allocation	2.7
Agreement	Preamble
Annual Financial Information	3.4(a)
Antitrust Division	5.3(b)
Asset Buyer	Preamble
Asset Seller	Recitals
Assumed Liabilities	2.4(a)
Balance Sheet	3.4(a)
Balance Sheet Date	3.4(a)
Benefits Transition Date	5.9(b)
Business	Recitals
Buyer 401(k) Plan	5.9(f)
Buyer FSA Plan	5.9(i)
Buyers	Preamble

Buyer Indemnified Persons 9.1	<u>Term</u>	Section
Buyer Related Party 8.2(c) Buyer Released Claims 5.24(a) Buyer Welfare Plans 5.9(e) Cap 9.1 Claim Notice 9.6 Closing 2.8 Closing Balance Sheet 2.6(b) Closing Date 2.8 Closing Purchase Price 2.5 Collective Bargaining Agreement 5.10(a) Commitment Letters 4.6(a) Company Preamble Company Marks 5.17 Competing Activity 5.12(a)(i) CPA Firm 2.6(c) Credit Support Obligations 3.29 Debt Commitment Letter 4.6(a) Debt Financing 4.6(a) Debt Epinancing 4.6(a) Debt Epinated Companies Recitals Effect 1.1 (Definition of "Businness Material Adverse Effect") End Date 8.1(b) Equity Buyer Preamble Equity Financing 4.6(a) Equity Financing 4.6(a) Equity Financing 4.6(a)	Buyer Indemnified Persons	9.1
Buyer Released Claims 5.24(a) Buyer Welfare Plans 5.9(e) Cap 9.1 Claim Notice 9.6 Closing Balance Sheet 2.6(b) Closing Balance Sheet 2.6(b) Closing Purchase Price 2.5 Collective Bargaining Agreement 5.10(a) Commitment Letters 4.6(a) Company Preamble Company Marks 5.17 Company Marks 5.17 Competing Activity 5.12(a)(i) CPA Firm 2.6(c) Credit Support Obligations 3.29 Debt Commitment Letter 4.6(a) Debt Documents 5.21(a) Debt Financing 4.6(a) Deductible 9.1 Divested Companies Recitals Effect 1.1 (Definition of "Business Material Adverse Effect") End Date 8.1(b) Equity Buyer Preamble Equity Financing 4.6(a) Equity Financing 4.6(a) Equity Financing 4.6(a) Equity Sellers Recitals Estimated Un	-	8.2(c)
Buyer Releasee 5.25(a) Buyer Welfare Plans 5.9(e) Cap 9.1 Claim Notice 9.6 Closing 2.8 Closing Balance Sheet 2.6(b) Closing Date 2.8 Closing Purchase Price 2.5 Collective Bargaining Agreement 5.10(a) Commitment Letters 4.6(a) Company Preamble Company Marks 5.17 Competing Activity 5.12(a)(i) CPA Firm 2.6(c) Credit Support Obligations 3.29 Debt Commitment Letter 4.6(a) Debt Financing 4.6(a) Debt Financing 4.6(a) Deductible 9.1 Divested Companies Recitals Effect 1.1 (Definition of "Business Material Adverse Effect") End Date 8.1(b) Equity Buyer Preamble Equity Commitment Letter 4.6(a) Equity Financing 4.6(a) Equity Financing 4.6(a) Equity S	•	• •
Buyer Welfare Plans	•	` '
Cap 9.1 Claim Notice 9.6 Closing 2.8 Closing Balance Sheet 2.6(b) Closing Date 2.8 Closing Purchase Price 2.5 Collective Bargaining Agreement 5.10(a) Commitment Letters 4.6(a) Company Preamble Company Marks 5.17 Competing Activity 5.12(a)(i) CPA Firm 2.6(c) Credit Support Obligations 3.29 Debt Commitment Letter 4.6(a) Debt Documents 5.21(a) Debt Financing 4.6(a) Deductible 9.1 Divested Companies Recitals Effect 1.1 (Definition of "Business Material Adverse Effect") End Date 8.1(b) Equity Buyer Preamble Equity Commitment Letter 4.6(a) Equity Financing 4.6(a) Equity Financing 4.6(a) Equity Financing 4.6(a) Equity Sellers Recitals Estimated	-	• •
Claim Notice 9.6 Closing 2.8 Closing Balance Sheet 2.6(b) Closing Date 2.8 Closing Purchase Price 2.5 Collective Bargaining Agreement 5.10(a) Commitment Letters 4.6(a) Company Preamble Company Marks 5.17 Competing Activity 5.12(a)(i) CPA Firm 2.6(c) Credit Support Obligations 3.29 Debt Commitment Letter 4.6(a) Debt Documents 5.21(a) Debt Financing 4.6(a) Deductible 9.1 Divested Companies Recitals Effect 1.1 (Definition of "Business Material Adverse Effect") End Date 8.1(b) Equity Buyer Preamble Equity Commitment Letter 4.6(a) Equity Financing 4.6(a) Equity Financing 4.6(a) Equity Investor 4.6(a) Equity Sellers Recitals Estimated Unpaid Expenses 2.6(a) <t< td=""><td></td><td>` '</td></t<>		` '
Closing Balance Sheet Closing Date Closing Purchase Price Collective Bargaining Agreement Commitment Letters Company Company Company Company Company Competing Activity CPA Firm Codit Support Obligations Debt Commitment Letter Debt Documents Debt Financing Deductible Divested Companies Effect L1.1 (Definition of "Business Material Adverse Effect") End Date Equity Buyer Equity Commitment Letter Equity Commitment Letter Quity Financing Activate Equity Sellers Equity Sellers Estimated Unpaid Expenses Estimated Undebtedness Estimated Working Capital Excluded Assets Estimated Working Capital Excluded Assets 2.6(a) Estimated Working Capital Excluded Assets 2.6(a) Excluded Assets 2.6(a) Excluded Assets 2.6(a) Excluded Assets 2.6(a) Excluded Assets	•	9.6
Closing Balance Sheet Closing Date Closing Purchase Price Collective Bargaining Agreement Commitment Letters Company Company Company Company Company Competing Activity CPA Firm Codit Support Obligations Debt Commitment Letter Debt Documents Debt Financing Deductible Divested Companies Effect L1.1 (Definition of "Business Material Adverse Effect") End Date Equity Buyer Equity Commitment Letter Equity Commitment Letter Quity Financing Activate Equity Sellers Equity Sellers Estimated Unpaid Expenses Estimated Undebtedness Estimated Working Capital Excluded Assets Estimated Working Capital Excluded Assets 2.6(a) Estimated Working Capital Excluded Assets 2.6(a) Excluded Assets 2.6(a) Excluded Assets 2.6(a) Excluded Assets 2.6(a) Excluded Assets	Closing	2.8
Closing Date Closing Purchase Price Collective Bargaining Agreement Commitment Letters 4.6(a) Company Preamble Company Marks Competing Activity CPA Firm 2.6(c) Credit Support Obligations Debt Commitment Letter 4.6(a) Debt Documents Debt Financing Deductible Divested Companies Effect 1.1 (Definition of "Business Material Adverse Effect") End Date Equity Buyer Equity Commitment Letter Equity Financing 4.6(a) Equity Financing 4.6(a) Equity Sellers Estimated Unpaid Expenses Estimated Balance Sheet Estimated Working Capital Estimated Working Capital Excluded Assets Estimated Working Capital Excluded Assets 2.5 Case Simated Mase Sin(a)	_	2.6(b)
Closing Purchase Price Collective Bargaining Agreement Commitment Letters 4.6(a) Company Preamble Company Marks 5.17 Competing Activity CPA Firm 2.6(c) Credit Support Obligations Debt Commitment Letter 4.6(a) Debt Documents Debt Financing Deductible Divested Companies Effect 1.1 (Definition of "Business Material Adverse Effect") End Date Equity Buyer Equity Commitment Letter Equity Financing 4.6(a) Equity Sellers Estimated Unpaid Expenses Estimated Indebtedness Estimated Working Capital Excluded Assets Estimated Working Capital Excluded Assets 2.6(a) Excluded Assets	-	` '
Collective Bargaining Agreement Commitment Letters 4.6(a) Company Preamble Company Marks 5.17 Competing Activity CPA Firm 2.6(c) Credit Support Obligations 3.29 Debt Commitment Letter 4.6(a) Debt Documents 5.21(a) Debt Financing 4.6(a) Deductible Divested Companies Effect 1.1 (Definition of "Business Material Adverse Effect") End Date Equity Buyer Preamble Equity Commitment Letter 4.6(a) Equity Financing 4.6(a) Equity Sellers Estimated Unpaid Expenses Estimated Balance Sheet Estimated Working Capital Excluded Assets 2.6(a) Estimated Working Capital Excluded Assets	-	2.5
Commitment Letters 4.6(a) Company Preamble Company Marks 5.17 Competing Activity 5.12(a)(i) CPA Firm 2.6(c) Credit Support Obligations 3.29 Debt Commitment Letter 4.6(a) Debt Documents 5.21(a) Debt Financing 4.6(a) Deductible 9.1 Divested Companies Recitals Effect 1.1 (Definition of "Business Material Adverse Effect") End Date 8.1(b) Equity Buyer Preamble Equity Commitment Letter 4.6(a) Equity Financing 4.6(a) Equity Financing 4.6(a) Equity Investor 4.6(a) Equity Sellers Recitals Estimated Unpaid Expenses 2.6(a) Estimated Balance Sheet 2.6(a) Estimated Indebtedness 2.6(a) Estimated Working Capital Excluded Assets 2.3	_	
Company Marks 5.17 Competing Activity 5.12(a)(i) CPA Firm 2.6(c) Credit Support Obligations 3.29 Debt Commitment Letter 4.6(a) Debt Documents 5.21(a) Debt Financing 4.6(a) Deductible 9.1 Divested Companies Recitals Effect 1.1 (Definition of "Business Material Adverse Effect") End Date 8.1(b) Equity Buyer Preamble Equity Commitment Letter 4.6(a) Equity Financing 4.6(a) Equity Investor 4.6(a) Equity Sellers Recitals Estimated Unpaid Expenses 2.6(a) Estimated Cash 2.6(a) Estimated Working Capital Excluded Assets 2.6(a) Estimated Working Capital Excluded Assets		
Company Marks Competing Activity CPA Firm 2.6(c) Credit Support Obligations Debt Commitment Letter Debt Documents Debt Financing Deductible Divested Companies Effect L1.1 (Definition of "Business Material Adverse Effect") End Date Equity Buyer Equity Commitment Letter Equity Financing A.6(a) Equity Investor Equity Sellers Estimated Unpaid Expenses Estimated Indebtedness Estimated Working Capital Excluded Assets 5.17 5.12(a)(i) 5.12(a)(i) 5.12(a)(i) 6.12(a)(i) 6.12(a)(a)(a)(a)(a)(a)(a)(a)(a)(a)(a)(a)(a)(Company	` '
Competing Activity CPA Firm 2.6(c) Credit Support Obligations Debt Commitment Letter Debt Documents Debt Financing Deductible Divested Companies Effect Support Obligations Effect Support Obligations Debt Financing Deductible Divested Companies Effect Support Obligations Effect Support Obligations Support		5.17
CPA Firm Credit Support Obligations Debt Commitment Letter Debt Documents Debt Documents Debt Financing Deductible Deductible Divested Companies Effect Series Detect Description Divested Companies Divested Companies Divested Companies Divested Companies Effect Series Description Divested Companies	• •	5.12(a)(i)
Credit Support Obligations Debt Commitment Letter Debt Documents Debt Documents Debt Financing Debt Financing Deductible Divested Companies Effect I.1 (Definition of "Business Material Adverse Effect") End Date Equity Buyer Equity Commitment Letter Equity Financing Equity Financing Equity Investor Equity Sellers Estimated Unpaid Expenses Estimated Unpaid Expenses Estimated Cash Estimated Indebtedness Estimated Working Capital Excluded Assets 3.29 4.6(a) 5.21(a) 9.1 1.1 (Definition of "Business Material Adverse Effect") 8.1(b) Freamble 4.6(a) Freamble 4.6(a) Equity Financing 4.6(a) Ecitals Estimated Unpaid Expenses 2.6(a) Estimated Sheet 2.6(a) Estimated Cash Estimated Norking Capital Excluded Assets		
Debt Commitment Letter Debt Documents Debt Financing Deductible Divested Companies Effect Effect Substitute the service of the	Credit Support Obligations	` '
Debt Documents Debt Financing Deductible Divested Companies Effect Effect Summary Equity Buyer Equity Commitment Letter Equity Financing Equity Financing Equity Investor Equity Sellers Estimated Unpaid Expenses Estimated Balance Sheet Estimated Indebtedness Estimated Working Capital Excluded Assets 5.21(a) 5.21(a) 6.2 4.6(a) 9.1 T.1 (Definition of "Business Material Adverse Effect") 8.1(b) Preamble Preamble 4.6(a) Equity Financing 4.6(a) Equity Financing Equity Investor 4.6(a) Ectilals Estimated Unpaid Expenses Estimated Unpaid Expenses Estimated Unpaid Expenses Estimated Vorking Capital Excluded Assets		
Debt Financing Deductible Divested Companies Effect Effect Effect End Date Equity Buyer Equity Commitment Letter Equity Financing Equity Financing Equity Investor Equity Sellers Estimated Unpaid Expenses Estimated Balance Sheet Estimated Cash Estimated Indebtedness Estimated Working Capital Excluded Assets 4.6(a) 4.6(a) Excluded Assets		` '
Deductible 9.1 Divested Companies Recitals Effect 1.1 (Definition of "Business Material Adverse Effect") End Date 8.1(b) Equity Buyer Preamble Equity Commitment Letter 4.6(a) Equity Financing 4.6(a) Equity Investor 4.6(a) Equity Sellers Recitals Estimated Unpaid Expenses 2.6(a) Estimated Balance Sheet 2.6(a) Estimated Cash 2.6(a) Estimated Indebtedness 2.6(a) Estimated Working Capital Excluded Assets 2.3		• •
Divested Companies Effect 1.1 (Definition of "Business Material Adverse Effect") End Date Equity Buyer Equity Commitment Letter Equity Financing Equity Investor Equity Sellers Estimated Unpaid Expenses Estimated Balance Sheet Estimated Cash Estimated Indebtedness Estimated Working Capital Excluded Assets Recitals 2.6(a) 2.6(a) Estimated Working Capital Excluded Assets	-	` '
Effect 1.1 (Definition of "Business Material Adverse Effect") End Date 8.1(b) Equity Buyer Preamble Equity Commitment Letter 4.6(a) Equity Financing 4.6(a) Equity Investor 4.6(a) Equity Sellers Recitals Estimated Unpaid Expenses 2.6(a) Estimated Balance Sheet 2.6(a) Estimated Cash Estimated Indebtedness 2.6(a) Estimated Working Capital Excluded Assets 2.6(a) Excluded Assets		
End Date 8.1(b) Equity Buyer Preamble Equity Commitment Letter 4.6(a) Equity Financing 4.6(a) Equity Investor 4.6(a) Equity Sellers Recitals Estimated Unpaid Expenses 2.6(a) Estimated Balance Sheet 2.6(a) Estimated Cash 2.6(a) Estimated Indebtedness 2.6(a) Estimated Working Capital 2.6(a) Excluded Assets 2.3	·	
End Date8.1(b)Equity BuyerPreambleEquity Commitment Letter4.6(a)Equity Financing4.6(a)Equity Investor4.6(a)Equity SellersRecitalsEstimated Unpaid Expenses2.6(a)Estimated Balance Sheet2.6(a)Estimated Cash2.6(a)Estimated Indebtedness2.6(a)Estimated Working Capital2.6(a)Excluded Assets2.3		
Equity BuyerPreambleEquity Commitment Letter4.6(a)Equity Financing4.6(a)Equity Investor4.6(a)Equity SellersRecitalsEstimated Unpaid Expenses2.6(a)Estimated Balance Sheet2.6(a)Estimated Cash2.6(a)Estimated Indebtedness2.6(a)Estimated Working Capital2.6(a)Excluded Assets2.3		Adverse Effect")
Equity Commitment Letter 4.6(a) Equity Financing 4.6(a) Equity Investor 4.6(a) Equity Sellers Recitals Estimated Unpaid Expenses 2.6(a) Estimated Balance Sheet 2.6(a) Estimated Cash 2.6(a) Estimated Indebtedness 2.6(a) Estimated Working Capital 2.6(a) Excluded Assets 2.3	End Date	8.1(b)
Equity Financing 4.6(a) Equity Investor 4.6(a) Equity Sellers Recitals Estimated Unpaid Expenses 2.6(a) Estimated Balance Sheet 2.6(a) Estimated Cash 2.6(a) Estimated Indebtedness 2.6(a) Estimated Working Capital 2.6(a) Excluded Assets 2.3	Equity Buyer	Preamble
Equity Investor Equity Sellers Estimated Unpaid Expenses Estimated Balance Sheet Estimated Cash Estimated Indebtedness Estimated Working Capital Excluded Assets 4.6(a) Recitals 2.6(a) 2.6(a) 2.6(a) 2.6(a) 2.6(a)	Equity Commitment Letter	4.6(a)
Equity Sellers Estimated Unpaid Expenses Estimated Balance Sheet Estimated Cash Estimated Indebtedness Estimated Working Capital Excluded Assets Recitals 2.6(a) 2.6(a) 2.6(a) 2.6(a) 2.6(a)	Equity Financing	4.6(a)
Estimated Unpaid Expenses 2.6(a) Estimated Balance Sheet 2.6(a) Estimated Cash 2.6(a) Estimated Indebtedness 2.6(a) Estimated Working Capital 2.6(a) Excluded Assets 2.3	Equity Investor	4.6(a)
Estimated Balance Sheet 2.6(a) Estimated Cash 2.6(a) Estimated Indebtedness 2.6(a) Estimated Working Capital 2.6(a) Excluded Assets 2.3	Equity Sellers	Recitals
Estimated Cash 2.6(a) Estimated Indebtedness 2.6(a) Estimated Working Capital Excluded Assets 2.6(a) 2.6(a)	Estimated Unpaid Expenses	2.6(a)
Estimated Indebtedness 2.6(a) Estimated Working Capital 2.6(a) Excluded Assets 2.3	Estimated Balance Sheet	2.6(a)
Estimated Working Capital 2.6(a) Excluded Assets 2.3	Estimated Cash	2.6(a)
Excluded Assets 2.3	Estimated Indebtedness	2.6(a)
	Estimated Working Capital	2.6(a)
Excluded Asset Seller Liabilities 2.4(b)	Excluded Assets	2.3
	Excluded Asset Seller Liabilities	2.4(b)
Execution Date Preamble	Execution Date	Preamble
Express Company Representations 3.31	Express Company Representations	3.31
Fee Letter 4.6(a)	Fee Letter	4.6(a)

<u>Term</u>	<u>Section</u>
Final Unpaid Expenses	2.6(b)
Final Cash	2.6(b)
Final Indebtedness	2.6(b)
Final Statement	2.6(c)
Final Working Capital	2.6(b)
Financial Information	3.4(a)
Financing	4.6(a)
Flex Plan Amount	5.9(i)
FTC	5.3(b)
Fundamental Representations	9.5
General Enforceability Exceptions	3.2
Indemnified Party	9.6
Indemnifying Party	9.6
Interim Financial Information	3.4(a)
Interim Period	5.1
Lamons Belgium	Recitals
Lamons Belgium Shares	Recitals
Lamons Canada	Recitals
Lamons Canada Shares	Recitals
Lamons Netherlands	Recitals
Lamons Singapore	Recitals
Lamons Singapore Shares	Recitals
Lamons Spain	Recitals
Lamons Spain Shares	Recitals
Lamons Thailand	Recitals
Lamons Thailand Shares	Recitals
Leased Real Property	3.18(a)
Lenders	4.6(a)
Limited Guarantee	4.8
Losses	9.1
Major Customers	3.19(a)
Major Suppliers	3.19(b)
Material Contracts	3.14(a)
Non-Assigned Asset	5.13(a)
Non-U.S. Business Employee	5.9(i)
Non-U.S. Transferred Employee	5.9(a)
OFAC	3.21(a)
Party	Preamble
Pre-Closing Tax Returns	5.4(c)
Post-Closing Statement	2.6(b)
Post-Closing Statement Objection	2.6(c)
Pre-Closing Statement	2.6(a)

<u>Term</u>	<u>Section</u>
Purchase Price	2.5
Real Property Lease	3.18(a)
Required Information	5.21(c)(iv)
Reverse Termination Fee	8.2(b)
R-L International Holdings	Recitals
R-L Netherlands Holdings	Recitals
R-L UK Holdings	Recitals
Sale Process Confidentiality Agreements	5.15(d)
Seller Benefit Plans	3.11(a)
Seller FSA Plan	5.9(i)
Seller Guarantor	Preamble
Seller Indemnified Persons	9.2
Seller Released Claims	5.24(b)
Seller Releasee	5.24(b)
Sellers	Recitals
Sellers' Welfare Plans	5.9(e)
Shares	Recitals
Sold Assets	2.1
Straddle Period Tax Returns	5.4(c)
Target Working Capital	2.6(a)
Tax Benefit	9.4
Tax Claim	5.6
Tax Refunds	5.7(a)
Third Party Claim	9.6
Total Consideration	2.5
Transferred Employees	5.9(a)
Transfer Taxes	5.4(d)
Trigger Date	2.6(d)
Union Employee	5.10(a)
U.S. Transferred Employee	5.9(a)

1.3 Certain Interpretive Matters.

- (a) The words "hereof," "herein," "hereby," "hereinafter" and "hereunder" and words of similar import when used in this Agreement, the Ancillary Agreements or the Exhibits or Schedules hereto, refer to this Agreement, the Ancillary Agreement or the Exhibit or Schedule in which any such word is used and not to any particular provision of this Agreement, the Ancillary Agreements or the Exhibits or Schedules in which any such word is used. References to any Article, Section, Schedule and Exhibit refer to an Article or Section of, or a Schedule or Exhibit to, this Agreement unless otherwise expressly specified.
- (b) In this Agreement, the Ancillary Agreements and the Exhibits and Schedules hereto, (i) the meaning of defined terms shall be equally applicable to the singular and plural forms of the defined terms, (ii) any pronoun or pronouns shall be deemed to include both the singular and

the plural, (iii) the term "or" is disjunctive and not exclusive, (iv) the words "include," "includes" or "including" shall be deemed to be followed by the words "without limitation," whether or not they are in fact followed by those words or words of like import, (v) the word "extent" in the phrase "to the extent" means the degree to which a subject or other thing extends, and such phrase shall not simply mean "if," (vi) references to agreements and other documents shall be deemed to include all subsequent amendments and other modifications thereto, (vii) any reference to any Law shall be deemed also to refer to all rules and regulations promulgated under such Law, unless the context expressly requires otherwise, (viii) references to dollars or "\$" means U.S. dollars, (ix) any reference to gender shall include all genders, (x) all accounting terms used herein and not expressly defined herein shall have the meanings given to them under GAAP and (xi) any document or item will be deemed "furnished," "delivered," "provided" or "made available" within the meaning of this Agreement or any Ancillary Agreement if such document or item is (A) posted in the electronic data room to which the Buyers have access at least two days prior to the Execution Date or (B) actually (including electronically) delivered or provided to the Buyers at least two days prior to the Execution Date.

- (c) The provision of a Table of Contents and the division of this Agreement, the Ancillary Agreements, the Exhibits and Schedules hereto into Articles, Sections and other subdivisions and the insertion of headings herein and therein are for convenience of reference only and shall not affect or be utilized in construing or interpreting this Agreement, the Ancillary Agreements or the Exhibits and Schedules hereto, as the case may be. All references in this Agreement to any "Section" are to the corresponding Section of this Agreement unless otherwise specified. All references to any "Section" in any Ancillary Agreement, Exhibit or Schedule hereto are to the corresponding Section of such Ancillary Agreement, Exhibit or Schedule in which such reference appears, unless otherwise specified.
- (d) The Parties have participated jointly in the negotiation and drafting of this Agreement, the Ancillary Agreements and the Exhibits and Schedules hereto. In the event an ambiguity or question of intent or interpretation arises, this Agreement, the Ancillary Agreements and the Exhibits and Schedules hereto must be construed as if drafted jointly by the Parties and no presumption or burden of proof will arise favoring or disfavoring any Party by virtue of the authorship of any of the provisions of this Agreement, the Ancillary Agreements or the Exhibits or Schedules hereto.
- (e) When calculating the period of time before which, within which or following which, any act is to be done or step taken pursuant to this Agreement, the Ancillary Agreements or any Exhibit or Schedule hereto, the date that is the reference date in calculating such period shall be excluded. If the last day of such period is a non-Business Day, the period in question shall end on the next succeeding Business Day.
- (f) The Exhibits and Disclosure Schedules to this Agreement are hereby incorporated and made a part hereof and are an integral part of this Agreement. All Exhibits and Disclosure Schedules annexed hereto or referred to herein are hereby incorporated into and made a part of this Agreement as if set forth in full herein. Any matter or item disclosed on one Disclosure Schedule shall be deemed to have been disclosed and incorporated by reference in each other Disclosure Schedule, but only to the extent that the relevance of such matter or item to such other Disclosure Schedule is reasonably apparent on its face. No disclosure on a Disclosure Schedule relating to a possible breach or violation of any contract, Law or Governmental Order shall be construed as an admission or indication that such breach or violation exists or has actually occurred.

Information, matters and items disclosed or reflected in the Disclosure Schedules are not necessarily limited to information, matters and items that are required by this Agreement to be disclosed in the Disclosure Schedules. The disclosure of any information, matter or item in any Disclosure Schedule shall not be deemed to constitute an acknowledgement that any such information, matter or item (or any non-disclosed information, matter or item of comparable or greater significance) is required to be disclosed or is otherwise material. Any capitalized terms used in any Disclosure Schedule or Exhibit but not otherwise defined therein shall be defined as set forth in this Agreement. Any disclosure in the Disclosure Schedules that refers to a document is qualified in its entirety by reference to the text of such document, including all amendments, exhibits, schedules and other attachments thereto, in each case to the extent made available to the Buyers. References herein to "Exhibits" shall be deemed to include "Annexes."

ARTICLE II PURCHASE AND SALE

- 2.1 Purchase and Sale of the Sold Assets. On the Closing Date and subject to the terms and conditions set forth in this Agreement, the Company shall cause the Asset Seller to sell, assign, transfer, convey and deliver to the Asset Buyer (or its designee(s)), free and clear of all Encumbrances other than Permitted Encumbrances, and the Asset Buyer (or its designee(s)) shall purchase and acquire from the Asset Seller, all of the right, title and interest of the Asset Seller in and to all of the assets, property, goodwill, accounts, and business of every kind or nature, whether real, personal or mixed, tangible or intangible, choate or inchoate, wherever situated, and whether owned, leased or licensed by the Asset Seller, in each case that relate to, or are used in or necessary for the conduct of, the Business, including the following assets, properties and rights of the Asset Seller (other than the Excluded Assets) (collectively, the "Sold Assets"):
 - (a) the Contracts and other instruments and commitments relating to the Business;
 - (b) all worldwide rights in Acquired Intellectual Property;
 - (c) all inventory of the Business, including raw materials, works in process and finished products;
- (d) all equipment, vehicles, tooling, fixtures, leasehold improvements, machinery, furniture and furnishings of the Business;
- (e) all computer hardware, firmware, databases, software, systems, servers, drivers, other electronic data processing and information technology infrastructure, and other similar or related items of automated, computerized or software systems (including design tools, systems documentation and instructions related thereto) of the Business;
 - (f) all accounts receivable, notes receivable and mortgages receivable of the Business;
- (g) all deposits (including security deposits), letters of credit of which the Business is a beneficiary, performance and surety bonds, prepayments and prepaid or advanced payments and expenses, trade accounts and other accounts and notes receivable of the Business;
- (h) except as set forth on Schedule 2.3(l), all rights to any action, suit or claim of any nature relating to the Business, whether arising by way of counterclaim or otherwise, including

any insurance proceeds in the form of (i) cash received by the Company or its Affiliates after the Execution Date in respect of an insurance claim relating to any Assumed Liability, Sold Asset or the Business, or (ii) any receivables or claims in respect of an insurance claim relating to any Assumed Liability, Sold Asset or the Business;

- (i) all rights under warranties, indemnities and all similar rights against third parties related to the Assumed Liabilities, the Sold Assets or the Business;
- (j) originals, or where not available, copies, of all books and records related to the Business (including, to the extent permitted by applicable Law, personnel files (or copies thereof) for all Active Employees who become Transferred Employees), the Sold Assets and the Assumed Liabilities;
 - (k) all customer and vendor lists, data and databases solely to the extent used in the Business; and
 - (I) all goodwill associated with any of the items described in the foregoing clauses (a) through (k).
- 2.2 Purchase and Sale of the Shares. On the Closing Date and subject to the terms and conditions set forth in this Agreement, the Company shall cause the Equity Sellers to sell, assign and transfer to the Equity Buyer (or its designee(s)), and the Equity Buyer (or its designee(s)) shall purchase and acquire, all of the Equity Sellers' right, title and interest in and to their respective Shares, free and clear of all Encumbrances other than Encumbrances arising from restrictions on sale, assignment or transfer of shares of capital stock under applicable Law.
- 2.3 Excluded Assets. The Sold Assets shall not include, and the Asset Seller shall not sell, assign, transfer, convey or deliver to the Asset Buyer (or its designee(s)), and the Asset Buyer (or its designee(s)) shall not purchase or acquire, any right, title or interest in or to any of the following assets (collectively, the "Excluded Assets"):
- (a) any assets of the Asset Seller that to do not relate to, or are not used or held for use in, or necessary for the conduct of, the Business;
 - (b) any Cash owned by the Asset Seller as of the Closing Date;
- (c) the organizational documents, taxpayer and other identification numbers, minute and record books and the company seals of the Asset Seller;
- (d) any rights to the Asset Seller's insurance policies, premiums or proceeds for insurance coverages relating to the Excluded Assets or the Excluded Liabilities for any period, and any other recovery by the Asset Seller for the benefit of or otherwise relating to the Excluded Assets or the Excluded Liabilities from any Person (except for any items included in the calculation of Estimated Working Capital or Final Working Capital);
- (e) any rights to any refunds, credits, prepayments, overpayments and deposits of the Asset Seller with any Governmental Authority;
- (f) all Tax Returns other than Tax Returns related to Asset Taxes with respect to the Sold Assets and financial statements of the Asset Seller and all records (including working

papers) related thereto, except to the extent related to the Business, any Sold Assets or any Assumed Liabilities;

- (g) all of the Asset Seller's causes of action, claims, credits, demands or rights of set-off against third parties, in each case, to the extent related to any Excluded Asset, including any books, records and privileged information to the extent relating thereto;
 - (h) all rights that accrue to the Asset Seller under this Agreement;
- (i) all rights of the Asset Seller under, all funds and property held in trust or any other funding vehicle pursuant to, and all insurance contracts providing funding for, any Seller Benefit Plans;
- (j) any trademark, trade name, service mark, service name, website or domain name owned by the Company or its Affiliates (other than those containing the name "LAMONS" and any other trademark, trade name, service mark, service name, website or domain name used exclusively in the Business), including those containing the names "TRIMAS," "RIEKE," and the names included on Schedule 2.3(j) and, in each case, the goodwill associated therewith;
 - (k) the Contracts set forth on Schedule 2.3(k);
 - (I) the assets listed in Schedule 2.3(I);
- (m) any assets of the Asset Seller that exclusively relate to, or are exclusively used or held for use in, a Disposed Lamons Business; and
 - (n) the Asset Seller's Equity Interests in TriMas International Holdings LLC.

Notwithstanding anything to the contrary contained in this Agreement or any of the Ancillary Agreements, the Buyers acknowledge and agree that all of the following shall remain the property of the Company, and neither the Buyers nor any of their respective Affiliates (including, after the Closing, the Divested Companies) shall have any interest therein: (A) all confidentiality agreements entered into with prospective purchasers of the Business or any portion thereof; (B) all records and reports prepared or received by the Company or any of its Affiliates in connection with the sale of the Business and the transactions contemplated hereby (other than any confidentiality or similar Contracts between the Company or its Affiliates and any prospective purchasers of the Business or any portion thereof), including all analyses relating to the Business or the Buyers so prepared or received; (C) all bids and expressions of interest received from third parties with respect thereto; and (D) all privileged materials, documents and records in the possession of any of the Company or its Affiliates (including the Divested Companies), to the extent such materials, documents and records are (1) not related to the Business or (2) related only to any Excluded Asset or Excluded Liability. The Buyers further acknowledge and agree that, with respect to any Proceeding or dispute between the Company or any of its Affiliates. on the one hand, and the Buyers or the Divested Companies (post-Closing), on the other hand, only the Company may waive any evidentiary privilege that may attach to a pre-Closing communication that exclusively concerns such Proceeding or dispute and is determined by a court of competent jurisdiction to be subject to attorney-client privilege, and neither the Buyers nor the Divested Companies nor any of their respective Affiliates, shall have the right hereunder to compel disclosure of such privileged information.

- 2.4 Assumption of Liabilities; Excluded Liabilities.
- (a) Assumed Liabilities. On the terms and subject to the conditions set forth in this Agreement, at the Closing, the Asset Buyer shall assume effective as of the Closing and shall (or shall cause its Affiliates to) thereafter pay, perform, be responsible for and discharge as and when due all Liabilities of the Asset Seller (other than the Excluded Asset Seller Liabilities), irrespective of whether the same shall arise prior to, on or following the Closing Date (collectively, the "Assumed Liabilities"). The Buyers' obligations under this Section 2.4(a) will not be subject to offset or reduction by reason of any actual or alleged breach of any representation, warranty, covenant or agreement contained in this Agreement or the Ancillary Agreements or any closing or other document contemplated by this Agreement or the Ancillary Agreements, any right or alleged right of indemnification hereunder or for any other reason.
- (b) Excluded Asset Seller Liabilities. The Buyers shall not assume, and shall not have been deemed to assume, any of the following Liabilities of the Asset Seller (collectively, the "Excluded Asset Seller Liabilities"):
 - (i) all Liabilities arising out of or related to the Excluded Assets whether arising on, prior to or after the Closing Date, including any such Liabilities for Taxes, and all Liabilities associated with the Disposed Lamons Businesses;
 - (ii) except as otherwise provided in Section 5.9, (x) all Liabilities under any Seller Benefit Plan or any other employee benefit or compensation plan, program or arrangement sponsored, maintained or contributed to by the Asset Seller or to which the Asset Seller was obligated to contribute at any time on or prior to the Closing and (y) all Controlled Group Liabilities of the Asset Seller;
 - (iii) all Specified Excluded Liabilities;
 - (iv) all Liabilities arising out of or related to any Debt Obligations of the Asset Seller, including all such Debt Obligations between the Asset Seller, on the one hand, and the Company or any of its Affiliates, on the other hand;
 - (v) all Liabilities arising out of or related to the Asset Seller's breach of any Ancillary Agreements to which it is a party; and
 - (vi) all Transaction Expenses.
- 2.5 Purchase Price. On the Closing Date and subject to the terms and conditions set forth in this Agreement, in consideration of the sale, assignment and transfer of the Shares and the Sold Assets, the Buyers shall (a) pay (or cause to be paid) to the Company or its designee(s) an aggregate amount equal to \$135,000,000 (as adjusted, the "Closing Purchase Price"), by wire transfer of immediately available funds in U.S. dollars (or the currency of a foreign country if payment in U.S. dollars is not permitted by applicable Law in a foreign country where an Equity Seller is located) to an account of the Company (such account to be designated by the Company in a written notice delivered to the Buyers at least two Business Days prior to the Closing Date) in its capacity as agent for the Asset Seller and the Equity Sellers, and (b) assume the Assumed Liabilities, as applicable (collectively, the "Total Consideration"). The Closing Purchase Price shall be adjusted prior to Closing Date pursuant to Section 2.6(a). After the Closing Date, the Closing Purchase Price shall be adjusted pursuant to Section 2.6(b)-(e). The Closing Purchase Price, plus or minus

the adjustment amount as finally determined pursuant to Section 2.6(b)-(e), shall be the "Purchase Price." For purposes of Sections 2.5 and 2.6, any monetary conversion from the currency of a foreign country to U.S. dollars or from U.S. dollars to the currency of a foreign country shall be calculated using the applicable exchange rates set forth in *The Wall Street Journal*, Eastern Edition as of the applicable measurement date. For the avoidance of doubt, the Purchase Price and Total Consideration specified in this Section 2.5 do not include any Transfer Taxes incurred in connection with this Agreement and the transactions contemplated hereby. Such Transfer Taxes shall be paid in accordance with Section 5.4(d) of this Agreement in addition to the Purchase Price and Total Consideration specified in this Section 2.5.

2.6 Purchase Price Adjustment.

Pre-Closing Statement. No later than three Business Days prior to the Closing Date, the Company shall prepare and deliver to the Buyers a written statement together with any relevant supporting materials (the "Pre-Closing Statement") setting forth (i) a balance sheet of the Business as of the Closing (the "Estimated Balance Sheet") and (ii) the Company's calculation of a good faith estimate of (A) the Closing Working Capital as derived therefrom (the "Estimated Working Capital"), (B) the combined amount of Cash of the Divested Companies anticipated to exist immediately prior to the Closing (the "Estimated Cash"), (C) the combined amount of Debt Obligations of the Divested Companies anticipated to exist immediately prior to the Closing (the "Estimated Indebtedness"), (D) the combined amount of any Transaction Expenses incurred by any Divested Company and that are unpaid immediately prior to Closing or that are payable by the Buyers or any of their respective Affiliates (including the Divested Companies) at or after the Closing (the "Estimated Unpaid Expenses") and (E) the Closing Purchase Price, as adjusted as provided below to give effect to the Estimated Working Capital, the Estimated Cash, the Estimated Indebtedness and the Estimated Unpaid Expenses. The Estimated Balance Sheet (and the Estimated Working Capital derived therefrom) shall be prepared in accordance with the Accounting Methods and presented in a manner consistent with the example attached hereto as Exhibit F. If the Estimated Working Capital (as set forth in the Pre-Closing Statement) is less than \$62,000,000 (the "Target Working Capital") minus \$500,000, then the Closing Purchase Price shall be adjusted downward by an amount equal to the amount of the deficiency between the Target Working Capital minus \$500,000 and the Estimated Working Capital. If the Estimated Working Capital (as set forth in the Pre-Closing Statement) is greater than the Target Working Capital plus \$500,000, then the Closing Purchase Price shall be adjusted upward by an amount equal to the amount of the excess between the Estimated Working Capital and the Target Working Capital plus \$500,000. If the Estimated Working Capital is not less than the Target Working Capital minus \$500,000 and not greater than the Target Working Capital plus \$500,000, then no adjustment shall be made to the Closing Purchase Price with respect to the Estimated Working Capital. In addition, the Closing Purchase Price shall be adjusted upward by the amount of any Estimated Cash and adjusted downward by the amount of any Estimated Indebtedness and Estimated Unpaid Expenses. During the period between the delivery of the Pre-Closing Statement and the Closing, the Buyers shall have an opportunity to review and provide comments to the Pre-Closing Statement, and the Company shall cooperate with the Buyers in good faith with the objective that the Buyers and the Company shall mutually agree upon the Pre-Closing Statement if the Buyers dispute any item proposed to be set forth in the Pre-Closing Statement; provided, however, that if the Company and the Buyers are not able to reach a mutual agreement prior to the Closing Date, then the Pre-Closing Statement provided by the Company to the Buyers shall be binding for purposes of this Section 2.6(a).

- Post-Closing Statement. Within 60 days after the Closing Date, the Buyers shall cause to be prepared and delivered to the Company a statement (the "Post-Closing Statement") setting forth (i) a balance sheet of the Business as of the Closing (the "Closing Balance Sheet") and (ii) the Buyers' calculation of (A) the Closing Working Capital as derived therefrom ("Final Working Capital"), (B) the combined amount of Cash of the Divested Companies actually existing as of the Closing at 12:01 a.m. (local time for each Divested Company) on the Closing Date (the "Final Cash"), (C) the combined amount of Debt Obligations of the Divested Companies actually existing as of the Closing at 12:01 a.m. (local time for each Divested Company) on the Closing Date (the "Final Indebtedness"), (D) the combined amount of any Transaction Expenses incurred by any Divested Company and that are unpaid immediately prior to Closing or that are payable by the Buyers or any of their respective Affiliates (including the Divested Companies) at or after the Closing (the "Final Unpaid Expenses") and (E) the Closing Purchase Price as adjusted as provided below to give effect to the Final Working Capital, the Final Cash, the Final Indebtedness and the Final Unpaid Expenses. The Closing Balance Sheet (and the Final Working Capital derived therefrom) shall be prepared in accordance with the Accounting Methods and presented in a manner consistent with the example attached hereto as Exhibit F. The Company will, and will cause its Affiliates (including the Sellers) to, assist and cooperate with the Buyers in the preparation of the Post-Closing Statement. If the Final Working Capital (as set forth in the Post-Closing Statement) is less than the Target Working Capital minus \$500,000, then the Closing Purchase Price shall be adjusted downward by an amount equal to the amount of the deficiency between the Target Working Capital minus \$500,000 and the Final Working Capital. If the Final Working Capital (as set forth in the Post-Closing Statement) is greater than the Target Working Capital plus \$500,000, then the Closing Purchase Price shall be adjusted upward by an amount equal to the amount of the excess between the Final Working Capital and the Target Working Capital plus \$500,000. If the Final Working Capital is not less than the Target Working Capital minus \$500,000 and not greater than the Target Working Capital plus \$500,000, then no adjustment shall be made to the Closing Purchase Price with respect to the Final Working Capital. In addition, the Closing Purchase Price shall be adjusted upward by the amount of any Final Cash and adjusted downward by the amount of any Final Indebtedness and Final Unpaid Expenses.
- Dispute. Within 45 days following receipt by the Company of the Post-Closing Statement, the Company shall deliver written notice to the Buyers of any dispute it has with respect to the Post-Closing Statement (the "Post-Closing Statement Objection") setting forth a specific description of the basis of the Post-Closing Statement Objection, the adjustments to the Post-Closing Statement which the Company believes should be made, and the Company's calculation of the Final Working Capital, the Final Cash, the Final Indebtedness and the Closing Purchase Price as adjusted thereby, together with relevant supporting materials. The Company shall be deemed to have accepted any items not specifically disputed in the Post-Closing Statement Objection. Failure to so notify the Buyers within such 45-day period shall constitute acceptance and approval of the Buyers' calculation of the Final Working Capital, the Final Cash, the Final Indebtedness and the Closing Purchase Price set forth in the Post-Closing Statement. During such 45-day period, the Buyers shall, at the written request of the Company, on reasonable prior notice from the Company, during normal business hours and at the Company's sole cost and expense, afford the Company reasonable access to the books and records with respect to the Business (to the extent relevant to the determination of the Final Working Capital, the Final Cash and the Final Indebtedness) and otherwise reasonably cooperate with the Company in connection with its preparation of the Post-Closing Objection. The Buyers shall have 30 days following the date they receive the Post-Closing Statement Objection to review and respond to the Post-Closing Statement Objection. If the Company and the Buyers are unable to resolve all of their disagreements with

respect to the determination of the foregoing items by the 30th day following the Buyers' response thereto, after having used their good faith efforts to reach a resolution, they shall refer their remaining differences to Grant Thornton LLP or. if such firm refuses to accept such engagement, another nationally recognized firm of independent public accountants as to which the Company and the Buyers mutually agree acting promptly and in good faith (in either case, the "CPA Firm") to resolve their dispute. As promptly as practicable, and in any event not more than 15 days thereafter, the Company and the Buyers shall each prepare and submit a written presentation detailing each Party's complete statement of proposed resolution of the dispute to the CPA Firm. As soon as practicable thereafter, the Company and the Buyers shall use commercially reasonable efforts to cause the CPA Firm to resolve such dispute. The scope of the disputes to be resolved by the CPA Firm is limited to the unresolved items on the Post-Closing Statement Objection. In resolving any disputed item, the CPA Firm (i) shall not assign a value to any item greater than the greatest value claimed for such item by either the Company or the Buyers or lower than the lowest value claimed for such item by either the Company or the Buyers and (ii) shall make such determination with respect to the disputed items in accordance with the Accounting Methods, and, in each case, only with respect to the specific remaining accounting-related differences so submitted in such written presentations. All fees and expenses of the CPA Firm relating to the work, if any, to be performed by the CPA Firm hereunder shall be borne between the Buyers, on the one hand, and the Company, on the other hand, based upon a fraction, the numerator of which is the portion of the aggregate amount of the disputed items not awarded to the applicable Party and the denominator of which is the aggregate amount of the disputed items. For example, if the Company challenges items underlying the calculations of Final Working Capital, Final Indebtedness, Final Cash or Final Unpaid Expenses in the net amount of \$1,000,000, and the CPA Firm determines that the Company has a valid claim for \$400,000 of the \$1,000,000, the Company shall bear 60% of the fees and expenses of the CPA Firm and the Buyers shall bear 40% of the fees and expenses of the CPA Firm. The Buyers and the Company each agree to execute, if requested by the CPA Firm, a reasonable engagement letter. The Company and the Buyers shall request that the CPA Firm use its best efforts to render its determination within 45 days after referral. All determinations made by the CPA Firm will be limited to the matters submitted to the CPA Firm by the Buyers and the Company and shall be final, conclusive and binding on the Parties and none of the Buyer, the Company or any of their respective Affiliates shall seek further recourse from Governmental Authorities, other than to enforce the CPA Firm's determination. Judgment may be entered to enforce such determination in any court of competent jurisdiction. Notwithstanding the foregoing, if the disputed items relate to the interpretation of the Parties' legal rights or obligations under this Agreement or the other Ancillary Agreements, including any rights to indemnification under Article IX, rather than financial or accounting matters pertinent to the calculation of the amounts set forth in the Post-Closing Statement, such disputed items shall instead be resolved in the manner set forth in Section 10.14. The Company and the Buyers shall make reasonably available to the CPA Firm all relevant books and records, any work papers (including those of the Parties' respective accountants) and supporting documentation relating to the Post-Closing Statement and all other items reasonably requested by the CPA Firm. The Company and the Buyers shall be afforded the opportunity to discuss the disputed items with the CPA Firm so long as each Party is present. The Company and the Buyers shall give each other copies of any written submissions at the same time as they are submitted to the CPA Firm. The "Final Statement" shall be (A) the Post-Closing Statement in the event that (1) no Post-Closing Statement Objection is delivered to the Buyers during the initial 30-day period specified above or (2) the Company and the Buyers so agree in writing, (B) the Post-Closing Statement, adjusted in accordance with the Post-Closing Statement Objection, in the event that (1) the Buyers do not respond to the Post-Closing Statement Objection during the 30-day period specified above following receipt by the Buyers of the Post-Closing Statement Objection or (2) the Company and

the Buyers so agree in writing or (C) the Post-Closing Statement, as adjusted pursuant to the agreement of the Buyers and the Company or as determined by the CPA Firm together with any other modifications to the Post-Closing Statement agreed upon in writing by the Company and the Buyers prior to the determination by the CPA Firm. Any adjustment or non-adjustment to the Purchase Price shall not form the basis for any claim for damages pursuant to this Agreement. The Parties' payment obligations under this Section 2.6 will not be subject to offset or reduction by reason of any actual or alleged breach of, or inaccuracy in, any representation, warranty, covenant or agreement contained in this Agreement or the Ancillary Agreements, and any right or alleged right of indemnification hereunder or for any other reason.

- (d) Downward Adjustment. If the Closing Purchase Price set forth on the Final Statement is less than the Closing Purchase Price paid at Closing, then the Closing Purchase Price shall be adjusted downward by an amount equal to the amount of the deficiency between the Closing Purchase Price as set forth in the Final Statement and Closing Purchase Price paid at Closing, and the Company shall pay or cause to be paid such amount by wire transfer of immediately available funds in U.S. dollars to an account designated by the Buyers. Such payment shall be made within three Business Days after the later of the date on which the Final Statement is determined and the date when the Buyers or the Company, as applicable, designates to the Company or the Buyers, as applicable, the payment account details (such later date, the "Trigger Date").
- (e) Upward Adjustments. If the Closing Purchase Price set forth on the Final Statement is greater than the Closing Purchase Price paid at Closing, then the Closing Purchase Price shall be adjusted upward by an amount equal to the amount of the excess between the Closing Purchase Price set forth on the Final Statement and the Closing Purchase Price paid at Closing, and the Buyers shall pay or cause to be paid such amount by wire transfer of immediately available funds in U.S. dollars to an account of the Company (such account to be designated in writing by the Company to the Buyers prior to Closing), in its capacity as agent for the Asset Seller and the Equity Sellers. Such payment shall be made within three Business Days after the Trigger Date.
- Allocation of Total Consideration. The Parties shall cooperate in good faith to determine (a) an 2.7 allocation of the Total Consideration attributable to the Sold Assets and the Divested Companies and (b) an allocation of the Total Consideration attributable to the Sold Assets among the Sold Assets (the "Allocation"). The Buyers shall deliver a proposed Allocation to the Company no later than 60 days after the Final Statement is determined pursuant to Section 2.6(c) and the Company will provide any comments, questions or objections with respect thereto no later than 30 days after the delivery of the Allocation by the Buyers. The Allocation shall be prepared in accordance with the fair market value of such Sold Assets and Section 1060 of the Code and, to the extent not inconsistent therewith, any other applicable Tax Law. If the Parties are unable to resolve any disagreements with respect to the Allocation by the 30th day following the receipt by the Buyers of the Company's comments, questions or objections thereto, after having used their good faith efforts to reach a resolution, they shall refer their remaining differences to the CPA Firm and any disagreements shall be resolved in accordance with the procedures provided for in Section 2.6(c). The Parties shall cooperate to comply with all substantive and procedural requirements of Section 1060 of the Code and any regulations thereunder and, to the extent not inconsistent therewith, any other applicable Tax Law, and the Allocation shall be adjusted or supplemented if, and to the extent, necessary to comply with the requirements of Section 1060 of the Code and, to the extent not inconsistent therewith, any other applicable Tax Law. Each of the Company and the Buyers agree that they shall attach to their respective Tax Returns for the tax year in which the Closing shall occur

an information statement on Form 8594, which shall be completed in accordance with the Allocation, and shall cause their respective Affiliates to, cooperate with each other in preparing and filing any supplements to such form as may be required under Section 1060 of the Code and any other applicable Tax Law. The Parties agree to promptly notify each other in writing with respect to the initiation of any inquiry, claim, assessment, audit or Proceeding by any Taxing Authority relating to the Allocation and agree to consult with each other with respect to any such inquiry, claim, assessment, audit or Proceeding by any Taxing Authority.

- 2.8 The Closing. Unless this Agreement shall have been terminated pursuant to Article VIII, and subject to Article VI and Article VII, the closing (the "Closing") of the transactions contemplated by this Agreement shall take place remotely via the electronic exchange of documents and signatures, on the third Business Day following the satisfaction or waiver (to the extent legally permitted) of all of the conditions set forth in Article VI and Article VII (other than those conditions that are to be satisfied at the Closing), or at such other place and time as may be agreed upon by the Parties; provided, however, that the Closing shall not occur prior to the date that is 60 days following the Execution Date unless otherwise agreed by the Parties in writing (such date on which the Closing occurs, the "Closing Date"). All proceedings to be taken and all documents to be executed and delivered by all Parties at the Closing shall be deemed to have been taken and executed and delivered simultaneously and no proceedings shall be deemed to have been taken nor documents executed or delivered until all have been taken, executed and delivered. Legal title, equitable title and risk of loss with respect to the Shares and the Sold Assets will be deemed transferred to or vested in the Buyers (as applicable), and the transactions contemplated by this Agreement will be deemed effective for Tax, accounting and other computational purposes, and the Parties will treat the Closing as if it had occurred, as of 12:01 a.m. (Eastern Time) on the Closing Date.
 - 2.9 Deliveries at the Closing.
- (a) Deliveries by the Company. At or prior to the Closing, the Company shall deliver or cause to be delivered to the Buyers the following:
 - (i) the items set forth on Annex A;
 - (ii) a certificate, dated as of the Closing Date and executed by an officer of the Company, certifying to the fulfillment of the conditions specified in Sections 7.1 and 7.2;
 - (iii) the Ancillary Agreements to which any of the Company or the Sellers or any of their respective Affiliates is a party, duly executed by the Company or such Sellers or such respective Affiliates, as applicable;
 - (iv) a certificate, to the extent applicable, from each of the Asset Seller and R-L International Holdings certifying, pursuant to Treasury Regulations Section 1.1445-2(b)(2), it is not a foreign person within the meaning of Section 1445 of the Code; and
 - (v) evidence, in a form reasonably satisfactory to the Buyers, of the release of all Encumbrances on the Sold Assets (other than Permitted Encumbrances), the assets of the Divested Companies and the Shares (other than other than Encumbrances arising from restrictions on sale, assignment or transfer of shares of capital stock under applicable Law).

- (b) Deliveries by the Buyers. At or prior to the Closing, the Buyers shall deliver or cause to be delivered to the Company the following:
 - (i) the items set forth on Annex B;
 - (ii) by wire transfer of immediately available funds in U.S. dollars to an account of the Company (such account to be designated in writing by the Company to the Buyers prior to Closing), in its capacity as agent for the Asset Seller and the Equity Sellers, an amount equal to the Closing Purchase Price (as adjusted pursuant to Section 2.6(a));
 - (iii) a certificate, dated as of the Closing Date and executed by an officer of each Buyer, certifying to the fulfillment of the conditions specified in Sections 6.1 and 6.2; and
 - (iv) the Ancillary Agreements to which the Buyers or any of their respective Affiliates or designees is a party, duly executed by the Buyers or such Affiliates or designees.
- 2.10 Further Assurances. Each Party covenants that it will do, execute and deliver, or will cause to be done, executed and delivered, all such further acts and instruments that the other Parties or any of their respective successors or permitted assigns may reasonably request in order to give effect to the transactions contemplated by this Agreement and the Ancillary Agreements, including to more fully evidence the assumption of the Assumed Liabilities provided for in Section 2.4 and the sale and transfer of the Sold Assets and the Shares, in each case, upon the terms and conditions contained in this Agreement and the Ancillary Agreements, as applicable.
- 2.11 Withholding. The Buyers and their respective Affiliates shall be entitled to deduct and withhold from any amounts otherwise payable or deliverable to the Company or any of its Affiliates (and the Company shall, and shall cause its Affiliates to, indemnify, defend and hold harmless the Buyers and their respective Affiliates against) such amounts as may be required to be deducted or withheld therefrom under applicable Law; provided, however, if the Buyers determine that an amount is required to be deducted and withheld with respect to any amounts payable (other than as compensation), at least five days prior to the date the applicable payment is scheduled to be made, the Buyers shall provide the Sellers with written notice of their intent to deduct and withhold, which notice shall include a copy of the calculation of the amount to be deducted and withheld and a reference to the applicable provision of Law pursuant to which such deduction and withholding is required, and the Buyers shall reasonably cooperate with the Sellers to eliminate or reduce the basis for such deduction or withholding (including providing the Sellers with a reasonable opportunity to provide forms or other evidence that would exempt such amounts from withholding). To the extent such amounts are so deducted or withheld and timely paid to the appropriate Taxing Authority, such amounts shall be treated for all purposes as having been paid to the Person to whom such amounts would otherwise have been paid absent such deduction or withholding.

ARTICLE III REPRESENTATIONS AND WARRANTIES OF THE COMPANY

Except as set forth in the Disclosure Schedules, the Company hereby represents and warrants to the Buyers as of the Execution Date and the Closing Date (except to the extent a specific date is referenced, in which case the Company represents and warrants to the Buyers as of such date) as follows:

3.1 Organization.

- (a) Each of the Company, the Sellers and the Divested Companies is a limited liability company, corporation or company, as applicable, duly incorporated, formed or organized, as applicable, validly existing and (to the extent any such jurisdiction recognizes the concept of good standing) in good standing under the Laws of its jurisdiction of incorporation, formation or organization, as applicable. Each of the Company, the Sellers and the Divested Companies has all requisite power and authority to own, lease and operate its assets and to carry on its business as now being conducted and is duly qualified or licensed to do business and (to the extent any such jurisdiction recognizes the concept of good standing) is in good standing in the jurisdictions (whether federal, state, local or foreign) in which the ownership of its property or the conduct of its business requires such qualification or license, except where the failure to be so qualified or licensed has not had, or would not reasonably be expected to have, individually or in the aggregate, a Business Material Adverse Effect or a Seller Material Adverse Effect.
- (b) True, correct and complete copies of the organizational documents of each Divested Company, together with any amendments thereto, have been made available to the Buyers.
- authority to execute and deliver this Agreement and each Ancillary Agreement to which the Company or the Sellers are or will be a party and to perform their respective obligations hereunder and thereunder. The execution and delivery of this Agreement and the Ancillary Agreements by each of the Company and the Sellers, as applicable, and the performance by each of them of their respective obligations hereunder and thereunder, have been duly authorized by all necessary action on the part of the Company, the Sellers and the Divested Companies, as applicable. This Agreement has been, and each of the Ancillary Agreements to which the Company or any Seller is or will be a party will be, at or prior to the Closing, duly executed and delivered by each of the Company and the Sellers party thereto and, assuming due authorization, execution and delivery by the Buyers, this Agreement constitutes, and each of the Ancillary Agreements will constitute, a valid and binding agreement of each of the Company and the Sellers party thereto, enforceable against each of the Company and the Sellers to the extent party thereto in accordance with its and their respective terms, subject to the effects of bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other similar Laws relating to or affecting creditors' rights generally and general equitable principles (whether considered in a proceeding in equity or at Law) (collectively, the "General Enforceability Exceptions").
- 3.3 Capital Stock of the Divested Companies. Schedule 3.3 sets forth for each of the Divested Companies (a) its jurisdiction of organization and (b) the number of authorized, issued and outstanding Equity Interests, the names of the holders thereof, and the number of Equity Interests held by each such holder. The Equity Interests set forth on Schedule 3.3 collectively constitute all of the issued and outstanding Equity Interests of each of the Divested Companies are owned of record by the holders listed on Schedule 3.3, free and clear of any Encumbrances, other than Encumbrances arising from restrictions on sale, assignment or transfer of shares of capital stock under applicable Law. All of the issued and outstanding Equity Interests of each of the Divested Companies have been validly issued, are fully paid and non-assessable and have not been issued in violation of any preemptive or similar rights. There are no outstanding Options, equity appreciation, phantom stock, profit participation, preemptive rights, registration rights, approval rights, proxies,

rights of first refusal or other agreements relating to the sale, issuance or voting of any Equity Interests of any Divested Company, or any securities or other instruments convertible into, exchangeable for or evidencing the right to purchase any Equity Interests of any Divested Company. Except as set forth on Schedule 3.3, none of the Divested Companies, nor the Asset Seller, owns or has owned any Equity Interest in any other Person.

- 3.4 Financial Information; No Undisclosed Liabilities; Solvency.
- (a) Attached as Schedule 3.4(a) are (i) the unaudited balance sheets of the Business, as of December 31, 2018, 2017 and 2016, and the unaudited statements of income of the Business for the years then ended (collectively, the "Annual Financial Information") and (ii) the unaudited balance sheet of the Business as of June 30, 2019 (such balance sheet, the "Balance Sheet" and such date, the "Balance Sheet Date") and the unaudited statements of income of the Business for the six-month period then ended (together with the Balance Sheet, the "Interim Financial Information," and together with the Annual Financial Information, the "Financial Information"). The Financial Information was derived from the books and records related to the Business, has been prepared in accordance GAAP applied on a consistent basis throughout the periods covered thereby (except as might be indicated in the notes thereto, if any) and fairly presents in all material respects the financial condition of the Business as of the dates thereof and the results of operations of the Business as of the indicated dates and for the indicated periods, subject, in the case of the Interim Financial Information, deviations from GAAP consistent with normal, immaterial year-end adjustments and the absence of notes required by GAAP.
- (b) The Asset Seller and the Divested Companies do not have any Liabilities, other than (i) Liabilities that are reflected in the Financial Information, (ii) Liabilities set forth on Schedule 3.4(b), (iii) the Excluded Asset Seller Liabilities, (iv) current Liabilities or obligations of the Business arising since the Balance Sheet Date in the ordinary course of business consistent with past practice (none of which results from, arises out of, relates to, is in the nature of or was caused by any breach of contract, breach of warranty, tort, infringement or violation of applicable Law) and (v) other Liabilities that are not, and would not reasonably be expected to be, individually or in the aggregate, material to the Business.
- (c) Except as set forth on Schedule 3.4(c), as of the Execution Date, none of the Divested Companies have any outstanding Debt Obligations.
 - 3.5 Sufficiency of the Assets.
- (a) Except for any assets or services to be made available to the Buyers pursuant to the Transition Services Agreement, taking into account the corporate-level services provided to the Business that are set forth on Schedule 3.5(a), and assuming receipt of all relevant consents, approvals and authorizations relating to the matters set forth in Schedule 3.6, the Sold Assets transferred to the Buyers at Closing and the assets owned, leased or licensed by the Divested Companies collectively constitute all of the properties, assets and rights (including Intellectual Property) used in, or held for use in, the Business by any Affiliate of the Company and constitute all of the rights, property, and assets (including Intellectual Property) necessary to conduct the Business immediately following the Closing in substantially the same manner as the Business is currently conducted on the Execution Date.
- (b) Except as set forth on Schedule 3.5(b), the Sold Assets and the assets owned by the Divested Companies are, in all material respects, in good maintenance, operating condition

and repair, normal wear and tear excepted, other than machinery and equipment under repair or out of service in the ordinary course of the Business.

- (c) Except as set forth on Schedule 3.5(c), the Divested Companies do not own any assets (and are not subject to any Liabilities) that are unrelated to the Business or are not otherwise used or held for use in, or necessary for the conduct of, the Business.
- No Approvals or Conflicts. The execution, delivery and performance by the Company of this Agreement and by the Company or any its Affiliates of the Ancillary Agreements to which they are or will be a party and the consummation by the Company and the Sellers of the transactions contemplated hereby and thereby do not and will not (a) violate, conflict with or result in a breach by the Company, any Seller or Divested Company of its organizational documents, (b) violate, conflict with or result in a breach of, or constitute a default by the Company, any Seller or Divested Company (or create an event which, with notice or lapse of time or both, would constitute a default) or give rise to any payment or other penalty or any right of termination, cancellation or acceleration under, or result in the creation of any Encumbrance upon any of the properties or assets of the Asset Seller, any Divested Company or on the Shares or the Sold Assets under, any note, bond, mortgage, indenture, deed of trust, Permit, lease or Contract to which the Asset Seller, any Divested Company or any of their respective properties may be bound, (c) violate or result in a breach of any Governmental Order or Law applicable to the Company, any Seller or Divested Company or any of their respective properties or (d) except for (i) such approvals, consents, filings, orders, authorizations and notices as have been obtained or made as of the Execution Date, (ii) the applicable requirements of the HSR Act and the requirements of any other applicable antitrust or competition Law, (iii) filings for payment of Duty required under applicable Law, and (iv) filings or approvals that may be required under the Exchange Act, ERISA, the Investment Canada Act or any other applicable Law, require any order, consent, approval, filing with, notice to, or authorization of any Person (including any Governmental Authority), except in the cases of clauses (b), (c) and (d) as has not had, or would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on the ability of the Company and its Affiliates to perform their obligations under this Agreement (including to consummate the transactions contemplated hereby) or otherwise be material to the Business.
- 3.7 Compliance with Law; Permits. Except as set forth on Schedule 3.7, since January 1, 2016, the Business has been conducted, the Sold Assets and the assets of the Divested Companies have been maintained and the Divested Companies and the Asset Seller are, in material compliance with all applicable Laws. Since January 1, 2016, none of the Company or any of its Affiliates has received any notice from any Governmental Authority alleging that the Business (or the Company or any of its Affiliates with respect thereto) is or has been in violation of any applicable Law in any material respects. Except as set forth on Schedule 3.7, the Divested Companies possess all of the Permits necessary to own, lease and operate their respective assets and conduct the Business conducted by them in all material respects as currently owned, leased, operated and conducted, and the Asset Seller possesses all of the Permits necessary to own, lease and operate the Sold Assets and the assets of the Divested Companies and conduct the Business conducted by it in the same manner in all material respects as currently owned, leased, operated and conducted. All such Permits are in full force and effect, and the Business has been conducted in accordance in all material respects with the requirements of such Permits.
- 3.8 Proceedings. Except as set forth on Schedule 1.1(i), there are no, and since January 1, 2016 there have been no, Proceedings pending or, to the Knowledge of the Company,

threatened against the Business or the Company or any of its Affiliates that are or would be material to the Business. There are no Proceedings pending or, to the Knowledge of the Company, threatened against the Company or any of its Affiliates challenging or seeking to enjoin the transactions contemplated by this Agreement or any Ancillary Agreements. None of the Company or any of its Affiliates are subject to any Governmental Order that is or would be material to the Business.

- 3.9 Absence of Certain Changes. During the period from the Balance Sheet Date through the Execution Date, except for the process conducted that gave rise to this Agreement and the transactions contemplated by this Agreement, (a) the Business has been conducted only in the ordinary course consistent in all material respects with past practice and (b) none of the Sellers or the Divested Companies has taken any action that, if Section 5.1 had applied in such period, would have constituted a breach thereof. Since the Balance Sheet Date, there has not been a Business Material Adverse Effect.
 - 3.10 Tax Matters. Except as set forth on Schedule 3.10:
- (a) All Asset Taxes with respect to the Sold Assets or that relate to the Business that have become due and payable have been duly and timely paid in full, and all Tax Returns with respect to such Asset Taxes required to be filed have been duly and timely filed.
- (b) All Tax Returns required to be filed by or on behalf of or with respect to any Divested Company or any subsidiary of a Divested Company have been duly and timely filed (subject to permitted extensions applicable to such filings), and such Tax Returns are true, correct and complete in all material respects. All Taxes owed by any Divested Company or any subsidiary of a Divested Company or for which any Divested Company or any subsidiary of a Divested Company may be liable that have become due have been paid in full within the prescribed period or any extension thereof (other than Taxes that are being contested in good faith for which adequate accruals or reserves have been established in accordance with GAAP). Each Divested Company and any subsidiary of a Divested Company has withheld and timely paid over to the appropriate Taxing Authority all Taxes which it is required to withhold from amounts paid or owing to any Person, including any employee, shareholder, creditor, holder of securities, non-resident Person or other third party.
- (c) There are no Encumbrances relating to Taxes encumbering any of the Shares, the Sold Assets or any assets of any Divested Company or any subsidiary of a Divested Company, except for Permitted Encumbrances.
- (d) There is no written claim against the Divested Companies, any subsidiary of a Divested Company or with respect to the Sold Assets for any Taxes, and no assessment, deficiency or adjustment has been asserted in writing, proposed in writing or, to the Knowledge of the Company, threatened in writing with respect to any Taxes or Tax Returns of or with respect to the Divested Companies, any subsidiary of a Divested Company or the Sold Assets. There are no examinations, audits, actions, proceedings, investigations or disputes being conducted, pending or, to the Knowledge of the Company, threatened in writing with respect to any Divested Company, any subsidiary of a Divested Company or the Sold Assets. No claim has ever been made by a Governmental Authority in a jurisdiction where the Divested Companies or any subsidiary of a Divested Company, or the Asset Seller solely with respect to the Sold Assets, does not file Tax Returns that a Divested Company, any subsidiary of a Divested Company, the Asset Seller or any of the Sold Assets is or may be subject to taxation in that jurisdiction.

- (e) There is not in force any waiver of any statute of limitations in respect of Taxes or any extension of time with respect to the assessment or payment of any Tax of or with respect to the Divested Companies, any subsidiary of a Divested Company or the Sold Assets. There is not in force any extension of time with respect to the due date for the filing of any Tax Return of or with respect to the Divested Companies, any subsidiary of a Divested Company or the Sold Assets.
- (f) No Divested Company is a party to or bound by any Tax allocation, sharing or indemnity Contracts or arrangements (excluding, for the avoidance of doubt, any Commercial Tax Agreements). No Divested Company has any liability for the Taxes of any Person under Treasury Regulations Section 1.1502-6 (or any similar provision of state, local or foreign Law), or as a transferee or successor, or by Contract other than any Commercial Tax Agreement. No Divested Company has ever been a member of a Consolidated Group.
- (g) None of the properties of the Divested Companies, any subsidiary of the Divested Companies or Sold Assets are subject to any Tax partnership Contract or are otherwise treated, or required to be treated, as held in an arrangement requiring a partnership income Tax Return to be filed under Subchapter K of Chapter 1 of Subtitle A of the Code.
- (h) No Divested Company or any subsidiary of a Divested Company has engaged in, is currently engaging in or has been a party to a transaction that constitutes a "listed transaction," as defined in Treasury Regulations Section 1.6011-4(b)(2).
- (i) Each Divested Company and each subsidiary of a Divested Company is, and has been since its formation, properly classified for U.S. federal income Tax purposes as a foreign corporation. For U.S. federal income tax purposes, the taxable year of each Divested Company ends on November 30th.
- (j) No Divested Company or any subsidiary of a Divested Company is, or has ever been, (i) a "surrogate foreign corporation" within the meaning of Section 7874(a)(2)(B) of the Code, (ii) treated as a "domestic corporation" under Section 7874(b) of the Code or (iii) a "passive foreign investment company" within the meaning of Section 1297 of the Code.
- (k) No Divested Company or any subsidiary of a Divested Company is subject to Tax in any jurisdiction, other than the country in which it is organized, by virtue of having, or being deemed to have, engaged in a trade or business, a permanent establishment or fixed place of business. All payments by, to or among the Divested Companies and their Affiliates comply with all applicable transfer pricing requirements (including documentary, retention and filing requirements) imposed by any Governmental Authority, and the Company has made available to the Buyers accurate and complete copies of all transfer pricing documentation prepared pursuant to Treasury Regulations Section 1.6662-6 (or any similar foreign statutory, regulatory or administrative provision) by or with respect to the Divested Companies or the subsidiaries thereof during the three years prior to the Execution Date.
- (I) Each Divested Company and each subsidiary of a Divested Company is in full compliance with all terms and conditions of any Tax exemption, Tax holiday or other Tax reduction Contract or order of any Taxing Authority, and the consummation of the transactions contemplated by this Agreement and the Ancillary Agreements will not have any adverse effect on the continued validity or effectiveness of such Tax exemption, Tax holiday or other Tax reduction Contract or order.

- (m) Neither any Divested Company nor any subsidiary of a Divested Company is a party to a gain recognition agreement under Section 367 of the Code.
- (n) None of the Shares are taxable Canadian property for purposes of section 116, and within the meaning of, the Income Tax Act (Canada).
- (o) None of the Sold Assets are (i) taxable Canadian property, (ii) Canadian resource property, (iii) real property, or immovable property, situated in Canada, (iv) timber resource property, (v) life insurance policy in Canada, or (vi) any interest, right or option in, or in respect of, any of the properties listed in (i) to (v) for purposes of section 116, and within the meaning of, the Income Tax Act (Canada).
- (p) For purposes of the Income Tax Act (Canada) or any other applicable Tax statute, no Person or group of Persons has ever acquired or had the right to acquire control of Lamons Canada or any of its subsidiaries.
- (q) None of sections 78, 80, 80.01, 80.02, 80.03 or 80.04 of the Income Tax Act (Canada), or any equivalent provision of the Tax legislation of any province or any other jurisdiction, have applied or will apply to Lamons Canada or any of its subsidiaries at any time up to and including the Closing Date.
- (r) None of Lamons Canada or any of its subsidiaries has acquired property from a non-arm's length Person, within the meaning of the Income Tax Act (Canada), for consideration, the value of which is less than the fair market value of the property acquired in circumstances which could subject it to a liability under section 160 of the Income Tax Act (Canada).
- (s) Lamons Canada is duly registered under subdivision (d) of Division V of Part IX of the Excise Tax Act (Canada) with respect to the goods and services tax and harmonized sales tax and the registration number of Lamons Canada is: 10076 4885 RT0001.

3.11 Employee Benefits.

- (a) Schedule 3.11(a) sets forth a true and correct list of (i) each "employee benefit plan" (within the meaning of Section 3(3) of ERISA, whether or not subject to ERISA), (ii) all other material severance, salary continuation, change in control, employment, consulting, personnel, vacation, incentive, bonus, Options, equity appreciation rights, restricted equity, phantom equity, equity-based incentive or supplemental income, retirement, pension, supplemental pension, redundancy, profit sharing or deferred compensation plans, programs, agreements or policies and (iii) all other employee benefit plans or programs, in each case, in which Active Employees or former employees of a Divested Company participate (other than any such plans, programs, agreements or policies required by Law to be provided to any such employees, including the Canada Pension Plan, workers' compensation or similar benefits and other than any immaterial fringe benefit programs) and (A) are sponsored or maintained by the Asset Seller or the Divested Companies or (B) with respect to which the Divested Companies or the Asset Seller have made or are required to make payments, transfers or contributions on behalf of Active Employees or otherwise could have any Liability (collectively, the "Seller Benefit Plans").
- (b) The Seller Benefit Plans are in compliance, in all material respects, with their terms and applicable requirements of ERISA, the Code and other Laws, to the extent applicable. The Asset Seller and the Divested Companies have performed all obligations required to be

performed by them under any Seller Benefit Plan and to the Knowledge of the Company, are not in default under or in violation of any Seller Benefit Plan. With respect to each Seller Benefit Plan that is intended to qualify under Section 401(a) of the Code, as of the Execution Date, such Seller Benefit Plan has received a favorable determination letter or such Seller Benefit Plan may rely upon an opinion letter from the Internal Revenue Service with respect to its qualification and to the Knowledge of the Company, no fact or event has occurred since the date of such letter that would reasonably be expected to adversely affect such qualification.

- (c) There are no pending or, to the Knowledge of the Company, threatened Proceedings with respect to any Seller Benefit Plans (other than routine claims for benefits by participants and beneficiaries) that, in either case, would reasonably be expected to result in material Liability to the Buyer, a Divested Company or any of their respective Affiliates.
- (d) None of the Asset Seller, the Divested Companies or any of their respective ERISA Affiliates has any Liability with respect to any employee benefit plan that is subject to Section 412 of the Code or Section 302 or Title IV of ERISA, in each case, that would reasonably be expected to result in any material Liability to the Buyer, the Divested Companies or any of their respective Affiliates after the Closing. There does not now exist, nor do any circumstances exist that could result in, any Controlled Group Liability of the Asset Seller, the Divested Companies or any of their respective ERISA Affiliates that would be, or could become, a liability following the Closing of the Buyers, the Divested Companies or any of their respective Affiliates.
- (e) In connection with the consummation of the transactions contemplated by this Agreement, no payments of money or property, acceleration of benefits, or provisions of other rights have or will be made hereunder or under the Seller Benefit Plans that, in the aggregate, would be reasonably likely to result in imposition of the sanctions imposed under Sections 280G and 4999 of the Code (determined without regard to the exception contained in Section 280G(b)(4) of the Code), whether or not some other subsequent action or event would be required to cause such payment, acceleration or provision to be triggered.
- (f) Except to the extent required pursuant to Section 4980B(f) of the Code and the corresponding provisions of ERISA or other applicable Laws, no Seller Benefit Plan provides retiree medical or retiree life insurance benefits to any Active Employee or former employee of a Divested Company, and neither the Asset Seller nor any Divested Company is contractually obligated (whether or not in writing) to provide any Active Employee with, and no Divested Company is contractually obligated to, or has ever represented that it will, provide any Person with, life insurance or medical benefits upon retirement or termination of employment.
- (g) No Active Employees or former employees of Lamons Canada participate in, or have ever participated in, the defined benefit component of the Canadian Pension Plan and Lamons Canada has no Liabilities thereunder. Except for the Canadian Pension Plan, no Seller Benefit Plan is a "registered pension plan" as such term is defined in subsection 248(1) of the Income Tax Act (Canada).
- 3.12 Active Employees; Labor Relations. The Active Employees represent the entirety of the employees of Asset Seller and its Affiliates whose job duties principally involve providing services with respect to the Business. Except as set forth on Schedule 3.12, none of the Asset Seller or the Divested Companies or any entity that employs any Active Employee is a party to or bound by any collective bargaining agreement, collective agreement, or other Contract, agreement or understanding with any labor union, trade union, works council, or other representative

of any Active Employees or any other present or former employees of the Asset Seller or the Divested Companies, nor is any such Contract, agreement or understanding presently being negotiated. There is no unfair labor practice charge or complaint pending or, to the Knowledge of the Company, threatened against the Asset Seller or any of its Affiliates or any of the Divested Companies. There is no labor strike, slowdown, work stoppage, picketing or lockout in effect, or, to the Knowledge of the Company, threatened against the Asset Seller, any of its Affiliates or any of the Divested Companies. There is no, and since January 1, 2016 there has been no, labor union certification or representation petitions or demands with respect to any Active Employees or other employee of a Divested Company whose employment was not already subject to the terms of a collective bargaining agreement or collective bargaining relationship and, to the Knowledge of the Company, no such labor union organizing campaign or similar effort is pending or threatened.

3.13 Intellectual Property.

- (a) Schedule 3.13 sets forth a complete and accurate list of all of the registered forms of the Acquired Intellectual Property and applications therefor, and all material unregistered Acquired Intellectual Property (other than Acquired Intellectual Property constituting Intellectual Property pursuant to clauses (d) and (e) of the definition of Intellectual Property). The Divested Companies or the Asset Seller are the sole owners of all right, title and interest in and to all of the Acquired Intellectual Property, free and clear of all Encumbrances, exclusive licenses granted to third parties, and non-exclusive licenses not granted in the ordinary course of business, and other encumbrances, other than Permitted Encumbrances and Encumbrances under the Credit Agreement. The registered Acquired Intellectual Property used in the Business as currently conducted is valid, subsisting, and enforceable. All governmental fees associated with the registered forms of the Acquired Intellectual Property and due as of the Execution Date have been paid in full, and all necessary documents and certificates in connection with such Acquired Intellectual Property have been filed with the relevant patent, copyright, trademark or other authorities in the United States or foreign jurisdictions, as the case may be, for the purposes of maintaining such Acquired Intellectual Property. Each item of Acquired Intellectual Property used in the Business as currently conducted will continue to be owned by or licensed to the Asset Buyer or applicable Divested Company, on identical terms and conditions immediately following the consummation of the transactions contemplated by this Agreement, as are in effect immediately prior to such consummation.
- (b) Except as set forth on Schedule 3.13(b), neither the Divested Companies nor the Asset Seller have, other than through shrink wrap or click wrap software licenses, licensed any material Intellectual Property from any Person, nor have the Divested Companies or the Asset Seller granted any license or other right that does or that will, subsequent to the Closing, permit or enable anyone other than the Buyers (or their respective designee(s)) to use any of the Acquired Intellectual Property. No current or former director, officer or employee of the Company or any of its Affiliates will, after giving effect to each of the transactions contemplated by this Agreement, own or retain any ownership rights in or to, or have the right to receive any royalty or other payment with respect to, any Acquired Intellectual Property.
- (c) There are no pending suits, actions, claims, proceedings or investigations alleging that the Company or any of its Affiliates is infringing, misappropriating, diluting or otherwise violating any Intellectual Property of a Person or that seek to limit or challenge the validity, enforceability, ownership or use of the Acquired Intellectual Property. Since January 1, 2016, none of the Company or any of its Affiliates have received any written claim, "cease and desist" letter, or

like notice of any claim and, to the Knowledge of the Company, there is no threatened claim, against the Company or any of its Affiliates asserting that any of the Acquired Intellectual Property infringes upon or otherwise conflicts with the Intellectual Property of any Person, nor have the Company or any of its Affiliates given any written notice to any Person asserting infringement by such Person of any of the Acquired Intellectual Property. To the Knowledge of the Company, none of the Divested Companies nor the Asset Seller, nor the Business as currently or previously conducted, has infringed, misappropriated, diluted or otherwise violated the Intellectual Property of any Person. To the Knowledge of the Company, no Person is engaging in any activity or business that infringes upon, dilutes, or otherwise violates the Acquired Intellectual Property or misappropriates the trade secrets or confidential information of the Divested Companies or the Asset Seller.

- (d) All employees, contractors and agents of the Company and its Affiliates involved in the conception, development, authoring, creation or reduction to practice of any material Acquired Intellectual Property have executed agreements that assign such Intellectual Property to the Divested Companies or the Asset Seller and confidentiality agreements, and the Company has provided copies of the forms of such agreements to the Buyers. The Company and its Affiliates have taken reasonable measures to protect the confidentiality of trade secrets and confidential information used in the Business.
- Except for IT Systems to be provided by TriMas Corporation under the Transition Services (e) Agreement, the Divested Companies or the Asset Seller own, lease or license IT Systems of a sufficient quantity and size to operate the Business as currently conducted. The Divested Companies or the Asset Seller have taken commercially reasonable steps to provide for the back-up and recovery of data and information, have commercially reasonable disaster recovery plans, procedures and facilities, and, as applicable, have taken commercially reasonable steps to implement such plans and procedures. To the Knowledge of the Company, the computer systems do not contain any "back door," "time bomb," "Trojan horse," "worm," "drop dead device," "virus" (as these terms are commonly used in the computer software industry), or other software routines or hardware components intentionally designed to permit (i) unauthorized access to a computer or network, (ii) unauthorized disablement or erasure of software, hardware or data, or (iii) any other similar type of unauthorized activities. There has been no failure, material substandard performance or breach of any IT Systems that have caused any material disruption to the Business or resulted in any unauthorized disclosure of or access to any data owned, collected or controlled by the Divested Companies or the Asset Seller. The Divested Companies and the Asset Seller have taken reasonable technical, administrative, and physical measures to protect the integrity and security of the IT Systems and the data stored thereon from unauthorized use, access, or modification by third parties.
- (f) The Divested Companies and the Asset Seller have complied with all contracts, standards, privacy policies, and Laws applicable to the Divested Companies and the Asset Seller regarding the collection, use, disclosure or retention of personally identifiable information, including any such data privacy laws, industry security standards, or agreements with third parties, in every jurisdiction where (i) the Divested Companies or the Asset Seller operate or (ii) residents of such jurisdiction have provided personally identifiable information to the Divested Companies or the Asset Seller. The Divested Companies and the Asset Seller have commercially reasonable physical, technical, organizational and administrative security measures and policies in place to protect all personally identifiable information collected or possessed by it or on its behalf from and against unauthorized access, use or disclosure. Neither the Divested Companies nor the Asset Seller have

provided or been legally required to provide any notices to data owners in connection with any unauthorized access, use or disclosure of personally identifiable information.

3.14 Contracts.

- (a) Schedule 3.14(a) sets forth a true, correct and complete list of all of the following Contracts to which a Divested Company or the Asset Seller is a party, by which any of them is bound or which otherwise binds any Sold Asset or asset of the Divested Companies, in each case, as of the Execution Date, other than Seller Benefit Plans (the Contracts required to be set forth on Schedule 3.14(a), collectively, the "Material Contracts"):
 - (i) Contracts that can reasonably be expected to result in aggregate expenditures by or on behalf of any of the Divested Companies or the Asset Seller of more than \$500,000 in any instance for the purchase or lease of materials, supplies, equipment or services for the 2019 calendar year or any future calendar year or that otherwise involves a remaining commitment to pay capital expenditures greater than \$500,000 for the 2019 calendar year or any future calendar year;
 - (ii) Contracts that can reasonably be expected to result in aggregate revenues for any Divested Company or the Asset Seller of more than \$500,000 for the 2019 calendar year or any future calendar year prior to the expiration of such Contract;
 - (iii) indentures, mortgages, loan agreements, capital leases, security agreements or other Contracts for the borrowing of money or other Debt Obligation;
 - (iv) guarantees of any obligations of other Persons involving the potential expenditure by any of the Divested Companies or the Asset Seller after the Execution Date;
 - (v) Contracts that restrict the Asset Seller or any Divested Company after the Execution Date from engaging in any line of business in any geographic area or competing with any Person that materially impairs the operation of the Business or granting to any Person a right of first refusal, first offer or right to purchase, any of the Assets owned by the Asset Seller or any Divested Company;
 - (vi) Contracts under which (A) a Divested Company has licensed Intellectual Property that is material to the Business or involves consideration in excess of \$25,000 per annum to or from any other Person (including Affiliates of the Company), other than license agreements for unmodified "off-the-shelf" software on generally standard terms and conditions involving total consideration of less than \$25,000, or (B) the Asset Seller has licensed Intellectual Property constituting part of the Sold Assets that is material to the Business or involves consideration in excess of \$25,000 per annum to or from any other Person (including Affiliates of the Company), other than license agreements for unmodified "off-the-shelf" software on generally standard terms and conditions involving total consideration of less than \$25,000;
 - (vii) partnership, limited liability company or joint venture Contracts entered into by any Divested Company or the Asset Seller;
 - (viii) Contracts relating to the acquisition or sale of any business enterprise, assets or Equity Interests of any other Person involving total consideration or more than \$500,000;

- (ix) Contracts listed on Schedule 3.26;
- (x) Contracts (including employment and consulting Contracts) with (A) any current or former director, officer or employee of the Asset Seller or any of the Divested Companies or (B) any current or former shareholder or holder of Options or other rights, in each case, to acquire shares of capital stock or other Equity Interests of any Divested Company;
- (xi) employment agreements, collective bargaining agreements or other Contracts providing for retention, bonus, incentive, severance, change of control or termination pay;
 - (xii) Contracts involving a derivative or financial swap, exchange, commodity option or hedge;
- (xiii) Contracts (including any settlement agreement) entered into in the three years prior to the Execution Date with respect to any Proceeding;
- (xiv) Contracts providing exclusive distributor, dealer or similar rights or involving the provision of services that contains a most favored nation pricing provision;
 - (xv) Government Contracts;
 - (xvi) Credit Support Obligations; and
- (xvii) Contracts creating a material Encumbrance (other than a Permitted Encumbrance) on any Sold Assets or any Divested Company's assets or properties.
- (b) True and complete copies (or, if oral, written summaries) of each of the Material Contracts have been made available to the Buyers or their respective representatives.
- (c) Each Material Contract is in full force and effect, and is a valid and binding agreement of the Asset Seller or applicable Divested Company and, to the Knowledge of the Company, each of the other parties thereto, enforceable by or against the Asset Seller or such Divested Company, and, to the Knowledge of the Company, each of such other parties thereto, in accordance with its terms, subject to the General Enforceability Exceptions. No condition exists or event has occurred that (whether with or without notice or lapse of time or both) would (i) constitute a breach or default by (A) the Asset Seller or any of the Divested Companies under any Material Contract or (B) to the Knowledge of the Company, any other party to any Material Contract or (ii) permit any other party to any such Material Contract to terminate such Material Contract. No party to a Material Contract has given written notice of or, to the Company's Knowledge, threatened, any material dispute or any potential exercise of termination, acceleration or modification rights with respect to any of the Material Contracts. For purposes of the representations and warranties set forth in this Section 3.14(c) when made at Closing, the term "Material Contract" shall include any Contract entered into following the date of this Agreement that would have been a Material Contract had such Contract been in effect on the date of this Agreement.

3.15 Environmental Matters.

(a) The Asset Seller, and each of the Divested Companies, is and has been for the past five years in compliance in all material respects with all Environmental Laws, and no unbudgeted capital expenditures are reasonably anticipated to be required to achieve or maintain

such continued compliance with Environmental Laws over the next one-year period following the Closing for operations of the Business as conducted as of the Closing.

- (b) The Asset Seller, and each of the Divested Companies, possesses all Permits required for its operations as currently conducted under all applicable Environmental Laws and is in material compliance with the terms and conditions of such Permits.
- (c) Neither the Asset Seller nor any of the Divested Companies is subject to any pending Environmental Claim or has received written or, to the Knowledge of the Company, oral notice of any threatened Environmental Claim.
- (d) There has been no Release of any Hazardous Material at, from, in, on or under the Leased Real Property or in connection with the operation of the Business at any location that requires investigation, assessment, cleanup or remediation by the Asset Seller or the Divested Companies pursuant to any Environmental Law.
- (e) There is no Hazardous Material used, generated, treated, stored, transported, disposed of, released or handled on the Leased Real Property associated with the Business except in compliance with Environmental Laws.
- (f) The Company has furnished, and has caused its Affiliates (including the Sellers) to furnish, the Buyers with complete and accurate copies of all material environmental audits, assessments, reports and correspondence with Governmental Authorities regarding noncompliance with Environmental Laws and any actual or potential liability thereunder that have been prepared within the last five years and relate to the ownership or operation of the Business or the Leased Real Property.
- 3.16 Insurance. All insurance policies covering the Business or any assets, employees or operations of the Asset Seller or any Divested Company are held in the name of TriMas Corporation or the applicable Divested Company and are in full force and effect, all premiums due thereon have been paid and, where applicable, the Company or the applicable Divested Company has complied, or in the case of the Company, has caused the Divested Companies and the Asset Seller to comply, in all material respects with the provisions thereof. There is no Action or material claim pending under any such insurance policy with respect to the Business, the Sold Assets or any assets or properties of any Divested Company, and none of the Company, any Seller, any Divested Company or any Affiliate thereof has received any notice disclaiming coverage, reserving rights with respect to a particular claim or any such insurance policy in general or canceling or materially amending any such insurance policy. The Company maintains, and has maintained, since January 1, 2016, insurance programs and coverage in such amounts and against such risk as is reasonable and customary in the industry in which the Business operates. There is no default with respect to any provisions contained in any insurance policies carried by or for the benefit of the Business, the Sold Assets or any Divested Company, nor has there been any failure to give notice of or present any claim under such policies in a due a timely fashion.

3.17 Personal Property Assets.

(a) The Asset Seller has good and valid title to, or holds by valid and existing lease or license, the Sold Assets, free and clear of all Encumbrances, except for Permitted Encumbrances and Encumbrances under the Credit Agreement.

(b) As of the Execution Date, the Divested Companies have good title to, or hold by valid and existing lease or license, all the material tangible personal property assets reflected as assets on the Balance Sheet or acquired after the Balance Sheet Date, except with respect to assets disposed of in the ordinary course of business consistent with past practice since such date, free and clear of all Encumbrances except for Permitted Encumbrances and Encumbrances under the Credit Agreement.

3.18 Real Property.

- (a) Leased Properties. Schedule 3.18(a) sets forth all of the real property leased by the Asset Seller or a Divested Company (such leased real property, together with all buildings, structures, fixtures and improvements located thereon, collectively, the "Leased Real Property"). The Leased Real Property collectively represents all of the leases and subleases covering leased or subleased real property used or held for in, or necessary for the conduct of, the Business. Each applicable Asset Seller or Divested Company, as the case may, has a good and valid leasehold estate or interest in and to each tract or parcel of Leased Real Property, free and clear of Encumbrances (other than Permitted Encumbrances and Encumbrances under the Credit Agreement). The Company has made available to the Buyers or their respective representatives true and complete copies of the leases, subleases and Contracts covering the Leased Real Property (together with all amendments, modifications, supplements, extensions, renewals, memoranda and guaranties thereof or thereto, collectively, the "Real Property Leases," and each, a "Real Property Lease"). With respect to each Real Property Lease:
 - (i) Such Real Property Lease is valid and effective and in full force and effect in accordance with its terms, there are no existing defaults or events of default, or to the Knowledge of the Company, events which with notice or lapse of time or both would constitute material defaults, thereunder on the part of the Asset Seller or the relevant Divested Company (as applicable), and none of the foregoing have been asserted in writing;
 - (ii) to the Knowledge of the Company, there is no default or claimed or purported or alleged material default on the part of any other party in the performance of any obligation to be performed or paid by such other party under any such Real Property Lease;
 - (iii) no such Real Property Lease has been mortgaged, deeded in trust or encumbered by the Asset Seller or the Divested Companies, subject to Permitted Encumbrances;
 - (iv) there are no pending disputes with respect to any Real Property Lease; and
 - (v) neither the Asset Seller nor any Divested Company has subleased, licensed or sublicensed, or otherwise granted any Person the right to use or occupy, any Leased Real Property, subject to Permitted Encumbrances.
- (b) Improvements. Except as set forth on Schedule 3.18(b), all buildings, structures, improvements, fixtures, building systems and equipment, and all components thereof located on any Leased Real Property are in good condition and repair (subject to normal wear and tear) sufficient for the operation of the Business as currently conducted thereon.
- (c) No Other Leased Real Property. Other than the Leased Real Property, none of the Divested Companies or the Asset Seller lease or otherwise use any real property.

- (d) No Owned Properties. None of the Divested Companies or the Asset Seller own any real property.
- 3.19 Customers and Suppliers.
- (a) Set forth on Schedule 3.19(a) are the 20 largest customers of the Business, by dollar volume, for the 12-month periods ended December 31, 2018 and December 31, 2017 (collectively, the "Major Customers"), and set forth opposite the name of each Major Customer is the dollar amount of sales attributable to such customer for such period. None of the Company, the Sellers, the Divested Companies, or any of their respective Affiliates has received notice of any material dispute with respect to the Business with any Major Customer. Since January 1, 2017, (i) no Major Customer has terminated or materially reduced or changed the pricing, scope, regularity or other terms of its business with respect to the Business or (ii) notified the Asset Seller or any Divested Company that it intends to terminate or materially reduce or change the pricing, scope, regularity or other terms of its business with respect to the Business.
- (b) Set forth on Schedule 3.19(b) are the 20 largest suppliers of the Business, by dollar volume, for the 12-month periods ended December 31, 2018 and December 31, 2017 (collectively, the "Major Suppliers"). None of the Company, the Sellers, the Divested Companies, nor any of their respective Affiliates has received notice of any material dispute with respect to the Business with any supplier identified on Schedule 3.19(b). Since January 1, 2017, (i) no Major Supplier has terminated or materially reduced or changed the pricing, scope, regularity or other terms of its business with respect to the Business or (ii) notified the Asset Seller or any Divested Company that it intends to terminate or materially reduce or change the pricing, scope, regularity or other terms of its business with respect to the Business.

3.20 Compliance with Anti-Corruption Laws.

- (a) None of the Company, its Affiliates or any of their respective officers, directors, employees, or, to the Company's Knowledge, any agents or representatives acting on behalf of the Divested Companies, in connection with the ownership or operation of the Divested Companies, the Sold Assets or the Business, have during the past five years, directly or indirectly, taken any action that would violate any applicable Anti-Corruption Laws. Specifically, none of the Asset Seller, the Divested Companies or any of their respective officers, directors, employees, or, to the Company's Knowledge, any of their agents or representatives acting on behalf of the Company, in connection with the ownership or operation of the Sold Assets or the Business, have made, offered or authorized the making or offering, directly or indirectly through any Person, anything of value to any official, employee or representative of any Governmental Authority or of any entity owned, operated, or controlled by any Governmental Authority, or to any political party, party official, or candidate for political office, for the purpose of obtaining or retaining business, or to otherwise gain some improper advantage in violation of applicable Anti-Corruption Laws.
- (b) The Company and its Affiliates have maintained books and records that fairly and accurately reflect all transactions relating to the operation of the Sold Assets and the Business, and have implemented policies, procedures and controls, including providing anti-corruption training to relevant officers, directors, and employees, which are designed to promote, and which are reasonably expected to promote, compliance with all applicable Anti-Corruption Laws.

- 3.21 Export Controls and Economic Sanctions.
- (a) None of the Company, its Affiliates or any of their respective officers, directors, employees, or, to the Company's Knowledge, any of their agents or representatives acting on behalf of the Divested Companies, in connection with the ownership or operation of the Sold Assets or the Business, have during the past five years, directly or indirectly, taken any action that would violate any applicable export control or economic sanctions laws, regulations or Executive Orders of the United States, including the Arms Export Control Act (22 U.S.C. § 2751, et seq.), the International Traffic in Arms Regulations (22 C.F.R. Parts 120-130), the Export Control Reform Act of 2018 (50 U.S.C. Chapter 58), the Export Administration Regulations (15 C.F.R. Parts 730-774), and regulations promulgated by the U.S. Department of the Treasury's Office of Foreign Assets Control ("OFAC") (31 C.F.R. Parts 500-599) and corresponding enabling statutes, including the International Emergency Economic Powers Act (50 U.S.C. Chapter 35) and the Trading With the Enemy Act (50 U.S.C. Chapter 53) and any similar export control or economic sanctions laws of any country in which the Asset Seller or any Divested Company has conducted business.
- (b) Neither the Company, nor its Affiliates, nor any of their respective officers, directors, employees, or, the Company's Knowledge, any of their agents or representatives acting on behalf of the Divested Companies, is, or is owned 50% or more in the aggregate by anyone that is, (i) listed on the List of Specially Designated Nationals and Blocked Persons of OFAC or any other list of persons subject to economic sanctions or export restrictions maintained by the U.S. Departments of the Treasury, Commerce or State (including the Foreign Sanctions Evaders List or the Sectoral Sanctions Identifications List, each of which is administered and maintained by OFAC; the Denied Persons, Entity or Unverified Lists of the Bureau of Industry and Security; the Debarred List of the Directorate of Defense Trade Controls; and any list of sanctioned persons administered and maintained by the U.S. Department of State relating to nonproliferation, terrorism, Iran or Russia) or (ii) located, organized or residing in a country or territory that is the subject or target of any comprehensive U.S. trade embargo, including Cuba, Iran, North Korea, Syria or the Crimea Region of Ukraine.
- 3.22 Bankruptcy. There are no bankruptcy, insolvency, reorganization, legal causes for dissolution or receivership proceedings pending against, being contemplated by, or threatened against the Asset Seller, the Equity Sellers or any of the Divested Companies.
- 3.23 Accounts Receivable. The accounts and commissions receivable of the Business reflected in the Financial Information, or to be reflected in the calculation of Working Capital, constitutes, on a collective basis, genuine, valid and legally enforceable indebtedness of account debtors (subject only to the General Enforceability Exceptions) arising from bona fide transactions for goods sold or services performed in the ordinary course of business. Appropriate reserves against the accounts receivable are reflected in the Balance Sheet.
- 3.24 Inventory. The Company owns its inventory free and clear of all Encumbrances (other than Permitted Encumbrances) and acquired such inventory for sale in the ordinary course of business. Such inventory items are, in all material respects, in good and saleable condition. Provision has been made for any slow moving, damaged or obsolete inventory, and is reflected in reserves set forth in the Balance Sheet.
- 3.25 Operating Areas. Except as set forth on Schedule 3.25, none of the Asset Seller or the Divested Companies owns or leases any material assets, or employs any Person, in

any jurisdictions other than the United States, Canada, Belgium, Spain, Singapore and Thailand. None of the Divested Companies own any assets or have any liabilities unrelated to the Business.

- 3.26 Affiliate Transactions. Except as set forth on Schedule 3.26, none of the Company, its Affiliates or any officer, director or employee thereof or any individual related by marriage or adoption to any such individual or any entity in which any such Person owns any material beneficial interest, is a party to any Contract with the Asset Seller or any Divested Company or has any interest in any of the Sold Assets or in any of the Divested Companies or their respective assets.
- 3.27 No Brokers' or Other Fees. Except for Bank of America Merrill Lynch, whose fees and expenses will be paid by the Company, no Person has acted, directly or indirectly, as a broker, finder, financial advisor or investment banker for the Company or any of its Affiliates in connection with the transactions contemplated by this Agreement and no Person is entitled to any fee or commission or like payment in respect thereof.
- 3.28 Transaction Expenses. As of the Closing, none of the Asset Seller, the Divested Companies or the Buyers (or the Buyer's respective Affiliates) will have any liability for any Transaction Expenses.
- 3.29 Credit Support Obligations. Schedule 3.29 sets forth all letters of credit, guarantees, surety bonds and other credit support instruments issued by the Company or by any of its Affiliates (other than Divested Companies) in respect of obligations of the Business, the Divested Companies, the Sold Assets, or the Assumed Liabilities (collectively, the "Credit Support Obligations"). True, correct and complete copies of all Credit Support Obligations have been provided to the Buyers.
- 3.30 Shutdown Costs. As of the date of this Agreement, the Company's estimate of the aggregate cash costs reasonably likely to be incurred in ceasing operations in Belgium and Spain is approximately \$2,000,000.
- No Other Representations or Warranties. Except for the representations and warranties of the Company expressly contained in this Article III (as modified by the Disclosure Schedules), none of the Company, any Affiliate of the Company, any of their respective officers, directors, employees, agents or representatives, or any other Person are making or have made any representation or warranty of any sort to or for the benefit of the Buyer, any Affiliate of the Buyer, any of their respective officers, directors, employees, agents or representatives, or any other Person, whether oral or written, express or implied (including any implied warranty of merchantability or of fitness for a particular purpose), including with respect to the Company, the Sellers, the Business, the Sold Assets, the Divested Companies, the Shares, the Assumed Liabilities and the transactions contemplated by this Agreement, and each of the Company and the Sellers expressly disclaim any other representations or warranties. Except for the specific representations and warranties expressly made by the Company contained in this Article III (as modified by the Disclosure Schedules), each of the Company and the Sellers (a) hereby expressly disclaims any representation or warranty, express or implied, at Law or in equity, at common law, by statute or otherwise relating to (i) the condition of the Sold Assets (including any implied or express warranty of merchantability or fitness for a particular purpose, or of conformity to models or samples of materials) and (ii) the assets, liabilities, operations, prospects or condition of the Business, and (b) hereby expressly disclaims all liability and responsibility for any representation, warranty, statement or information made, communicated or furnished (orally or in writing) to the Buyer or its Affiliates or representatives

(including any opinion, information or advice that may have been or may be provided to the Buyer by any director, officer, employee, agent, consultant or representative of the Company, or any of its Affiliates). The Company makes no representations or warranties to the Buyer regarding the probable success or profitability of the Business. The Parties hereby expressly exclude the applicability of Title 1 book 7 of the Dutch Civil Code.

ARTICLE IV REPRESENTATIONS AND WARRANTIES OF THE BUYERS

Each Buyer hereby, jointly and severally, represents and warrants to the Company as of the Execution Date and the Closing Date (except to the extent a specific date is referenced, in which case each Buyer represents and warrants to the Company as of such date) as follows:

- 4.1 Organization. Such Buyer and its designee(s) is a limited liability company, corporation or company, as applicable, duly incorporated, formed or organized, as applicable, validly existing and (to the extent any such jurisdiction recognizes the concept of good standing) in good standing under the Laws of its jurisdiction of incorporation, formation or organization, as applicable. Such Buyer and each of its designee(s) has all requisite power and authority to own, lease and operate its assets and to carry on its business as now being conducted and is duly qualified or licensed to do business and (to the extent any such jurisdiction recognizes the concept of good standing) is in good standing in the jurisdictions (whether federal, state, local or foreign) in which the ownership of its property or the conduct of its business requires such qualification or license, except where the failure to be so qualified or licensed would not reasonably be expected, individually or in the aggregate, to have a material adverse effect on the ability of such Buyer or its designee(s) to consummate the transactions contemplated by this Agreement.
- 4.2 Authorization; Enforceability. Such Buyer and each of its designee(s) has all requisite power and authority to execute and deliver this Agreement and each Ancillary Agreement to which it is or will be a party and perform its obligations hereunder and thereunder. The execution and delivery of this Agreement and such Ancillary Agreements by such Buyer and its designee(s), as applicable, and the performance by each of them of their respective obligations hereunder and thereunder, have been duly authorized by all necessary action on the part of such Buyer and its designee(s), as applicable. This Agreement has been, and each of the Ancillary Agreements to which such Buyer is a party will be, at or prior to the Closing, duly executed and delivered by such Buyer and, assuming due authorization, execution and delivery by the Company, the Sellers party thereto or any of their respective Affiliates, as applicable, this Agreement constitutes, and each of the Ancillary Agreements will constitute, a valid and binding agreement of such Buyer, enforceable against such Buyer in accordance with its or their respective terms, subject to the General Enforceability Exceptions.
- 4.3 No Approvals or Conflicts. The execution, delivery and performance by such Buyer and its designee(s) of this Agreement and the Ancillary Agreements to which they are or will be a party and the consummation by such Buyer and its designee(s) of the transactions contemplated hereby and thereby do not and will not (a) violate, conflict with or result in a breach by any of such Buyer or its designee(s) of their respective organizational documents (including its certificate of incorporation, by-laws or similar documents), (b) violate, conflict with or result in a breach of, or constitute a default by any of such Buyer or its designee(s) (or create an event which, with notice or lapse of time or both, would constitute a default) or give rise to any payment or other penalty or any right of termination, cancellation or acceleration under, or result in the creation of

any Encumbrance upon any of the properties of such Buyer or its designee(s) under, any note, bond, mortgage, indenture, deed of trust, license, franchise, Permit, lease, Contract, agreement or other instrument to which any of such Buyer or its designee(s) or any of their respective properties may be bound, (c) violate or result in a breach of any Governmental Order or Law applicable to any of such Buyer or its designee(s) or any of their respective properties or (d) except for applicable requirements of the HSR Act or any other applicable antitrust, competition or investment Law or filings under the Investment Canada Act, require any order, consent, approval or authorization of, or notice to, or declaration, filing, application, qualification or registration with, any Person (including any Governmental Authority), except, with respect to the foregoing clauses (b), (c) and (d) above, as would not have a material adverse effect on the ability of such Buyer or its designee(s) to consummate the transactions contemplated by this Agreement.

- 4.4 Proceedings. There are no Proceedings pending or, to the Knowledge of the Buyers, threatened against such Buyer or its designee(s) that would have a material adverse effect on the ability of such Buyer or its designee(s) to consummate the transactions contemplated by this Agreement. Neither such Buyer nor its designee(s) is subject to any Governmental Order that would have a material adverse effect on the ability of such Buyer or its designee(s) to consummate the transactions contemplated by this Agreement.
- 4.5 Compliance with Laws; Permits. Neither such Buyer nor its designee(s) is in violation of any Governmental Order or Law applicable to them or any of their respective properties, except where noncompliance would not have a material adverse effect on the ability of such Buyer or its designee(s) to consummate the transactions contemplated by this Agreement. Such Buyer and its designee(s) have all Permits necessary to conduct their business as currently conducted, except where the failure to have such Permits would not have a material adverse effect on the ability of such Buyer or its designee(s) to consummate the transactions contemplated by this Agreement.

4.6 Financing.

(a) As of the Execution Date, the Buyers have received an executed commitment letter dated as of the Execution Date (the "Debt Commitment Letter") from the lenders party thereto (collectively, the "Lenders") pursuant to which the Lenders have agreed, subject to the terms and conditions thereof, to provide debt financing in the amounts set forth therein (the "Debt Financing") for the purposes of funding the Purchase Price and related fees and expenses required to be paid by the Buyers in connection with the transactions contemplated by this Agreement, including in connection with the Financing, on the Closing Date. The Buyers have received an executed equity commitment letter, dated as of the Execution Date (the "Equity Commitment Letter" and, together with the Debt Commitment Letter, the "Commitment Letters") from First Reserve Fund XIV, L.P., a Cayman Islands exempted limited partnership (the "Equity Investor"), pursuant to which the Equity Investor has agreed, subject to the terms and conditions thereof, to invest in the Buyers the amount set forth therein (the "Equity Financing"). The Equity Commitment Letter provides that the Company is a third-party beneficiary thereof and is entitled to enforce such agreement, subject to the terms of the Equity Commitment Letter. The Debt Financing pursuant to the Debt Commitment Letter and the Equity Financing pursuant to the Equity Commitment Letter are collectively referred to in this Agreement as the "Financing." The Buyers have delivered to the Company complete and correct copies of the executed Commitment Letters and the fee letter referenced in the Debt Commitment Letter (the "Fee Letter") (provided, that the Fee Letter has been redacted in a

customary manner, including, with respect to fee amounts, pricing caps, time periods, the rates and economic terms and "market flex" provisions therein).

- (b) Except as expressly set forth in the Commitment Letters and the Fee Letter, as of the Execution Date, there are no conditions precedent to the obligations of the Lenders and the Equity Investors to provide the Financing or any contractual contingencies that would permit the Lenders or the Equity Investors to reduce the total amount of the Financing in a manner that would result in the Buyers being unable to consummate the transactions contemplated by this Agreement. Assuming the truth and completeness of the representations and warranties of the Company contained in this Agreement, the satisfaction of the conditions to Closing set forth in Article VII and the performance by the Company of its other obligations in this Agreement, as of the Execution Date, the Buyers do not have any reason to reasonably believe that they will be unable to satisfy all terms and conditions to be satisfied by it in any of the Commitment Letters on or prior to the Closing Date.
- (c) Assuming the truth and completeness of the representations and warranties of the Company contained in this Agreement and the performance by the Company of its other obligations in this Agreement, the Financing, when funded in accordance with the Commitment Letters, shall provide the Buyers with cash proceeds on the Closing Date sufficient for the satisfaction of all of the Buyers' obligations to pay (i) the Purchase Price and (ii) any fees and expenses of or payable by the Buyers under this Agreement and under the Commitment Letters and due on the Closing Date.
- (d) As of the Execution Date, the Commitment Letters are valid and binding obligations in full force and effect of the Buyers and, to the Knowledge of the Buyers, of each of the other parties thereto, except to the extent limited by applicable bankruptcy, insolvency, reorganization, fraudulent conveyance, moratorium or other similar Laws relating to or affecting the rights of creditors generally, or by general equitable principles. As of the Execution Date, no event has occurred that, with or without notice, lapse of time, or both, would reasonably be expected to constitute a material failure to satisfy a condition precedent on the part of the Buyers under the terms and conditions of the Commitment Letters. The Buyers have paid in full any and all commitment fees or other fees required to be paid pursuant to the terms of the Commitment Letters on or before the Execution Date. Except as otherwise disclosed to the Company, none of the Commitment Letters has been modified or amended, as of the Execution Date (provided, that that the existence or exercise of "market flex" provisions contained in the Fee Letter shall not be deemed to constitute a modification or amendment of the Debt Commitment Letter). As of the Execution Date, to the Knowledge of the Buyers, none of the commitments under any of the Commitment Letters has been withdrawn or rescinded in any respect.
- 4.7 Solvency. Immediately after giving effect to the consummation of the transactions contemplated by this Agreement (including the Financing): (i) the fair saleable value (determined on a going concern basis) of the assets of the Buyers shall be greater than the total amount of their Liabilities (including all Liabilities, whether or not reflected in a balance sheet prepared in accordance with GAAP) (ii) the Buyers shall be able to pay their debts and obligations in the ordinary course of business as they become due; and (iii) the Buyers shall have adequate capital to carry on their businesses and all businesses in which they are about to engage.
- 4.8 Limited Guaranty. Concurrently with the execution of this Agreement, the Buyers have delivered to the Company the Limited Guarantee executed by the Equity Investor (the

"Limited Guarantee"), pursuant to which, and subject to the terms and conditions of which, the Equity Investor has guaranteed certain obligations of the Buyers under this Agreement. The Limited Guarantee is in full force and effect and is a valid and binding obligation of the Equity Investor, enforceable against the Equity Investor in accordance with its terms, except as such enforcement may be limited by applicable bankruptcy, insolvency, reorganization, fraudulent conveyance, moratorium or other similar Laws relating to or affecting the rights of creditors generally, or by general equitable principles, and no event has occurred, which, with or without notice, lapse of time or both, would constitute a default on the part of the Equity Investor under the Limited Guarantee.

- 4.9 No Brokers' or Other Fees. Except as set forth on Schedule 4.9, whose fees and expenses will be paid by the Buyers, no Person has acted, directly or indirectly, as a broker, finder, financial advisor or investment banker for the Buyers or their respective designee(s) in connection with the transactions contemplated by this Agreement and no Person is entitled to any fee or commission or like payment in respect thereof.
- Condition of the Business. Such Buyer acknowledges and agrees that (a) the Company and its Affiliates have not made and are not making any representations or warranties whatsoever, express or implied, at Law or in equity, beyond those expressly given by the Company in Article III (as modified by the Disclosure Schedules) and in the certificate delivered pursuant to Section 7.3, (b) except for the representations and warranties of the Company expressly contained in Article III (as modified by the Disclosure Schedules) and in the certificate delivered pursuant to Section 7.3, and the representations and warranties of the Company or its Affiliates in any Ancillary Agreement, as applicable, the Sold Assets, the Shares and the Business are being transferred on a "where is" and, as to condition, "as is" basis, and (c) no officer, director, employee, agent or representative of the Company or any of its Affiliates has any authority, express or implied, to make any representations or warranties not specifically set forth in Article III (as modified by the Disclosure Schedules), in the certificate delivered pursuant to Section 7.3 or in any Ancillary Agreement. Except for the representations and warranties of the Company expressly contained in Article III (as modified by the Disclosure Schedules) and in the certificate delivered pursuant to Section 7.3, and the representations and warranties of the Company or its Affiliates in any Ancillary Agreement, as applicable, (x) such Buyer specifically disclaims, for itself and on behalf of its Affiliates any obligation or duty by the Company to make any disclosures of facts not required to be disclosed pursuant to the specific representations and warranties set forth in Article III, and (y) such Buyer is executing and delivering this Agreement and consummating the transactions contemplated hereby subject only to the specific representations and warranties set forth in Article III, in the certificate delivered pursuant to Section 7.3 and in any Ancillary Agreement, as further limited by the specifically bargained-for exclusive remedies as set forth herein. Any claims such Buyer may have for breach of representation or warranty shall be based solely on the representations and warranties of the Company expressly set forth in Article III (as modified by the Disclosure Schedules) and in the certificate delivered pursuant to Section 7.3, and the representations and warranties of the Company or its Affiliates in any Ancillary Agreement, as applicable. Such Buyer further represents that neither the Company nor any of its Affiliates, nor any of their respective officers, directors, employees, agents or representatives nor any other Person has made any representation or warranty, express or implied, as to the accuracy or completeness of any information regarding any of the Company, the Business or the transactions contemplated by this Agreement not expressly set forth in Article III (as modified by the Disclosure Schedules), in the certificate delivered pursuant to Section 7.3 or in any Ancillary Agreement, and none of the Company, any of its Affiliates, any of their respective officers, directors, employees, agents or representatives or any other Person will have or be subject to any liability to such Buyer, any Affiliate of such Buyer, any of their respective officers, directors,

employees, agents or representatives or any other Person resulting from the distribution to such Buyer (or its Affiliates) or its representatives, or such Buyer's use of, any such information, including any confidential memoranda distributed on behalf of the Company relating to the Business or other publications, projections, data room or virtual data room information provided to such Buyer (or its Affiliates) or its representatives, or any other document or information in any form provided to such Buyer (or its Affiliates) or its representatives, including management presentations, in connection with the sale of the Shares and the Sold Assets, the transactions contemplated hereby and in respect of any other matter or thing whatsoever. Except for the representations and warranties of the Company expressly contained in Article III (as modified by the Disclosure Schedules) and in the certificate delivered pursuant to Section 7.3, and the representations and warranties of the Company or its Affiliates in any Ancillary Agreement, as applicable, such Buyer acknowledges that it has conducted to its satisfaction its own independent investigation of the Business and, in making the determination to proceed with the transactions contemplated by this Agreement, such Buyer has relied on the results of its own independent investigation. The Parties hereby expressly exclude the applicability of Title 1 book 7 of the Dutch Civil Code.

4.11 No Other Representations or Warranties. Notwithstanding anything to the contrary in this Agreement, except for (a) the representations and warranties of such Buyer expressly contained in this Article IV (as modified by the Disclosure Schedules) and in the certificate delivered by such Buyer pursuant to Section 6.3, and (b) the representations and warranties of such Buyer or its Affiliates in any Ancillary Agreement, as applicable, (i) such Buyer is not making and has not made any representation or warranty of any sort to or for the benefit of the Company, any Seller, any of their respective Affiliates, any of their respective officers, directors, employees, agents or representatives, or any other Person, whether oral or written, express or implied (including any implied warranty of merchantability or of fitness for a particular purpose), with respect to its business and the transactions contemplated by this Agreement, (ii) such Buyer expressly disclaims any such other representations or warranties and (iii) such Buyer hereby expressly disclaims all liability and responsibility for any representation, warranty, statement or information made, communicated or furnished (orally or in writing) to the Company, any Seller or their respective Affiliates or representatives (including any opinion, information or advice that may have been or may be provided to any such Persons by any director, officer, employee, agent, consultant or representative of such Buyer or any of its Affiliates).

ARTICLE V COVENANTS AND AGREEMENTS

5.1 Conduct of Business Prior to the Closing. Without the prior written consent of the Buyers (which shall not be unreasonably withheld, conditioned or delayed), except as expressly required or permitted by this Agreement or as disclosed on Schedule 5.1, from and after the Execution Date until the Closing (the "Interim Period"), the Company shall cause the Divested Companies and the Asset Seller to (x) conduct the operations of the Business in the ordinary course consistent with past practice, (y) use commercially reasonable efforts to maintain satisfactory relationships with all suppliers, customers and others having material business relationships with the Business and (z) use commercially reasonable efforts to (1) maintain in effect all Permits necessary for the conduct of the Business and all insurance policies covering the Business or any assets, employees or operations of the Asset Seller or any Divested Company, and (2) amend the Articles of Association of Lamons Belgium to remove Article 10 of such Articles of Association. Without limiting the generality of the foregoing, except as expressly required or permitted by this Agreement and except as set forth on Schedule 5.1, the Company shall cause the Divested

Companies, the Asset Seller and their respective Affiliates with respect to the Business not to do any of the following without the prior written consent of the Buyers, which consent shall not be unreasonably withheld, conditioned or delayed:

- (a) issue, sell, repurchase or grant any Equity Interests of the Asset Seller or any Divested Company or grant or make any Option or agreement of any character in respect of such Equity Interests;
- (b) sell or otherwise dispose of any Sold Assets or assets or properties of a Divested Company having a value, individually or in the aggregate, in excess of \$200,000, excluding sales of inventory in the ordinary course of business consistent with past practice;
- (c) acquire any assets having a value, individually or in the aggregate, in excess of \$200,000, excluding (i) acquisitions of inventory in the ordinary course of business consistent with past practice and (ii) capital commitments or expenditures permitted by clause (e) below;
- (d) merge or consolidate with any Person, or effect any recapitalization, reclassification, Equity Interest split, combination or similar change in the capitalization of any Divested Company;
- (e) make any capital commitments or expenditures, individually or in the aggregate, in excess of \$200,000;
- (f) (i) in the case of the Divested Companies, incur, assume or guarantee any Debt Obligation, (ii) in the case of the Asset Seller, incur, assume or guarantee any Debt Obligation that would become an Assumed Liability or (iii) in the case of the Divested Companies, make any loans to or invest in any Person;
- (g) incur any Encumbrance on any assets of any Divested Company or any Sold Asset, in each case, other than Permitted Encumbrances;
- (h) increase the base salary or hourly wage rate of employees of the Business (including Active Employees) other than (i) in the ordinary course of business consistent with past practice, (ii) as required by any Contract in effect as of the Execution Date or (iii) as required by applicable Law;
- (i) incur any Encumbrance on any of the Shares, other than Encumbrances arising from restrictions on sale, assignment or transfer of shares of capital stock under applicable Law;
- (j) (i) make any material change in the accounting methods or practices followed by any Divested Company or any Affiliate thereof with respect to the Business (other than such changes required by applicable Law or GAAP) or (ii) waive, impair or write down any receivable other than in the ordinary course of business consistent with past practice;
- (k) enter into any Contract that restricts or will restrict any Divested Company or the Business after the Execution Date from engaging in any line of business in any geographic area or competing with any Person;
 - (I) enter into any partnership, limited liability company or joint venture Contract;

- (m) enter into or amend any collective bargaining Contracts or other contract with a labor union, trade union, works council, or representative of Active Employees;
- (n) (i) make any settlement or compromise any material Tax liability with respect to any Divested Company or any Affiliate thereof or with respect to the Sold Assets solely with respect to any Asset Tax, change any material Tax election or Tax method of accounting with respect to any Divested Company or any Affiliate thereof or with respect to the Sold Assets solely with respect to any Asset Tax or make any new Tax election or adopt any new Tax method of accounting, in each case, other than any settlement, compromise, change, method made in the ordinary course of business with respect to any Divested Company or any Affiliate thereof or with respect to the Sold Assets solely with respect to any Asset Tax, (ii) surrender any right to claim a refund of Taxes with respect to any Divested Company or any Affiliate thereof or with respect to the Sold Assets solely with respect to any Asset Tax or (iii) consent to any extension or waiver of the limitations period applicable to any Tax claim or assessment with respect to any Divested Company or any Affiliate thereof or with respect to the Sold Assets solely with respect to any Asset Tax;
- (o) other than in the ordinary course of business, transfer the employment of any Active Employee, terminate the employment (other than for cause) of any Active Employee or hire any individual who would be an Active Employee (other than any hiring necessary to replace an Active Employee whose employment has terminated and on terms no more favorable to such newly hired individual as those provided to the individual whose employment terminated);
- (p) other than (i) in the ordinary course of business consistent with past practice, (ii) as expressly required by any Material Contract in effect as of the Execution Date or (iii) as required by applicable Law, adopt, terminate, or amend any material employment Contract or Seller Benefit Plan or any other employee benefit or compensation plan, program, policy, Contract or arrangement that, in either case, would result in increased costs to any Divested Company, any Buyer or any of their respective Affiliates after the Closing;
 - (q) amend any organizational documents of any Divested Company;
- (r) compromise or settle any material claim or Action with respect to any Sold Assets or any assets or properties of any Divested Company;
- (s) sell, transfer, assign, exclusively license, abandon, permit to lapse or otherwise dispose of any Acquired Intellectual Property other than in the ordinary course of business consistent with past practice;
- (t) enter into any Contract that, if in effect as of the Execution Date, would be a Material Contract, or any Real Property Lease, or terminate (other than as a result of expiration in accordance with its terms), waive the performance of any obligation under or amend any Material Contract or Real Property Lease, in each, case, except in the ordinary course of business consistent with past practice;
 - (u) liquidate, dissolve or otherwise wind up the affairs of the Asset Seller or any Divested Company; or
 - (v) agree or commit to do any of the foregoing.

- 5.2 Access; Preservation of Books and Records.
- (a) During the Interim Period, upon reasonable prior written notice, the Company shall, and shall cause its Affiliates to provide the Buyers and their respective representatives reasonable access, during normal business hours, to the properties, personnel and the books and records of the Company and its Affiliates to the extent related to Business, the Sold Assets, the assets and properties of the Divested Companies, or the Assumed Liabilities, in each case, solely for the purpose of consummating the transaction contemplated by this Agreement or preparing for the operations of or financial reporting with respect to the Business from and after the Closing; provided, however, that (i) all access shall be at the Buyer's sole cost and expense, (ii) in no event shall the Company be obligated to provide any bids, the identity of any bidder, confidentiality or non-disclosure agreements, letters of intent, expressions of interest or other proposals received in connection with transactions comparable to those contemplated by this Agreement or any information or analysis relating to any such communications, (iii) prior to the Closing, the Buyers shall not undertake, or permit or cause any Person to undertake, the taking of soil, groundwater, surface water, soil gas, air or building material samples or other invasive testing on any Leased Real Property, and (iv) such investigation shall not unreasonably interfere with any of the businesses or operations of the Company, the Sellers or their Affiliates; and (v) neither the Company nor any of its Affiliates shall be required to provide (or cause to be provided) access to or to disclose (or cause to be disclosed) information that (A) based on the advice of their respective legal counsel, is subject to an attorney-client privilege or an attorney work-product privilege, (B) their respective legal counsel reasonably concludes would give rise to antitrust or competition Law issues or would contravene any applicable Law or contractual confidentiality obligation or (C) their respective legal counsel reasonably concludes would give rise to antitrust or competition Law issues or would contravene any applicable Law, provided, however, that in the case of the foregoing clauses (A), (B) or (C), the Company shall provide notice to the Buyers that such information is being withheld and the Buyers and the Company shall, and shall cause its Affiliates to, reasonably cooperate to seek an appropriate remedy to permit the access contemplated hereby.
- (b) For a period of six years following the Closing Date (or such longer periods as may be required by applicable Law), the Buyers agree that they shall, and shall cause their respective Affiliates to, preserve and keep all pre-Closing records of the Asset Seller and the Divested Companies in its or their possession, as applicable, and the Company and its Affiliates agree to preserve the records held by them relating to the Business. The Buyers and the Company agree that each shall make, or cause their respective Affiliates to make, such records and personnel available to the other and its or their respective Affiliates, at such Party's expense and during normal business hours, as may be reasonably required in connection with, among other things, pursuing or defending any insurance claims by, legal proceedings against (other than legal proceedings by the Buyers or any of their respective Affiliates against the Company or any of its Affiliates or vice versa) or governmental investigations involving, the Buyers or the Company or any of their respective Affiliates, or in order to enable the Buyers or the Company or any of their respective Affiliates to comply with their respective obligations under this Agreement and the Ancillary Agreements; provided, however, the Parties and their respective Affiliates shall not be required to provide (or cause to be provided) access to or to disclose (or cause to be disclosed) information that (i) based on the advice of their respective legal counsel, is subject to an attorney-client privilege or an attorney work-product privilege, (ii) their respective legal counsel reasonably concludes would give rise to antitrust or competition Law issues or would contravene any applicable Law or contractual confidentiality obligation or (iii) their respective legal counsel reasonably concludes would give rise to antitrust or competition Law issues or would contravene any applicable Law,

provided, however, that in the case of the foregoing clauses (i), (ii) or (iii), the withholding Party shall provide notice to the other Party that such information is being withheld and the Parties shall, and shall cause its Affiliates to, reasonably cooperate to seek an appropriate remedy to permit the access contemplated hereby. If the Buyers or the Company or any of their respective Affiliates wish to destroy (or permit to be destroyed) any such records prior to the end of the six-year period (or such longer period as may be required by applicable Law) described above, then the destroying Party shall first give 60 days' prior written notice to the other Party or Parties, as the case may be, and such Party or Parties will have the right, at such Party's or Parties' option and expense, upon prior written notice given to the destroying Party within such 60-day period, to take possession of any such records. If such Party does not retrieve the applicable records within 30 days after the date of such destruction notice, then the destroying Party may destroy (or permit to be destroyed) any such records.

- 5.3 Regulatory Filings; Reasonable Best Efforts.
- (a) Upon the terms and subject to the conditions set forth in this Agreement, each Party shall (and to the extent applicable, shall cause their respective Affiliates to) use its reasonable best efforts to take, or cause to be taken, all actions, to do, or cause to be done, and to assist and cooperate with the Party in doing, all things necessary, proper or advisable to consummate and make effective, as promptly as practicable, the transactions contemplated hereby in accordance with the terms of this Agreement, including: (i) the obtaining of all necessary approvals under any applicable Laws required in connection with this Agreement and the transactions contemplated hereby; (ii) the obtaining of all necessary actions or non-actions, waivers, consents, approvals and authorizations from Governmental Authorities and the making of all necessary registrations and filings (including filings with Governmental Authorities) and the taking of all steps as may be necessary to obtain an approval or waiver from, or to avoid a Proceeding by, any Governmental Authority: (iii) the obtaining of all necessary waivers, consents, approvals and authorizations from, and making any required notices to, any third party; and (iv) the execution and delivery of any additional instruments necessary to consummate the transactions contemplated hereby in accordance with the terms of this Agreement and to fully carry out the purposes of this Agreement. Unless otherwise expressly agreed herein, the Sellers or any of their respective Affiliates shall not be obligated to pay any consideration, fee or other similar payment to any third party from who consent or approval is required or requested from or by such third party in connection with the consummation of the transactions contemplated by this Agreement in order to obtain any such consent or approval. The Parties undertake and agree (and to the extent applicable, cause their Affiliates to undertake and agree) to file (A) a Notification and Report Form under the HSR Act with the FTC and the Antitrust Division within ten Business Days after the Execution Date; and (B) such other filings and applications for approvals and consents as are required under any other applicable Laws as soon as practicable. The Parties shall seek early termination of the waiting period under the HSR Act. The Company, on the one hand, and the Buyers, on the other hand, shall each bear 50% of the responsibility for all filings fees associated with filings pursuant to the HSR Act and all other antitrust and other regulatory filings, including those that may be required to be filed in any other jurisdiction.
- (b) For the avoidance of doubt, the Buyers and their respective Subsidiaries shall: (i) agree or commit to any and all divestitures, licenses or hold separate or similar arrangements with respect to its businesses, properties or assets as well as, subject to the Closing, the businesses, properties or assets of the Business, and agree or commit to any restrictions or limitations on the conduct of its businesses, as well as, subject to the Closing, on the conduct of the businesses of

the Divested Companies, in each case, as a condition to obtaining any and all approvals from any Governmental Authority for any reason in order to consummate and make effective, as promptly as practicable, the transactions contemplated hereby, including taking any and all actions necessary in order to ensure that no: (A) requirement for nonaction, a waiver, consent, approval or authorization of the U.S. Federal Trade Commission (the "FTC"), the Antitrust Division of the U.S. Department of Justice (the "Antitrust Division"), any authority enforcing applicable Law, any state attorney general or any other Governmental Authority; (B) Governmental Order or Proceeding; and (C) other matter relating to any applicable antitrust, competition or other applicable Law, in each case, would preclude consummation of the transactions contemplated hereby by the End Date; and (ii) agree and commit to litigate or participate in the litigation of any Proceeding involving the FTC, the Antitrust Division or any other Person, whether judicial or administrative, in order to: (1) oppose or defend against any Proceeding to prevent or enjoin the consummation of the transactions contemplated by this Agreement; or (2) overturn any Proceeding to prevent consummation of the transactions contemplated by this Agreement, including by defending any Proceeding in order to avoid the entry of, or to have vacated, overturned, terminated or appealed any Governmental Order. In addition, upon the terms and subject to the conditions herein provided and subject to the Parties' (and to the extent applicable, their respective Affiliates') obligations under applicable Law, prior to the expiration of the waiting period under the HSR Act or early termination thereof, neither Party shall (and each Party shall cause, to the extent applicable, their respective Affiliates not to) knowingly take, or cause to be taken, any action that would reasonably be expected to materially delay or prevent the satisfaction by the End Date of the conditions set forth in Sections 6.4, 6.5, 7.4 and 7.5.

- (c) Each Party shall and, to the extent applicable, shall cause their respective Affiliates to (i) respond as promptly as practicable to all inquiries and requests received from the FTC, the Antitrust Division, any state attorney general or any other Governmental Authority in connection with antitrust matters or applicable Laws and (ii) not extend any waiting period under the HSR Act or enter into any agreement with the FTC, the Antitrust Division, any state attorney general or any other Governmental Authority not to consummate the transactions contemplated by this Agreement, except, in each case, with the prior written consent of the other Party.
- (d) Subject to applicable Law, each Party shall, and to the extent applicable, shall cause their respective Affiliates to: (i) promptly notify the other Party or Parties of any written communication from the FTC, the Antitrust Division, any state attorney general or any other Governmental Authority concerning this Agreement or the transactions contemplated hereby and permit the other Party or Parties to review in advance any proposed communication to any of the foregoing; (ii) to the extent reasonably practicable, consult with the other Party or Parties prior to participating in any meeting, telephone call or discussion with any Governmental Authority with respect to any filing, investigation or inquiry concerning this Agreement or the transactions contemplated hereby and provide the other Party or Parties the opportunity to attend and participate in any such meeting, telephone call or discussion to the extent that such Party or Parties is or are permitted by such Governmental Authority to attend and participate; and (iii) furnish the other Party with copies of all correspondence, filings and written communications (or a reasonably detailed summary of any oral communications) between it and its representatives, on the one hand, and any Governmental Authority or members of their respective staffs, on the other hand, with respect to this Agreement or the transactions contemplated hereby and provide a reasonable opportunity to the other Party or Parties to comment on letters, presentations, whitepapers and other substantive communications to the Governmental Authority and consider, in good faith, any reasonable comments on such correspondences, filings and written communications; provided, however, that

in the event of any disagreement concerning such matters contemplated by clauses (ii) and (iii), the determination of the Buyers shall be final and conclusive. The Buyers shall, on behalf of the Parties, control and lead all communications and strategy relating to the HSR Act.

- (e) Nothing in this Agreement shall impose obligations on a Party or its Affiliates to give the other Party or Parties or their counsel, accountants or other authorized representatives access to information if such access would reasonably be expected to cause such Party or any of its Affiliates to be in breach of any duty of confidence or any other duty or obligation under applicable Law (including Laws affecting privacy, personal information, competition and the collection, handling, storage, processing, use or disclosure of data).
 - 5.4 Tax Matters; Cooperation; Preparation of Returns; Tax Elections.
- (a) The Buyers and the Company agree to furnish or cause to be furnished to each other, upon request, as promptly as practicable, such information and assistance relating to any of the Sold Assets or the Divested Companies (including access to any books and records, employees, contractors and representatives) as is reasonably necessary for the filing of all Tax Returns, the making of any election related to Taxes, the preparation for or defense of any audit by any Taxing Authority and the prosecution or defense of any claim, suit or other Proceeding with respect to any Taxes relating to any of the Sold Assets, the Divested Companies, or any subsidiary of a Divested Company. Further, the Company and the Sellers shall be permitted to retain, in their discretion, copies of any such books and records relating to any of the Sold Assets or the Divested Companies as is reasonably necessary for any of such purposes as set forth above. The Buyers shall, and shall cause their respective designee(s) to, and the Company shall, and shall cause its Affiliates to, retain all books and records with respect to Taxes pertaining to the Sold Assets and the Divested Companies until the expiration of all relevant statutes of limitations (and, to the extent notified by the Buyers and the Company, any extensions thereof).
- (b) The Company shall, and shall cause its Affiliates to, prepare and timely file, or cause to be prepared and timely filed, all Tax Returns of a Consolidated Group that includes any of the Divested Companies and shall pay any and all Taxes owed with respect to such Tax Returns.
- (c) The Buyers shall prepare, or cause to be prepared, all Tax Returns required to be filed by or on behalf of the Divested Companies or with respect to the Sold Asset for any Straddle Period ("Straddle Period Tax Returns") and all Tax Returns required to be filed by or on behalf of the Divested Companies after the Closing Date for any Pre-Closing Tax Period ("Pre-Closing Tax Returns"). The Buyers shall provide to the Company drafts of any such Straddle Period Tax Returns and Pre-Closing Tax Returns for review and reasonable comment at least 30 days prior to the due date for the filing of each such Straddle Period Tax Return and Pre-Closing Tax Return (including any applicable extensions). Not later than seven days after the Buyers have provided each such Straddle Period Tax Return and Pre-Closing Tax Return to the Company, the Company shall notify the Buyers of any reasonable comments that the Company may have to any item set forth on such draft Straddle Period Tax Return or Pre-Closing Tax Return. The Buyers shall revise any such Straddle Period Tax Return or Pre-Closing Tax Return to incorporate any reasonable comments received from the Company. If the Buyers object to any such comments that the Company may have to any item set forth on such Straddle Period Tax Return or Pre-Closing Tax Return, the Buyers shall, within seven days after delivery of such Straddle Period Tax Return or Pre-Closing Tax Return, notify the Company in writing that the Buyers so object, specifying with particularity any such item and stating the basis for any such objection. If a notice of objection shall

be duly delivered to the Company by the Buyers, then the Buyers and the Company shall negotiate in good faith and use their commercially reasonable efforts to resolve such items. If the Buyers and the Company are unable to reach such agreement within three days after receipt by the Company of such notice, then the disputed items shall be resolved by the CPA Firm and any determination by the CPA Firm shall be final and binding on the Parties. The CPA Firm shall resolve any disputed items within six days of having the item referred to it pursuant to such procedures as it may require. If the CPA Firm is unable to resolve any disputed items before the due date for such Tax Return, then the Tax Return shall be filed without the comments that the Company provided to the Buyers and then amended, if necessary, to reflect the CPA Firm's resolution. The costs, fees and expenses of the CPA Firm incurred under this Section 5.4(c) shall be borne equally by the Buyers, on the one hand, and the Company, on the other hand. The Company shall pay or cause to be paid to the Buyers, at least five days prior to the filing of any Straddle Period Tax Return or Pre-Closing Tax Return, the amount of any Seller Taxes determined in accordance with Section 5.5(b) reflected on any such Straddle Period Tax Return or Pre-Closing Tax Return. Notwithstanding anything in this Agreement to the contrary, Buyers shall be permitted to make an election under Section 256(9) of the Income Tax Act (Canada) on any Tax Return filed with respect to the taxable year of Lamons Canada that ends as a result of the Buyers' acquisition of the Lamons Canada Shares.

- (d) All transfer, documentary, sales, value added, use, registration, Duty and other such Taxes (including all applicable real estate transfer Taxes, but excluding any Income Taxes or Asset Taxes) and related fees (including any penalties, interest and additions to Tax) incurred in connection with this Agreement and the transactions contemplated hereby ("Transfer Taxes") shall be paid 50% by the Buyers and 50% by the Company. The Parties shall cooperate in good faith to minimize, the extent permissible under applicable Law, the amount of any such Transfer Taxes. The Buyers, at their expense, will prepare and timely file all necessary Tax Returns and other documentation with respect to all such Transfer Taxes, and, if required by applicable Law, the Company and the Buyers will, and will cause their respective Affiliates to, cooperate and join in the execution of any such Tax Returns and other documentation.
- (e) Effective as of the Closing, any Tax sharing or Tax allocation Contract or arrangement, in each case, other than any Commercial Tax Agreement, between the Company or any of its Affiliates (other than a Divested Company) and any Divested Company shall be terminated and shall have no further effect thereafter without any Liability accruing to or being imposed on such Divested Company with respect thereto.
- (f) Neither the Buyers nor any Divested Company shall make any election pursuant to Section 336 or Section 338 of the Code with respect to any Divested Company, unless the Company consents to such election in writing.

5.5 Tax Indemnity.

(a) The Company shall indemnify the Buyers and their respective Affiliates (including, after the Closing, the Divested Companies) and each of their respective officers, directors, partners, members, equity holders, employees, agents and representatives and hold them harmless from, against and in respect of (i) any and all Seller Taxes and (ii) all Liabilities, costs, expenses (including reasonable expenses of investigation and attorneys' fees and expenses), losses, damages, assessments, settlements or judgments arising out of or incident to the imposition, assessment or assertion of any such Seller Taxes. Other than with respect to an election under

Section 336 or Section 338, the Tax indemnity provided under this Section 5.5(a) shall not cover any Tax liabilities resulting from any transactions of any Divested Company not in the ordinary course of business that occur on the Closing Date after the Closing. Notwithstanding anything in this Agreement to the contrary, the indemnification obligations provided for in this Section 5.5 shall survive the Closing until the date that is 60 days after the expiration of the applicable statute of limitations.

- (b) For purposes of this Article V, in the case of any Taxes of any Divested Company or with respect to the Sold Assets that are imposed on a periodic basis and are payable for a Straddle Period, the amount of such Taxes apportioned to the portion of such Straddle Period ending on the Closing Date shall be (i) in the case of any Taxes other than sales or use taxes, value-added taxes, employment taxes, withholding taxes, and any Tax based on or measured by income, receipts or profits earned during a Straddle Period, be deemed to be the amount of such Tax for the entire taxable period multiplied by a fraction, the numerator of which is the number of days in the taxable period ending on and including the Closing Date and the denominator of which is the number of days in the Straddle Period, and (ii) in the case of any sales or use taxes, value-added taxes, employment taxes, withholding taxes, and any Tax based on or measured by income, receipts or profits earned during the Straddle Period, be deemed equal to the amount that would be payable if the Straddle Period ended on and included the Closing Date. In the case of any Taxes that are imposed on the Buyers or any of their Affiliates as result a gross income inclusion under Section 951 or Section 951A of the Code with respect to a Divested Company or subsidiary of a Divested Company, the amount of such Taxes apportioned to the portion of a U.S. federal Income Tax Straddle Period ending on the Closing Date shall be the Taxes that would have been imposed on the Company and its Affiliates (not taking into account any reduction in such Taxes resulting from the utilization of any loss carryforwards, credits or any other offsets or reductions by the Company or any of their Affiliates) with respect to a U.S. federal Income Tax taxable year of such Divested Company or subsidiary of a Divested Company beginning on the first day of the Straddle Period and ending on the Closing Date. All determinations necessary to give effect to the foregoing allocations shall be made in a manner consistent with the past practice of the Equity Seller, the relevant Divested Company or the Asset Seller, as applicable.
- (c) Payment by the indemnitor of any amount due under this Section 5.5 shall be made within 10 days following written notice by the indemnitee that payment of such amounts has been made or is due and payable. In the case of a Tax that is contested in accordance with the provisions of Section 5.6, payment of the Tax to the appropriate Taxing Authority shall not be considered to be due earlier than the date a final determination to such effect is made by the appropriate Taxing Authority, provided, however, that to the extent Lamons Canada is required by a Taxing Authority to pay any portion of a contested amount to such Taxing Authority prior to the date a final determination is made, the indemnitor shall pay such amount at such time.
- 5.6 Procedures Relating to Indemnity of Tax Claims. The Buyers shall promptly notify the Company in writing (a) of the receipt by the Buyers or any Divested Company of any written notice of any inquiries, claims, assessments, audits or similar events with respect to Taxes for any Pre-Closing Tax Period or for the portion of any Straddle Period ending on the Closing Date (determined in accordance with Section 5.5(b)), for which the Company or any of its Affiliates may be liable under Section 5.5(a) or (b) prior to any voluntary contact with any Taxing Authority relating to, or self-assessment of, any Taxes of any Divested Company for any Pre-Closing Tax Period or for the portion of any Straddle Period ending on the Closing Date (determined in accordance with Section 5.5(b)) (a "Tax Claim"), stating the nature and basis of such Tax Claim and the amount

thereof, to the extent known. To the extent notice of any Tax Claim is not promptly given to the Company, the Company shall not be liable pursuant to Section 5.5(a) to the extent that the Company's ability to effectively contest such Tax Claim is prejudiced as a result thereof. With respect to any Tax Claim that relates solely to a Pre-Closing Tax Period, the Company shall have the exclusive right to elect to control all proceedings taken in connection with such Tax Claim (including selection of counsel) and, without limiting the foregoing, may in its sole discretion pursue or forego any and all administrative appeals, proceedings, hearings and conferences with any Taxing Authority with respect thereto and may, in its sole discretion, either pay the Tax claimed and sue for a refund where applicable Law permits such refund suits or contest the Tax Claim in any permissible manner; provided, however, that the Company shall not take or advocate any position that would reasonably be expected to result in an increase in Taxes of the Buyers or any of their respective Affiliates (including the Divested Companies) or that involves a change in (or adoption of) a new Tax method of accounting applicable to any Post-Closing Tax Period without the written consent of the Buyers, which consent shall not be unreasonably withheld, conditioned or delayed. With respect to any Tax Claim that relates solely to a Post-Closing Tax Period, the Buyers shall have the exclusive right to elect to control all proceedings taken in connection with such Tax Claim (including selection of counsel) and, without limiting the foregoing, may in their sole discretion pursue or forego any and all administrative appeals, proceedings, hearings and conferences with any Taxing Authority with respect thereto and may, in their sole discretion, either pay the Tax claimed and sue for a refund where applicable Law permits such refund suits or contest the Tax Claim in any permissible manner; provided, however, that the Buyers shall not take or advocate any position that would reasonably be expected to result in an increase in Taxes of the Company or any of its Affiliates (including the Divested Companies with respect to a Pre-Closing Tax Period) pursuant to Section 5.5(a) without the written consent of the Company, which consent shall not be unreasonably withheld, conditioned or delayed. With respect to any Tax Claim that relates to both a Pre-Closing Tax Period and a Post-Closing Tax Period, the Parties shall mutually control all proceedings taken in connection with such Tax Claim (including selection of counsel); provided, however, that if, with the written consent of the other Party or Parties, one Party is permitted to control the Tax Claim, the controlling Party or Parties shall not take or advocate any position that would reasonably be expected to result in an increase in Taxes of the other Party or Parties or any Affiliates thereof, without the written consent of the other Party or Parties or any Affiliates thereof, which consent shall not be unreasonably withheld, conditioned or delayed.

5.7 Refunds and Tax Treatment.

(a) Any Tax Refunds of or with respect to a Divested Company or the Sold Assets attributable to any Pre-Closing Tax Period received or credited to the Buyers or any Divested Company (other than any Tax Refund resulting from the carryback of a net operating loss or other Tax attribute from a taxable period beginning after the Closing Date to a Pre-Closing Tax Period, which Tax Refund shall be for the account of the Buyers) shall be for the account of the Company. Any Tax Refunds of or with respect to a Divested Company or the Sold Assets for any Straddle Period shall be equitably apportioned between the Company and the Buyers in accordance with the principles set forth in Section 5.5(b). All Tax Refunds of or with respect to a Divested Company or the Sold Assets attributable to any Post-Closing Tax Period shall be for the account of the Buyers. Each Party shall forward, and shall cause its Affiliates to forward, to the Party entitled to receive a Tax Refund pursuant to this Section 5.7(a) the amount of such Tax Refund within 10 days after such Tax Refund is received or utilized, net of (i) any reasonable out-of-pocket costs or expenses (including Taxes) incurred by such Party or its Affiliates in procuring such Tax Refund and (ii) any amounts of such Tax Refund that are legally required to be paid to any Person other than an Affiliate

of such Party. At the Company's request, the Buyers shall use commercially reasonable efforts to cooperate with the Company in obtaining such refunds, at the Company's expense, including through the filing of amended Tax Returns or refund claims as prepared by the Company; provided, that such cooperation shall not be required if it could reasonably be expected to have an adverse impact on any Buyer or its Affiliates in any Post-Closing Tax Period.

- (b) The Buyers and the Company agree to treat any amounts payable pursuant to Section 5.5, Section 5.6, this Section 5.7 and Article IX as an adjustment to the Purchase Price, unless otherwise required by applicable Law.
- 5.8 Post-Closing Tax Actions. Except as required by applicable Law to reflect a determination within the meaning of Section 1313(a) of the Code or any analogous or similar foreign, state or local Law, without the prior written consent of the Company, which consent shall not be unreasonably withheld, conditioned or delayed, the Buyers and their respective Affiliates shall not permit any Divested Company or any of their Affiliates to, (a) amend, refile, revoke or otherwise modify any Tax Return for a Pre-Closing Tax Period or Straddle Period, (b) make or change any Tax election or accounting period, method or practice with respect to, or that has retroactive effect to, any Pre-Closing Tax Period, or (c) make or initiate any voluntary contact with a Governmental Authority regarding any Pre-Closing Tax Period or Straddle Period with respect to substantive Tax matters (including by entering into any voluntary disclosure or similar program or arrangement and excluding, for the avoidance of doubt, the ordinary course filing of Tax Returns or the contesting of any Tax audits or proceedings).
 - 5.9 Employees; Employment Matters.
- Offers/Transfer of Employment. Prior to the Closing Date (or such time as is required by applicable Law), the Buyers or their respective Affiliate will make an offer of employment, in accordance with the requirements of this Section 5.9, to each Active Employee who is then employed by the Company or its Affiliate, to be effective as of the Closing Date (or, if later, the date such Active Employee returns from leave as set forth in this paragraph) unless such Active Employee is employed by a Divested Company. No later than five (5) days prior to the Closing Date, the Buyers will inform the Company of the identity of those Active Employees who received and accepted such offers. The Parties shall use commercially reasonable efforts to encourage all such Active Employees who are offered employment with a Buyer or its Affiliate as described in this Section 5.9(a) to accept such offers of employment, and the Company shall waive (or cause its applicable Affiliate(s) to waive) any and all restrictions (including those with respect to noncompetition, non-solicitation, and non-disclosure) that would prevent any Active Employee from assuming employment with any Buyer or its Affiliate or from performing any services on Buyers' or their Affiliates' behalf. The Company will provide, and cause its Affiliates to provide, such assistance as Buyers may reasonably request from time to time with respect to facilitating the offer and employee transition process described in this Section 5.9. All Active Employees who continue employment with or who accept an offer of employment with a Buyer, a Divested Company, or one of their respective Affiliates as of the Closing Date will cease employment with the Asset Seller or its applicable Affiliates (other than the Divested Companies) and begin employment with a Buyer or an Affiliate of such Buyer or continue employment with a Divested Company effective as of the Closing Date and are hereinafter referred to as the "Transferred Employees." Each Transferred Employee who is employed in the United States is referred to as a "U.S. Transferred Employee," and each Transferred Employee who is employed outside of the United States is referred to as a "Non-U.S. Transferred Employee." With respect to each Active Employee who is located in the

United States and not actively at work as of the Closing Date due to a long-term absence covered under a written long-term disability Seller Benefit Plan, the offer of employment from Buyer or its Affiliate described in this Section 5.9(a) shall be for employment effective as of the date that such Active Employee is able to return to his or her position (so long as such date is within 90 days following the Closing Date or such longer date as may be required by applicable Law), and any such Active Employee will be deemed to become a U.S. Transferred Employee as of the date he or she begins employment with Buyer or its Affiliate.

- (b) Cessation of Active Participation in Seller Benefit Plans. Effective as of the date the Asset Seller ceases to provide employee benefits to a Transferred Employee pursuant to the Transition Services Agreement or, in the case of an Active Employee described in the last sentence of Section 5.9(a) who has not become an employee of a Buyer or its Affiliate as of such date, the later date that he or she becomes an employee of a Buyer or its Affiliate (such date for such Transferred Employee, the "Benefits Transition Date"), except as required by Law or otherwise by this Section 5.9, all Transferred Employees will cease active participation in, and any benefit accrual under, each of the Seller Benefit Plans that is not sponsored by a Divested Company.
- (c) Compensation and Benefit Plans/Prior Service. As of the date that a Transferred Employee assumes employment with a Buyer or its Affiliate and for a period through the date that is one year after the Closing Date, for so long as a Transferred Employee continues employment during such period, such Buyer shall, or shall cause its Affiliate or a Divested Company to provide to each Transferred Employee, (i) base salary or hourly wages, as applicable, and annual cash incentive compensation opportunities that are no less favorable than provided to such Transferred Employee immediately prior to the Closing Date and (ii) employee benefit plans, programs, policies and arrangements that are substantially comparable in the aggregate to the Seller Benefit Plans (other than equity-based and nonqualified deferred compensation plans, programs and arrangements and defined benefit pension plans) covering such Transferred Employees as in effect immediately prior to the Closing. To the extent not otherwise required by or resulting from the operation of Law, the Buyers shall recognize or cause to be recognized each Transferred Employee's service with the Asset Seller, the applicable Divested Company or any of their Affiliates or predecessors as of the Closing as service with the applicable Buyer and its Affiliates for purposes of eligibility to participate in, vesting under, and accrual of benefits under any employee benefit plan, program or arrangement of such Buyer or its Affiliate in which Transferred Employees participate on or after the Closing Date (other than equity-based and other incentive compensation arrangements and defined benefit pension plans), but only to the extent such recognition does not result in duplication of benefits.
- (d) Severance Benefits. The Buyers shall, or shall cause their respective Affiliates or a Divested Company to, provide each US Transferred Employee severance benefits as provided on Schedule 5.9(d) upon any involuntary termination without "cause" (as such term is defined on Schedule 5.9(d)) that occurs during the one year period immediately following the Closing Date.
- (e) Welfare Plans. Coverage for all Transferred Employees and their respective dependents under the Seller Benefit Plans not sponsored by a Divested Company that are welfare benefit plans within the meaning of Section 3(1) of ERISA (the "Sellers' Welfare Plans") shall cease to be effective as of the applicable Benefits Transition Date. The Buyers shall, or shall cause their respective Affiliates to, as applicable, take commercially reasonable efforts to cause the plans sponsored by the Buyers or their respective Affiliates (including the Divested Companies after the Closing) that are welfare benefit plans within the meaning of Section 3(1) of ERISA (the "Buyer")

Welfare Plans") to provide coverage and benefits for all Transferred Employees and their respective eligible spouses and dependents effective as of the applicable Benefits Transition Date. Subject to the terms of the Transition Services Agreement, the Buyers and their respective Affiliates (including the Divested Companies after the Closing) and Buyer Welfare Plans shall be liable for all claims of any Transferred Employees and their respective eligible spouses and dependents incurred on or after the date such Transferred Employee becomes a Transferred Employee. Subject to the terms of the Transition Services Agreement, the Asset Seller and its Affiliates shall retain responsibility and liability for all claims of the Transferred Employees incurred before the date such Transferred Employee becomes a Transferred Employee; provided that the Divested Companies shall, with respect to and in accordance with the terms of any Seller Benefit Plan sponsored by a Divested Company, retain responsibility and liability for all claims of the Transferred Employees and any other participant covered by such plan whether incurred before, on, or after the Closing Date. For purposes of this Section 5.9(e), a claim shall be deemed "incurred" on the date that the event that gives rise to the claim occurs (for purposes of life insurance, severance, sickness, accident and disability programs) or on the date that the service was rendered or the supply was purchased (for purposes of health care programs) and all claims incurred during a continuous period of hospitalization shall be deemed to have occurred upon the commencement of such period. The Buyers shall, or shall cause the Divested Companies or its Affiliate to, take commercially reasonable efforts to waive any pre-existing condition limitations and eligibility waiting periods under the Buyer Welfare Plans (but only to the extent such pre-existing condition limitations and eligibility waiting periods were satisfied under the Sellers' Welfare Plans as of the date such Transferred Employee ceases to be employed by the Company or its Affiliate) and shall recognize (or cause to be recognized) the dollar amount of all expenses incurred by Transferred Employees and their respective dependents during the calendar year in which the Benefits Transition Date occurs under the Sellers' Welfare Plans for purposes of satisfying the deductibles and co-payment or out-of-pocket limitations for such calendar year under the relevant Buyer Welfare Plans. Within 30 Days after the applicable Benefits Transition Date of each Transferred Employee, Sellers shall provide to the Buyers all information necessary for the Buyers to comply with their obligations pursuant to this Section 5.9(e), and the Buyers' obligations under this Section 5.9(e) shall apply only to the extent such information is timely received by the Buyers.

- (f) 401(k) Plan Rollovers. Effective as of the Benefits Transition Date, the Buyers shall establish or provide a defined contribution plan qualified under section 401(a) of the Code ("Buyer 401(k) Plan") for the benefit of the U.S. Transferred Employees. The Buyers shall take commercially reasonable efforts to cause the Buyer 401(k) Plan to accept direct rollovers (described in Section 402(c) of the Code) of distributions to which the U.S. Transferred Employees are entitled under the TriMas Corporation Salaried Retirement Program and the TriMas Corporation Hourly Retirement Program including outstanding loan notes applicable to U.S. Transferred Employees.
- (g) Canadian Pension Plan. The Company shall cause Reike Canada ULC to do all things reasonably required to provide for Lamons Canada to cease to be recognized as a participating employer and for the Transferred Employees to cease participation under the TriMas Canadian Retirement Plan (the "Canadian Pension Plan") effective as of the Closing Date. The Company and its Affiliates shall retain responsibility for any and all liabilities and obligations (i) in respect of the defined contribution component of the Canadian Pension Plan that are incurred by Lamons Canada prior to the Closing Date, and (ii) in respect of the defined benefit component of the Canadian Pension Plan, and shall indemnify and hold harmless the Buyer Indemnified Persons from and against any Liability or Loss incurred by any Buyer Indemnified Person in respect thereof. If requested by the Buyers, the Company shall, or shall cause one of its Affiliates to, take

commercially reasonable efforts, subject to such approval of the applicable Governmental Authorities as may be required, to provide for the transfer of the defined contribution pension assets attributable to the applicable Transferred Employees from the Canadian Pension Plan to any successor defined contribution pension plan that may be established by the Buyers or their Affiliates (if any).

- (h) Accrued Vacation. The Buyers or their respective Affiliates shall, or shall cause the Divested Companies to, credit each Transferred Employee with the accrued and unused vacation days to which such Transferred Employee is entitled as of the time immediately before he or she becomes a Transferred Employee, and any personal and sickness days, or other service based leave entitlements, accrued by such employees as of the time immediately before such Transferred Employee becomes a Transferred Employee (in each case, after giving effect to whether Sellers or any of their Affiliates has made any payment with respect to any accrued but unused vacation or personal and sickness days as of the date such Transferred Employee ceases to be employed by the Company or its Affiliate).
- (i) Flexible Benefits. To the extent U.S. Transferred Employees participate in a dependent care or medical expense reimbursement account under a Seller Benefit Plan ("Seller FSA Plan") during the calendar year that includes the Benefits Transition Date, the Buyers shall take commercially reasonable efforts to establish or provide (or cause an Affiliate, if applicable, to establish or provide) one or more comparable plans ("Buyer FSA Plan") that will recognize the elections that such U.S. Transferred Employees had in effect for purposes of the plan year in which the Benefits Transition Date occurs under the Seller FSA Plan. The Buyer FSA Plan shall (i) assume the obligations of Seller FSA Plan with respect to U.S. Transferred Employees as of the applicable Benefits Transition Date and (ii) provide the same level of dependent care and medical expense reimbursement account benefits as those provided under the Seller FSA Plan at least through the end of the plan year in effect as of the Benefits Transition Date. As soon as practicable after the Benefits Transition Date, the Sellers shall transfer to the Buyers in cash an amount equal to the total amount that U.S. Transferred Employees have contributed to the Seller FSA Plan through the Benefits Transition Date for the calendar year that includes the Benefits Transition Date less all amounts that have been paid from the Seller FSA Plan through the Benefits Transition Date for medical expense and dependent care claims incurred by the U.S. Transferred Employees in the calendar year that includes the Benefits Transition Date (such difference, the "Flex Plan Amount"). If the Flex Plan Amount is less than \$0, as soon as practicable after the Benefits Transition Date, the Buyers shall transfer to the Sellers in cash an amount equal to all amounts that have been paid from the Seller FSA Plan through the Benefits Transition Date for medical expense and dependent care claims incurred by the U.S. Transferred Employees in the calendar year that includes the Benefits Transition Date less the total amount that U.S. Transferred Employees have contributed to the Seller FSA Plan through the Benefits Transition Date for the calendar year that includes the Benefits Transition Date. After the Benefits Transition Date, the Buyer FSA Plan will be responsible for reimbursement of all previously unreimbursed reimbursable medical expense and dependent care claims incurred by U.S. Transferred Employees, regardless of when the claims were incurred. Schedule 5.9(i) sets forth the account balance of each U.S. Transferred Employee under the Seller FSA Plan. Sellers will update Schedule 5.9(i) to reflect the current such account balances three Business Days prior to the Benefits Transition Date. Within 30 Days after the applicable Benefits Transition Date of each Transferred Employee, Sellers shall provide to the Buyers all information necessary for the Buyers to comply with their obligations pursuant to this Section 5.9(i), and the Buyers' obligations under this Section 5.9(i) shall apply only to the extent such information is timely received by the Buyers.

- (j) Continuation Coverage. Subject to the terms of the Transition Services Agreement, the Buyer shall have the sole responsibility to offer "continuation coverage" benefits from and after the date Transferred Employees become Transferred Employees to each Transferred Employee and the "qualified beneficiaries" of such Transferred Employees for whom a "qualifying event" occurs after the applicable Benefits Transition Date. The terms "continuation coverage," "qualified beneficiaries," and "qualifying event" shall have the meanings ascribed to them under Section 4980B of the Code and Sections 601-608 of ERISA.
- (k) Worker's Compensation. With respect to workers' compensation claims, from and after the date that a Transferred Employee assumes employment with Buyer or its Affiliate, the Buyers shall assume and become solely responsible (or cause the Divested Companies, if applicable, to assume and become solely responsible) for all workers' compensation claims of any Transferred Employee arising from injuries that relate entirely to an accident that occurred or injury that was identifiably sustained on or after the date that a Transferred Employee assumed employment with Buyer or its Affiliate.
- (I) Vesting. On or before the Benefits Transition Date of each Transferred Employee, the Asset Seller or its Affiliate shall take any necessary action to fully vest as of such date such Transferred Employee's account balances and other benefits under all Seller Benefit Plans, if any, that (i) are employee pension benefit plans (as such term is defined in Section 3(2) of ERISA), or (ii) provide nonqualified deferred compensation benefits.
- (m) Non-U.S. Business Employees. Notwithstanding anything to the contrary herein, any Active Employee who is employed in a jurisdiction outside of the United States immediately prior to the Closing (collectively, "Non-U.S. Business Employees"), and who is required by applicable Law to transfer to a Buyer or its Affiliates in connection with the transactions contemplated by this Agreement will transfer automatically on the Closing Date to such Buyer or its Affiliates, as applicable, in accordance with such applicable Law and shall be considered a Transferred Employee as of the Closing Date. Notwithstanding anything to the contrary herein, the following terms will apply, as indicated:
 - (i) To the extent that (A) the applicable Law of any jurisdiction, (B) any collective bargaining agreement, labor agreement or other agreement with a works council or economic committee, or (C) any employment agreement, would require the Buyers or their respective Affiliates to provide particular terms of employment to any Non-U.S. Transferred Employee that are different than those otherwise provided for by this Section 5.9 (or modify the period of time for which such standards are met), in connection with the transactions contemplated by this Agreement, then the Buyers will, or will cause one of their respective Affiliates to, provide such Non-U.S. Transferred Employee with such different term, and otherwise provide terms of employment in accordance with this Section 5.9.
 - (ii) To the extent provided under applicable Law, (A) any employment agreements between the Company, the Asset Seller, any Divested Companies or their Affiliates, on the one hand, and any Non-U.S. Transferred Employee, on the other hand, and (B) any collective bargaining agreements applicable to the Non-U.S. Transferred Employees, will, in each case, have effect after the Closing as if originally made between the Buyers and the other parties to such employment agreement or collective bargaining agreement.
 - (iii) With respect to any Non-U.S. Business Employee whose employment will not automatically transfer to a Buyer or its Affiliates in connection with the transactions

contemplated by this Agreement, the Company and the Buyers shall use their commercially reasonable efforts to effect the transfer of the employment of each such Non-U.S. Business Employee to the applicable Affiliate of a Buyer by methods as may reasonably be required or permitted to obtain the consent of such Non-U.S. Business Employee to the transfer and in a manner reasonably designed to minimize the obligation on the part of the Company and the Buyers to pay such Non-U.S. Business Employees any severance, end-of-service pay or gratuity, dismissal compensation, or other separation or termination payments or liabilities in connection therewith.

- (iv) Any severance, end-of-service pay or gratuity, dismissal compensation or other separation or termination payments or liabilities owed to a Non-U.S. Business Employee arising following the transactions contemplated by this Agreement will be liabilities of the Buyer and will be the sole responsibility of the Buyer.
- (v) The Buyers shall, or shall cause a Divested Company or one of their respective Affiliates to, assume any Seller Benefit Plan or portion of any Seller Benefit Plan that, under the applicable Law of the relevant jurisdiction, is required to transfer to the Buyers or such Divested Company or their respective Affiliates in connection with the transactions contemplated by this Agreement.
- (n) Cooperation. The Parties agree to furnish each other with such information concerning employees, employee payroll and employee benefit plans, subject to confidentiality and privacy considerations, and to take all such other action, as is necessary and appropriate to effect the transactions contemplated hereby.
- (o) No Right to Employment. Nothing herein expressed or implied shall confer upon any of the employees of the Company, the Buyers, the Divested Companies or any of their respective Affiliates (including any Active Employees), any additional rights or remedies, including any additional right to employment, or continued employment for any specified period, of any nature or kind whatsoever under or by reason of this Agreement.
- (p) No Third Party Beneficiary. No provision in this Section 5.9 shall (i) create any third party beneficiary or other rights in any employee or former employee (including any beneficiary or dependent thereof) of the Sellers, the Buyers, the Divested Companies or any other Person (including any Active Employees or any representatives thereof) other than the Parties and their respective successors and permitted assigns, (ii) constitute or create, or be deemed to constitute or create, an employment agreement or (iii) constitute or be deemed to constitute an amendment to any employee benefit plan sponsored or maintained by the Buyers, a Divested Company, the Company or any of their respective Affiliates.
- (q) Limitations. Notwithstanding anything to the contrary herein, (i) the terms of Section 5.9 shall not apply to the Active Employees set forth as numbers 1 and 6 on Schedule 1.1(b) (however if any such Active Employee becomes an employee of the Buyer or its Affiliates, Buyers shall use reasonable efforts to comply with the provisions of this Section 5.9 with respect to such Active Employee) and (ii) the terms of Section 5.9 shall govern with respect to actions to be taken by the Parties with respect to each other Specified Employee (as defined in the Transition Services Agreement) except that the terms "Closing Date" and "Closing" in Section 5.9(a) and the last sentence of Section 5.9(c) shall mean the earlier of the (i) last day of the Term (as defined in the Transition Services Agreement) or (ii) the date on which such Specified Employee commences employment with the Buyers or any of their respective Affiliates (such date the "Termination Date").

Buyers will be liable for, and will reimburse Sellers and their Affiliates for, any costs or expenses Sellers or their Affiliates incur for severance, end-of-service pay or gratuity, dismissal compensation or other separation or termination payments or liabilities or other separation related costs owed with respect to any Specified Employee who (i) does not become an employee of the Buyer or its Affiliates on the Termination Date and (ii) is employed by the Company or any of its Affiliates as of the Termination Date.

5.10 Labor Matters.

- (a) Collective Bargaining Agreements. Each of the Active Employees who is employed subject to the terms of the collective bargaining agreement (the "Collective Bargaining Agreement") listed on Schedule 5.10 is referred to herein as a "Union Employee." Notwithstanding any provision of Section 5.9 to the contrary, the terms of employment provided by the Buyers or their respective Affiliate pursuant to Section 5.9 and this Section 5.10 to Union Employees shall be: (i) as of the date such Union Employees become employed by a Buyer or its Affiliate, consistent with the terms of the Collective Bargaining Agreement (and consistent with an assumption of the Collective Bargaining Agreement to reflect a Buyer or its Affiliate) after giving effect to any amendment to such Collective Bargaining Agreement to reflect a Buyer or its applicable Affiliate (and not Seller Benefit Plans); and (ii) following the date such Union Employees become employed by Buyer or its Affiliate, consistent with the terms of any collectively bargained agreements between the Buyers and their respective Affiliates and any applicable collective bargaining representative of Union Employees as may be in effect from time to time (including any successor agreement to the Collective Bargaining Agreement). Neither Asset Seller nor any of its Affiliates shall make any representations to Union Employees regarding the terms of their post-Closing employment with the Buyers or their respective Affiliates that are inconsistent with the terms set forth in this Agreement.
- (b) Effects Bargaining. Prior to the Closing Date, Asset Seller will engage in good faith bargaining, if so requested by the applicable labor union and required by applicable Law, with the representative of the Union Employees as to the effects of the transactions contemplated by this Agreement, to the extent required by applicable Law. Upon the request of the Buyers, an in accordance with and as permitted by applicable Law, the Asset Seller will facilitate communications between the Buyers or their respective Affiliate, on the one hand, and the representative of the Union Employees, on the other hand, prior to the Closing Date.
- (c) Notifications. The Company will use commercially reasonable efforts to take, or cause to be taken, and to cause its Affiliates (including the Sellers) to take or cause to be taken, any and all actions in connection with any required notification to, or any required consultation with, the employees, employee representatives, work councils, unions, labor boards and relevant Governmental Authorities concerning the transactions contemplated by this Agreement with respect to Transferred Employees, and the Buyers will reasonably cooperate with the Company or its Affiliates in connection with the foregoing.
- 5.11 Contact With Customers and Suppliers. The Buyers shall not (and shall not permit their Affiliates or their representatives to) contact any employee, customer, supplier, licensor distributor or other material business relation of the Company or the Business regarding the Company, the Business or the transactions contemplated by this Agreement without the prior written consent of the Company (which consent shall not be unreasonably withheld, conditioned or delayed,

but may be conditioned upon a designee of the Company being present at any meeting or conference, consistent with the HSR Act and any other applicable antitrust or competition Laws); provided, however, that the Buyers, their respective Affiliates and their respective representatives may, without the Company's consent and without a designee of the Company being present, (i) contact such customers, suppliers, licensors or distributors of the Buyers' business in the ordinary course as long as such contact does not make reference to the Company or the Business in a manner that departs from ordinary course actions that Buyers would have undertaken absent the transactions contemplated by this Agreement and (ii) contact the Persons set forth on Schedule 5.11.

- 5.12 Non-Competition; Non-Solicitation and Confidentiality. As a material inducement for the Buyers to enter into this Agreement, and in order to protect the confidential and proprietary information and the goodwill that is conveyed by the Company hereunder, the Company hereby agrees that:
- (a) The Company shall not (and shall cause its Affiliates, including the Sellers, not to), directly or indirectly:
 - (i) during the three-year period commencing on the Closing Date, engage in any business in competition with the Business as conducted immediately prior to Closing in the geographic areas set forth on Schedule 5.12(a), which area the Company acknowledges represents the areas in which the Business sells products and services immediately prior to Closing (each, a "Competing Activity"); provided, that the foregoing shall not prohibit the Company or any of its Affiliates from:
 - (A) the acquisition of all or any portion of the Equity Interests of any Person engaged in a Competing Activity; provided, that the Competing Activity accounts for no more than 10% of the revenues of such Person (based on its latest annual financial statements);
 - (B) the acquisition, holding of investments or direct or indirect ownership of any voting stock, capital stock or other equity interest of any Person engaged in a Competing Activity, so long as such ownership interest represents less than 5% of the aggregate voting power or outstanding capital stock or other equity interests of such Person; or
 - (C) owning, engaging in, conducting or operating any of the Company's (or it Affiliates') retained businesses in the manner operated as of the Closing Date (so long as such businesses are not competitive with the Business);
 - (ii) during the three-year period commencing on the Closing Date, employ or solicit for employment (A) any individual employed in the Business at any time during such period or (B) any former employee of the Business who was employed in the Business during the three months prior to the Closing Date and was hired by the Buyers or any of their respective Affiliates on or after the Closing Date, in each case, without first obtaining the written consent of the Buyers (unless such employee is terminated by the Buyers or any Divested Company without cause after the Closing Date, in which case any such employee shall be deemed not to have been solicited for employment if such employee responded to a general solicitation not targeted at such employee or a group of any other current or former employees of the Business); or

- (iii) disclose to any third parties any trade secrets or other confidential or proprietary information relating to the Business.
- (b) Each Party (i) acknowledges that a breach or threatened breach of this Section 5.12 by a Party or its Affiliates would give rise to irreparable harm to the non-breaching Party, for which monetary damages would not be an adequate remedy, and (ii) hereby agrees that, in the event of a breach or a threatened breach by a Party or its Affiliates of any such obligations, the non-breaching Party shall, in addition to any and all other rights and remedies that may be available to it in respect of such breach, be entitled to seek equitable relief, including a temporary restraining order, an injunction, specific performance and any other relief that may be available from a court of competent jurisdiction (without any requirement to post bond).
- (c) The Parties mutually agree this Section 5.12 is reasonable in all respects and necessary to protect and preserve the Buyers' and the Company's legitimate business interests and the value of the goodwill conveyed by the Company hereunder, and the value of the Business, the Divested Companies and the retained businesses of the Company and its Affiliates, and to prevent any unfair advantage conferred on any Party and their respective successors. The Parties further agree and acknowledge that: (i) no proceeds shall be received or receivable by the Company or any of its Affiliates for granting the restrictive covenants under this Section 5.12; and (ii) the restrictive covenants under this Section 5.12 are integral to this Agreement and have been granted to maintain or preserve the fair market value of the Business conveyed hereunder.
- (d) It is the desire and intent of the Parties that the provisions of this Section 5.12 (and portions thereof) be enforced to the fullest extent permitted under applicable Law. The Company acknowledges that this Section 5.12 is reasonable and enforceable in all respects; nonetheless, if any of the restrictions of this Section 5.12 (or portions thereof) are found by an arbitrator or court of competent jurisdiction to be unreasonable, overly broad, or otherwise unenforceable, the Parties intend that such restrictions (and portions thereof) shall not be thereby terminated but shall be modified by such arbitrator or court so as to be valid and enforceable and, as so modified, to be fully enforced.

5.13 Further Actions.

(a) Notwithstanding anything in this Agreement to the contrary, this Agreement shall not constitute an agreement to sell, convey, assign or transfer any Sold Asset if any attempted sale, conveyance, assignment or transfer of such Sold Asset, without the consent of another Person to such transfer, would constitute a violation of applicable Law or breach of any Contract or Permit with respect to such Sold Asset if such consent shall not have been received (each, a "Non-Assigned Asset"). If any required consent is not obtained on or prior to the Closing, then such Non-Assigned Asset shall be deemed an "Excluded Asset" pursuant to this Agreement at the time of Closing, and the Company shall, and shall cause the Asset Seller to, use their respective commercially reasonable efforts to (i) enter into a lawful arrangement with the Asset Buyer (or its designee(s)), pursuant to which the Asset Buyer (or its designee(s)) would receive the same economic claims, rights and benefits associated with such Non-Assigned Asset, (ii) enforce, at the request of the Asset Buyer and for the account of the Asset Buyer (or its designee(s)) and at the Asset Buyer's expense, any rights of the Asset Seller arising from any such Non-Assigned Asset (including, in the case of any Contract, the right to elect to terminate such Contract in accordance with the terms thereof upon the written request of the Asset Buyer), and (iii) assist the Asset Buyer (or its designee(s)) in obtaining any such Non-Assigned Asset as quickly as possible following the Closing.

If any such Non-Assigned Asset is subsequently sold, conveyed, assigned or transferred to the Asset Buyer (or its designee(s)) following the Closing, such Non-Assigned Asset shall, as of the date of such sale, conveyance, assignment or transfer, be deemed a "Sold Asset" pursuant to this Agreement.

- (b) The Parties shall cooperate to prepare as soon as reasonably practicable any and all Ancillary Agreements not prepared as of the Execution Date. In addition, the Parties hereby agree that each agreement, arrangement or other instrument as shall be required under applicable Law in order to transfer the Sold Assets, the Assumed Liabilities, and the Shares shall include only those representations, warranties and indemnities provided for in this Agreement and such other provisions as are required by applicable Law to give effect to such transfer. It is the intention of the Parties, notwithstanding the provisions of any such agreement, arrangement or other instrument, that no purchase and sale contemplated by any such agreement, arrangement or instrument shall be consummated earlier than simultaneous with the Closing. Accordingly, each of the Parties will take such action as may be necessary to ensure that no closing under any such agreement, arrangement or instrument occurs prior to the Closing.
- (c) The Buyers acknowledge that all insurance policies maintained by the Company and its Affiliates may be terminated as of Closing with respect to the Sold Assets, the Divested Companies and the Business, but solely for coverage for claims made and occurrences after Closing. In respect of any claim made or occurrence prior to Closing, the Buyers and the Divested Companies, subject to the terms and conditions of the applicable policies, may make claims on any of the Company's or its Affiliates' insurance policies covering claims made or occurrences prior to the Closing. The Company shall (and shall cause its Affiliates to) maintain or cause to be maintained all such insurance policies for such purposes and use commercially reasonable efforts to assist the Buyers and the Divested Companies in asserting any such claims and make the benefits of any such insurance policies available to the Buyers and the Divested Companies; provided, that the Buyers shall be solely liable for, and the Company shall have no obligation to pay or reimburse the Buyers or the Divested Companies for, all deductibles and retentions associated with such claim and, solely to the extent such claim relates to facts and circumstances that do not include (A) a breach of any of the representations and warranties set forth in Article III of this Agreement or (B) an obligation of the Company to otherwise indemnify the Buyers pursuant to Article IX, Buyer shall be responsible for, and shall reimburse the Company for, any reasonably documented increase in insurance premiums resulting from such claim.
- 5.14 Bulk Transfer Laws. The Buyers and the Company hereby waive compliance with any bulk transfer Laws applicable to the transactions contemplated by this Agreement.

5.15 Confidentiality.

- (a) The Buyers hereby confirm and agree that, with respect to any information directly or indirectly furnished by or on behalf of any Seller, whether before, on or after Execution Date, the Buyers shall continue to be bound by the terms of the Confidentiality Agreement, which shall continue in full force and effect pursuant to the terms thereof until, and shall terminate upon, the Closing.
- (b) The Buyers understand and agree that the Company is making available confidential information and trade secrets to the Buyers concerning the operations of the Company, the Sellers, the Divested Companies and the Business, which information may be damaging to the Company, the Sellers, the Divested Companies and their Affiliates if disclosed to a competitor or

made available to any other Person in violation of the Confidentiality Agreement, and that such information has been divulged in confidence. The Buyers acknowledge that after the Closing Date, the Company, the Sellers and their Affiliates could be irreparably damaged if any nonpublic or proprietary information about the Company, the Sellers or their Affiliates that does not relate to the Business, the Sold Assets, the Assumed Liabilities or the Divested Companies were disclosed by the Buyers or their respective Affiliates after the Closing and in violation of the Confidentiality Agreement, and accordingly, the Buyers will not, and will cause their respective officers, directors, employees and other Affiliates not to, for a period of two years following the Closing Date, without the prior written consent of the Company, disclose or use (or permit to be disclosed or used) in any way any such information, unless (i) compelled to disclose such confidential information by judicial or administrative process or, in the opinion of its counsel, by other requirements of Law and, in any such event, the Buyers shall, to the extent practicable, give the Company prompt written notice of any such requirement prior to any such disclosure, (ii) such confidential information is generally available to the public through no fault of the Buyers or any of their respective Affiliates, or (iii) such confidential information is publicly disclosed by the Buyers or their respective Affiliates with the Company's prior written consent.

- (c) The Company acknowledges that the Buyers and their Affiliates could be irreparably damaged if any nonpublic or proprietary information about the Buyers, their respective Affiliates, the Divested Companies, the Sold Assets or the Assumed Liabilities were disclosed by the Company or its Affiliates, and accordingly, the Company will not, and will cause its officers, directors, employees and other Affiliates not to, beginning on the Execution Date and for a period of two years following the Closing Date, without the prior written consent of the Buyers, disclose or use (or permit to be disclosed or used) in any way any such information, unless (i) compelled to disclose such confidential information by judicial or administrative process or, in the opinion of its counsel, by other requirements of Law and, in any such event, the Company shall, to the extent practicable, give the Buyers prompt written notice of any such requirement prior to any such disclosure, (ii) such confidential information is generally available to the public through no fault of the Company or any of its Affiliates, or (iii) such confidential information is publicly disclosed by the Company or its Affiliates with the Buyers' prior written consent.
- (d) On or before the Closing Date, the Company shall, and shall cause its Affiliates and their respective representatives to, deliver "return or destroy" notices to each counterparty under any confidentiality agreement entered into by the Company, any Seller or any of their respective Affiliates with respect to the sale of all or any portion of the Business, the Sold Assets, the Shares or any of the assets or properties of any Divested Company (the "Sale Process Confidentiality Agreements"). The Company shall, and shall cause its Affiliates to, use their respective commercially reasonable efforts to enforce, at the request and direction of the Buyers and at the Buyers' sole cost and expense, any rights of the Company or any of its Affiliates arising from any Sale Process Confidentiality Agreements. At or promptly following the Closing, the Company shall deliver to the Buyers (i) a true and complete copy of the form confidentiality agreement used with respect to the Sale Process Confidentiality Agreements and (ii) a list of counterparties that entered into the Sale Process Confidentiality Agreements (but solely to the extent the disclosure of such counterparties is permitted by the terms and conditions of such Sale Process Confidentiality Agreements).
- 5.16 Exclusivity. From and after the Execution Date until the Closing Date or the earlier termination of this Agreement pursuant to Section 8.1, the Company shall not, and shall cause the Sellers, the Divested Companies and their respective Affiliates, officers managers,

employees and representatives not to, directly or indirectly, (a) solicit, initiate or encourage any inquiry, proposal or offer from any Person relating to any transaction involving the sale of all or any material part of the Business, the Shares, the Sold Assets or any of the assets or properties of any Divested Company, or any merger, consolidation, business combination or similar transaction involving the Business, the Asset Seller or any Divested Company (each, an "Acquisition Transaction"), (b) participate in any discussions or negotiations or enter into any Contract with, or provide any information to, any Person (other than the Buyers, their respective Affiliates and their respective representatives) relating to or in connection with a possible Acquisition Transaction or facilitate an Acquisition Transaction in any manner, or (c) accept any proposal or offer from any Person (other than the Buyers, their respective Affiliates and their respective representatives) relating to a possible Acquisition Transaction; provided, that for the avoidance of doubt, a transaction (or series of transactions) to acquire voting control of the Company or substantially all of the assets of the Company (including by merger) shall not constitute an "Acquisition Transaction." The Company shall promptly inform the Buyers in writing of any offers or expressions of interest to pursue an Acquisition Transaction that are made during the Interim Period and the material terms thereof.

5.17 Names Following Closing.

- (a) After the Closing, the Buyers and their respective Affiliates will not market, promote or advertise any products or services using the name "TRIMAS" or "RIEKE" or any marks confusingly similar with any of the foregoing, or any other names owned by the Company and its Affiliates that are set forth on Schedule 5.17 (the "Company Marks"), alone or in combination with other names or marks, including as a brand name, trademark, service mark, service name, logo, domain name or web address. Notwithstanding the foregoing, it shall not be deemed a violation of this Section 5.17 by reason of any Buyer's or its Affiliate's or representative's use of the Company Marks (i) in a neutral, non-trademark manner (e.g., for purposes of conveying to customers, any Governmental Authority or the general public that the name of Divested Companies' respective businesses have changed, that there was a change in ownership of the Business or the historical origins of the Business, the Sold Assets or the Divested Companies), (ii) to the extent required by Law, (iii) on internal office supplies or signage not visible to consumers or the general public, provided that such supplies or signage are replaced promptly in the ordinary course of business, or (iv) to the extent in legal or business documents already in existence on the Closing Date.
- (b) As soon as practicable and no later than six months following the Closing (other than with respect to Lamons Gasket (Zhangjiagang) Co. Ltd., which shall be as soon as practicable and no later than 18 months following the Closing), the Company will (and will cause its Affiliates to) change the legal names of its retained subsidiaries so as not to include "LAMONS" and will (and will cause its Affiliates to) amend the organizational documents of its retained subsidiaries as necessary to reflect such name changes. After the Closing, the Company and its Affiliates will not use the name "LAMONS," and any other word, expression or identifier of source confusingly similar thereto or constituting an abbreviation, derivation or extension thereof, alone or in combination with other names or marks, including as a brand name, trademark, service mark, service name, logo, domain name or web address. Each Seller acknowledges that, after the Closing Date, it and its Affiliates have no rights whatsoever to the Acquired Intellectual Property.
- 5.18 Notification of Certain Matters. The Company, on the one hand, and the Buyers, on the other hand, shall give prompt notice to the other (a) of the occurrence, or failure to occur, of any event of which the occurrence or failure to occur would cause any of such Party's

representations or warranties contained in this Agreement to be untrue or inaccurate at any time from the Execution Date to the Closing Date, (B) any failure on such Party's part to comply with or satisfy any covenant or agreement to complied with or satisfied by such Party hereunder and (c) the occurrence of any event that may make the satisfaction of the conditions in Article VI or Article VII, as the case may be, impossible; provided, however, that no notification or failure to notify pursuant to this Section 5.18 shall affect or be deemed to modify any representations or warranties made herein or the conditions to the obligations of the respective Parties to consummate the transactions contemplated hereby.

- 5.19 Intercompany Obligations. The Company shall, and shall cause the Sellers and their respective Affiliates to, take such action and make such payments as may be necessary so that, as of the Closing Date, there shall be no Intercompany Obligations. Immediately prior to the Closing, the Company shall, and shall cause the Sellers to, terminate or cause to be terminated all of the Contracts required to be set forth on Schedule 3.26. No such Contract (including any provision thereof which purports to survive termination) shall be of any further force or effect after the Closing, and all Divested Companies shall be released from all Liabilities thereunder effective as of the Closing.
- 5.20 Post-Closing Conduct. From and after the Closing, (a) the Company shall cause the Asset Seller not to liquidate, dissolve, or enter into any Proceeding relating to bankruptcy, insolvency, liquidation or dissolution without the Buyers' prior written consent and the Company shall cause the Asset Seller to do or cause to be done all things reasonably necessary to preserve, renew and keep in full force and effect the Asset Seller's legal existence (provided, that with the Buyers' prior written consent (not to be unreasonably withheld, conditioned or delayed), the Asset Seller shall be permitted to redomesticate to a different jurisdiction in the United States), and (b) if any Buyer or any of its respective Affiliates are named in any Proceeding or Proceedings involving claims for personal injury or death related to any Specified Excluded Liability, then the Company shall cause the Asset Seller to use commercially reasonable efforts, in cooperation with such Buyer, to seek dismissal of such Buyer as a party to such Proceeding. Notwithstanding anything to the contrary in this Agreement, this Section 5.20 shall be further subject to the procedures set forth on Schedule 9.6 with respect to the Specified Excluded Liabilities.

5.21 Financing.

(a) The Buyers shall use their reasonable best efforts to obtain the proceeds of the Financing on terms and conditions described in the Commitment Letters and the Fee Letter (including any "market flex" provisions in the Fee Letter), including reasonable best efforts to (i) maintain in effect the Commitment Letters (subject to the Buyers' right to amend, modify, supplement, restate, assign, substitute or replace the Debt Commitment Letter in accordance herewith), (ii) negotiate definitive agreements with respect to the Debt Financing having terms and conditions no less favorable to the Buyers than the terms and conditions contained in the Debt Commitment Letter and Fee Letter (including any "market flex" provisions contained in the Fee Letter) or on other terms that are (A) acceptable to the Lenders and (B) in the aggregate not materially less favorable, taken as a whole, to the Buyers (unless otherwise agreed by the Buyers) and (iii) satisfy (or obtain a waiver of) on a timely basis all conditions in the Commitment Letters that are reasonably within the Buyers' control in accordance with the Debt Commitment Letter. In the event that all conditions contained in each Commitment Letter have been satisfied (or upon funding will be satisfied) and all closing conditions contained in Article VI and Article VII of this Agreement have been satisfied or waived, to the extent permitted by the party entitled to the benefit

thereof, the Buyers shall use their reasonable best efforts to cause each Lender or Equity Investor under such Commitment Letter to fund its committed portion of the Financing required to consummate the transactions contemplated by this Agreement and to pay related fees and expenses at the Closing. The Buyers shall not, without the prior written consent of the Company (not to be unreasonably withheld, conditioned or delayed), (A) permit any amendment, supplement or modification to, or any waiver of any material provision or remedy under, or replace, the Commitment Letters if such amendment, supplement, modification, waiver or replacement (1) would be reasonably expected to make the timely funding of the Financing or satisfaction of the conditions to obtaining the Financing materially less likely to occur, (2) reduces the amount of the Financing such that the Buyers do not have sufficient cash proceeds to consummate the transactions contemplated by this Agreement and to pay related fees and expenses at Closing, (3) materially and adversely affects the ability of the Buyers to enforce their rights against any of the other parties to the Commitment Letters as so amended, supplemented, modified, waived or replaced, relative to the ability of the Buyers to enforce its rights against any of such other parties to the Commitment Letters as in effect on the Execution Date or (4) adds new (or modifies any existing) conditions to the consummation of all or any portion of the Financing in a manner that would reasonably be expected to prevent or materially delay the consummation of the transactions contemplated by this Agreement; provided, that the Buyers may amend the Debt Commitment Letter to add lenders, lead arrangers, bookrunners, syndication agents or similar entities that have not executed the Debt Commitment Letter as of the Execution Date or (B) terminate any Commitment Letter (unless such Commitment Letter has been replaced by a new commitment letter (or similar agreement) in accordance herewith). The Buyers shall give the Company reasonably prompt written notice (A) of any material breach or default (or any event or circumstance that, with or without notice, lapse of time or both, would reasonably be expected to result in breach or default) by any party to any Commitment Letter or other Debt Document of which Buyer becomes aware that, in each case, would be reasonably expected to result in the Debt Financing not being provided at Closing in an amount sufficient for the Buyers to comply with their obligations under this Agreement, (B) if and when Buyer becomes aware that any portion of the Financing contemplated by any Commitment Letter may not be available that would be reasonably expected to result in the Debt Financing not being provided at Closing in an amount sufficient for the Buyers to comply with their obligations under this Agreement, (C) of the receipt of any written notice or other written communication from any Person with respect to any (1) actual or potential breach, default, termination or repudiation by any party to any Commitment Letter or other Debt Document or (2) material dispute or disagreement between or among any parties to any Commitment Letter or other Debt Document (but excluding, for the avoidance of doubt, any ordinary course negotiations with respect to the terms of the Financing or Debt Documents), in each case of this clause (C), that would be reasonably expected to result in the Debt Financing not being provided at Closing in an amount sufficient for the Buyers to comply with their obligations under this Agreement, (D) if for any reason Buyer believes in good faith that it will not be able to obtain any portion of the Financing on the terms, in the manner and from the sources contemplated by any Commitment Letter (including, without limitation, after giving effect to any related flex terms) or the definitive agreements with respect thereto (such definitive agreements related to the Debt Financing, collectively, with the Debt Commitment Letter, the "Debt Documents"), in each case, that would be reasonably expected to result in the Financing not being provided at Closing in an amount sufficient for the Buyers to comply with their obligations under this Agreement, and (E) of any expiration or termination of any Commitment Letter or other Debt Document. Upon any such amendment, supplement, modification, waiver or replacement of the Debt Commitment Letter in accordance with this Section 5.21(a), the terms "Debt Commitment Letter" and "Debt Financing" shall refer to the Debt Commitment Letter as so amended, supplemented, modified, waived or replaced and the debt financing contemplated thereby. If any

portion of the Debt Financing becomes, or would reasonably be expected to become, unavailable on the terms and conditions contemplated in the applicable Commitment Letter (after taking into account flex terms) and such unavailable amounts are required for the Buyers to make any required payment under this Agreement, Buyer shall, if no other equity or debt financing is available pursuant to binding and enforceable agreements to cover such unavailable amounts at such time use its reasonable best efforts to arrange to obtain alternative financing, including from alternative sources, on terms in the aggregate not materially less favorable to Buyer (unless otherwise agreed by the Buyers) than the Debt Financing contemplated by the Debt Commitment Letter in an amount sufficient to replace any unavailable portion of the Debt Financing ("Alternative Financing") as promptly as practicable following the occurrence of such event. Notwithstanding the foregoing or anything else in this Agreement to the contrary, in no event shall the term "reasonable best efforts" as used in this Section 5.21 be deemed or construed to require the Buyers to, and the Buyers shall not be required to, (x) pay any fees in excess of those contemplated by the Debt Commitment Letter and Fee Letter or (y) agree to conditionality or economic terms of the Debt Financing that are less favorable in the aggregate than those contemplated by the Debt Commitment Letter or the Fee Letter (including any "market flex" provisions therein).

- (b) The Buyers shall provide the Company with prompt written notice (i) of (A) any material breach or default by any party to any Commitment Letters of which the Buyers become aware or (B) the receipt of any written notice or other written communication from any Lender or Equity Investor with respect to any material breach, default, termination or repudiation by any party to any Commitment Letters or of any provision thereof, and (ii) if at any time for any reason the Buyers reasonably believe in good faith that they will not be able to obtain all or any portion of the Financing on the terms and conditions, in the manner or from the sources contemplated by any of the Commitment Letters reasonably necessary to permit the Buyers to consummate the transactions contemplated by this Agreement and to pay related fees and expenses at the Closing.
- (c) The Buyers acknowledge and agree that the obtaining of the Financing or any Alternative Financing is not set forth as a condition to Closing in Article VII.
- (d) Prior to the Closing, the Company shall, and shall cause its Affiliates (including the Sellers and each of the Divested Companies) to, and shall use their reasonable best efforts to cause their respective non-legal representatives, to cooperate as may be reasonably and customarily requested by the Buyers in connection with the arrangement of the Debt Financing (provided, that such requested cooperation does not unreasonably interfere with the ongoing operations of the Company and its Subsidiaries), including using their reasonable best efforts in the following:
 - (i) participating in and making senior management of the Company and its Affiliates available, at reasonable times, for a reasonable number of meetings, presentations and sessions with rating agencies;
 - (ii) providing authorization letters to Debt Financing Sources authorizing the distribution of information to prospective lenders or investors (including customary 10b-5 and material non-public information representations);
 - (iii) assisting the Buyers and the Debt Financing Sources in the preparation of syndication memoranda, bank information memoranda or other marketing materials and memoranda for the Debt Financing, including customary lender presentations and a confidential information memorandum to be used in the syndication of the Debt Financing,

materials for rating agency presentations and business and financial projections reasonably requested by the Buyers;

- (iv) furnishing the Buyers and the Debt Financing Sources as promptly as reasonably practicable following request therefor such financial, business, and other information regarding the Business and customarily required in connection with the execution of financings of a type similar to the Debt Financing or otherwise reasonably requested by the Buyers in connection therewith, including: (A) unaudited consolidated balance sheet of the Business as of December 31, 2018, December 31, 2017, and December 31, 2016, and the related statements of income of the Business for the fiscal years then ended, (B) the unaudited consolidated balance sheet of the Business as of the Balance Sheet Date, (C) to the extent available, unaudited balance sheets of the Business as of the end of any fiscal quarter ended at least 45 days before the Closing Date and income statements for any such fiscal quarter, (D) a quality of earnings analysis with respect to the Business, dated as of September 6, 2019, and prepared by Deloitte Touche Tohmatsu Limited, and (E) information reasonably necessary for the Buyers to prepare a customary pro forma combined balance sheet and related pro forma combined statement of income as of and for the twelve-month period ending on the last day of the most recently completed four-fiscal guarter period ended at least 45 days before the Closing Date, or if the most recently completed fiscal period is the end of a fiscal year, ended at least 90 days before the Closing Date, prepared after giving effect to the transactions contemplated hereby, in each case of the foregoing clauses (A)-(E) prepared in accordance with GAAP (all such financial statements and information in Sections 5.21(c)(iv)(A)-(E), together with (1) any replacements or restatements thereof, and any supplements thereto and (2) authorization letters referred to in clause (ii) above, collectively, the "Required Information") (it being understood that none of the Company or any of its Subsidiaries shall be required to provide, and Buyer shall be solely responsible for, the preparation of pro forma financial information, projections, risk factors or other forward-looking statements relating to all or any component of the Financing);
- (v) executing and delivering a certificate of the chief financial officer (or other comparable officer) with respect to solvency of the Business (after giving effect to the transactions contemplated hereby) and other customary certificates or documents and back-up therefor and for legal opinions as may be reasonably requested by the Buyers or Debt Financing Sources and otherwise reasonably facilitating the pledging or granting of any liens or security interests on any collateral;
- (vi) furnishing the Buyers and the Debt Financing Sources promptly, and, in any event, at least three Business Days prior to the Closing Date, with all documentation and other information that any Debt Financing Source has requested in writing at least ten days prior to the Closing Date and that such Debt Financing Source has reasonably determined is required by regulatory authorities under applicable "know your customer," "beneficial ownership" and anti-money laundering rules and regulations, including the PATRIOT Act and 31 C.F.R. § 1010.230;
- (vii) delivering lien terminations and instruments of discharge to be delivered at the Closing and giving any other notices, releases or terminations requested by the Buyers in order to allow for the release of all claims and Encumbrances (other than the Permitted Encumbrances) that are burdening the Sold Assets or the Shares;

- (viii) cooperating with the Buyers' legal counsel in connection with any legal opinions that such legal counsel may be required to deliver in connection with the Debt Financing;
- (ix) cooperating with the Buyers and the Buyers' respective efforts to obtain corporate and facilities ratings, consents, landlord waivers and estoppels, non-disturbance agreements as reasonably requested by the Buyers;
- (x) solely to the extent necessary, and to the extent no Liability to the Company or any of its Affiliates could arise therefrom, causing the organizational documents of the Companies to be amended in a manner to permit the Debt Financing and the security interests granted in connection therewith, effective as of Closing; and
- (xi) otherwise cooperating with the Buyers to satisfy the conditions precedent to the Debt Financing to the extent within the control of the Company and its Affiliates.

provided, that (A) none of the Company or any of its Subsidiaries shall be required to incur any liability in connection with the Financing prior to the Closing, (B) the pre-Closing Board of Directors of the Company and the directors, managers and general partners of the Company's Subsidiaries shall not be required to adopt resolutions approving the agreements, documents and instruments pursuant to which the Financing is obtained, (C) none of the Company or any of its Subsidiaries shall be required to execute prior to the Closing any definitive financing documents, including any credit or other agreements, pledge or security documents, or other certificates, legal opinions or documents in connection with the Financing, and (D) solely except to the extent necessary to comply with Section 5.21(c)(x) above, neither the Company nor any of its Subsidiaries shall be required to take any corporate actions prior to the Closing to permit the consummation of the Financing, The foregoing notwithstanding, none of the Company or its Affiliates shall be required to pay any commitment or other similar fee or incur any other cost or expense that is not subject to the expense reimbursement provision contained in Section 5.21(d) in connection with the Debt Financing prior to the Closing.

- (e) The Buyers shall, promptly upon written request by the Company, reimburse the Company for all reasonable and documented out-of-pocket costs incurred by the Company, its Affiliates or their respective representatives in connection with such cooperation and shall indemnify and hold harmless the Company and its Affiliates and their respective representatives for and against any and all losses actually suffered or incurred by them in connection with the arrangement of the Debt Financing, any action taken by them at the request of the Buyers pursuant to this Section 5.21 and any information utilized in connection therewith (other than information provided by the Company or its Affiliates or representatives), in each case, except to the extent (i) suffered or incurred as a result of any such indemnitee's, or such indemnitee's respective Representative's, gross negligence, bad faith, willful misconduct or material breach of this Agreement, or (ii) with respect to any material misstatement or omission in information provided hereunder by any of the foregoing Persons.
- (f) Except for the representations and warranties of the Company set forth in Article III of this Agreement and in the certificate delivered pursuant to Section 7.3, the Company shall not have any liability to the Buyers in respect of any Financial Information provided pursuant to this Section 5.21. The Buyers shall promptly reimburse the Company and its Subsidiaries for all reasonable and documented out-of-pocket costs incurred by the Company or its Affiliates in

connection with such cooperation. Subject to the Buyers' indemnification obligations under this Section 5.21, the Company hereby consents to the use of all of its and its Affiliates' corporate logos in connection with the initial syndication or marketing of the Debt Financing; provided, that such logos are used solely in a manner that is not intended to or reasonably likely to harm or disparage the Company or its Subsidiaries or the reputation or goodwill of the Company or any of its Subsidiaries.

(g) The Buyers and the Debt Financing Sources may disclose confidential information regarding the Company or its Affiliates (including, without limitation, with respect to any of the Sold Assets, the Divested Companies, or the Business) obtained by the Buyers or their respective Representatives pursuant to this Section 5.21 to (i) potential lenders and investors (including, without limitation, the Debt Financing Sources) in connection with the marketing and syndication of the Debt Financing and (ii) rating agencies in connection with confirming or obtaining ratings for the Buyers or the Debt Financing; provided, that (i) each of the foregoing are reasonably informed by the Buyers that such information is being disclosed on a confidential basis and (ii) the Parties acknowledge and agree that the disclosure of such information shall be subject to the terms and conditions of the Confidentiality Agreement, including that the Buyers and their Affiliates shall assume responsibility for any disclosure of such information by the recipients thereof in contravention of the Confidentiality Agreement.

5.22 R&W Insurance Policy.

- (a) As of the Execution Date, the Buyers have obtained and conditionally bound the R&W Insurance Policy. Prior to the Closing, each Party shall use its commercially reasonable efforts to satisfy the conditions set forth in the binder agreement for the R&W Insurance Policy as of the Closing. The R&W Insurance Policy shall provide that the R&W Insurer shall have no right of subrogation against the Company or any of its Affiliates, and that the R&W Insurer has waived any such right of subrogation, except in the case of Fraud. For the avoidance of doubt, the Buyers acknowledge and agree that the obtaining of the R&W Insurance Policy is not a condition to the Closing and the Buyers shall remain obligated, subject only to the satisfaction or waiver of the conditions set forth in Article VII of this Agreement, to consummate the transactions contemplated by this Agreement. The Buyers shall pay all R&W Insurance Costs as such R&W Insurance costs are due and payable.
- (b) The Buyers shall not agree to any amendment, variation or waiver of the R&W Insurance Policy which would adversely impact the Company or the Sellers without the prior written consent of the Company. From and after the Closing, the Company shall, and shall cause the Sellers to, use commercially reasonable efforts to cooperate with the Buyers and their respective Affiliates as may be reasonably requested from time to time in connection with any claim made by the Buyers or their respective Affiliates under the R&W Insurance Policy. Notwithstanding anything to the contrary in this Agreement, none of (i) the Company, (ii) the Sellers, (iii) their respective Affiliates or (iv) their respective officers, directors, employees, agents or representatives, shall be entitled to any proceeds from the R&W Insurance Policy.
- 5.23 Books and Records. The Company shall, as soon as practicable following the Closing, deliver and cause its Affiliates to deliver to the Buyers all organizational documents, minute and record books, bank books, registered company seal(s) and other books and records of the Divested Companies.

5.24 Releases.

- (a) Effective as of the Closing, the Company, on behalf of itself and each of its Affiliates, hereby releases and forever discharges each of the Buyers and the Divested Companies and each of their respective controlled Affiliates, and any individual, joint or mutual, past, present and future representatives, stockholders, partners, members, controlling persons, subsidiaries, successors and assigns of any of the foregoing (individually, a "Buyer Releasee" and, collectively, "Buyer Releasees"), from any and all claims, demands, proceedings, causes of action, orders, obligations, contracts, agreements, debts and liabilities whatsoever, whether known or unknown, suspected or unsuspected, both at law and in equity, which the Company or any of its Affiliates, or any of their respective heirs, executors, administrators or assigns, now has, has ever had, or may hereafter have against any Buyer Releasee arising prior to the Closing or on account of or arising out of any matter, cause or event occurring prior to the Closing, in each case, related to the Business, the Sold Assets or the Divested Companies (all of the foregoing collectively referred to herein as the "Buyer Released Claims"); provided, however, that nothing contained herein shall operate to release any obligations of the Buyers or the Divested Companies or any of their Respective Affiliates (i) under this Agreement or under any Ancillary Agreement executed and delivered to the Company by the Buyers at the Closing in connection with the transactions contemplated hereby and thereby and (ii) under the organizational documents of the Divested Companies or any insurance policies, in each case, to the extent related to director and officer indemnification obligations.
- (b) The Company hereby acknowledges and intends that this release shall be effective as of the Closing to bar to each and every one of the Buyer Released Claims hereinabove mentioned or implied. The Company expressly consents that this release shall be given full force and effect in accordance with each and every express term or provision, including those (i) relating to any Buyer Released Claims hereinabove mentioned or implied or (ii) relating to unknown and unsuspected Buyer Released Claims (notwithstanding any state statute that expressly limits the effectiveness of a general release of unknown, unsuspected and unanticipated Buyer Released Claims).
- (c) Except in the case of Fraud, effective as of the Closing, each of the Buyers, on behalf of themselves and each of their Affiliates (including, after the Closing, the Divested Companies), hereby releases and forever discharges the Company each of its respective Affiliates, and any individual, joint or mutual, past, present and future representatives, stockholders, partners, members, controlling persons, subsidiaries, successors and assigns of any of the foregoing (individually, a "Seller Releasee" and, collectively, "Seller Releasees"), from any and all claims, demands, proceedings, causes of action, orders, obligations, contracts, agreements, debts and liabilities whatsoever, whether known or unknown, suspected or unsuspected, both at law and in equity, which the Buyers or any of their Affiliates (including, after the Closing, the Divested Companies), or any of their respective heirs, executors, administrators or assigns, now has, has ever had, or may hereafter have against any Seller Releasee arising prior to the Closing or on account of or arising out of any matter, cause or event occurring prior to the Closing, in each case, related to the Business, the Sold Assets or the Divested Companies (all of the foregoing collectively referred to herein as the "Seller Released Claims"); provided, however, that nothing contained herein shall operate to release any obligations of the Company, Seller Guarantor or any of their respective Affiliates (i) under this Agreement or under any Ancillary Agreement executed and delivered to the Buyers by the Company at the Closing in connection with the transactions contemplated hereby and thereby and (ii) under the organizational documents of the Company,

Seller Guarantor or any of their respective Affiliates or any insurance policies, in each case, to the extent related to director and officer indemnification obligations.

- (d) The Buyers and the Divested Companies hereby acknowledge and intend that this release shall be effective as of the Closing to bar to each and every one of the Seller Released Claims hereinabove mentioned or implied. The Buyers and the Divested Companies expressly consent that this release shall be given full force and effect in accordance with each and every express term or provision, including those (i) relating to any Seller Released Claims hereinabove mentioned or implied or (ii) relating to unknown and unsuspected Seller Released Claims (notwithstanding any state statute that expressly limits the effectiveness of a general release of unknown, unsuspected and unanticipated Seller Released Claims).
- 5.25 Seller Guarantee. Seller Guarantor hereby irrevocably, absolutely and unconditionally guarantees, as a primary obligation and not as a surety, to the Buyers the payment and performance of all obligations of the Company and the Sellers under this Agreement. This guaranty is an absolute, unconditional and continuing guaranty of payment and performance and not of collectability, irrespective of the validity, legality or enforceability of this Agreement or any Ancillary Agreement. Seller Guarantor waives promptness, diligence, presentment, demand, protest, notice of acceptance, notice of any obligations incurred and all other notices of any kind, all defenses which may be available by virtue of any valuation, stay, moratorium Law or other similar Law now or hereafter in effect, any right to require the marshalling of assets of the Buyers, any of its Affiliates or any other entity or other Person primarily or secondarily liable with respect to any of the guaranteed obligations, and all suretyship defenses generally. If any payment in respect of any of the guaranteed obligations is rescinded after receipt by the Buyers, the guaranty hereunder shall be automatically reinstated as if no such payment had ever been made. Seller Guarantor agrees that the Buyer shall not be required to prosecute collection, enforcement or other remedies against Seller or to enforce or resort to any rights or remedies pertaining thereto, before calling on Seller Guarantor for payment or performance. Seller Guarantor hereby waives any and all notice of the creation, renewal, extension or accrual of the obligations of Seller Guarantor set forth in this Agreement and notice of or proof of reliance by upon this Section 5.25 or acceptance of this Section 5.25. Seller Guarantor acknowledges that it will receive substantial direct and indirect benefits from the transactions contemplated by this Agreement and that the waivers set forth in this Section 5.25 are made knowingly in contemplation of such benefits. Seller Guarantor represents and warrants that (a) it is duly organized, validly existing and in good standing under the Laws of its jurisdiction of incorporation, (b) it has all requisite corporate power and authority to execute, deliver and perform its obligations under this Agreement and this Agreement has been duly executed and delivered by it and, assuming due authorization, execution and delivery by the other parties hereto, constitutes a valid and binding obligation of Seller Guarantor, enforceable against Seller Guarantor in accordance with its terms, except as enforcement may be limited by the General Enforceability Exceptions, and (c) the execution, delivery and performance of this Agreement does not contravene any Law to which Seller Guarantor is subject. Any agreement or arrangement with respect to, or effecting, an asset sale, distribution or liquidation of all or substantially all of Seller Guarantor's assets in one or as series of transactions shall expressly provide for the assumption of the obligations of Seller Guarantor hereunder by the counterparty(ies) to such transaction.

5.26 Wrong Pocket.

(a) From and after the Closing Date, the Company shall deliver to the Buyers, within three Business Days after receipt, any cash, checks, insurance proceeds, mail, packages,

notices and other similar communications it receives pertaining to the Business, including any accounts receivable acquired by the Buyers hereunder. The Company shall endorse in favor of the Buyers any checks or other instruments of payment payable to the Company or any of its Affiliates but acquired by the Buyers pursuant to this Agreement. The Company shall promptly provide written notice to the Buyers of any telephone calls, and shall promptly forward to the Buyers any written correspondence, e-mail or other communications, the Company or any of its Affiliates receives from any Person related to the Business, including any telephone calls, written correspondence, e-mail or other communications from any customers, distributors, suppliers and creditors of the Company or any of its Affiliates. For the avoidance of doubt, any payment received by the Company after the Closing Date in respect of accounts receivable that are Sold Assets shall be deemed to be made in trust for the benefit of the Buyers.

- (b) From and after the Closing Date, the Buyers shall deliver to the Sellers, within three Business Days after receipt, any cash, checks, insurance proceeds, mail, packages, notices and other similar communications they receive pertaining to the Excluded Assets. The Buyers shall endorse in favor of the Company any checks or other instruments of payment payable to the Buyers or any of their respective Affiliates but that are Excluded Assets pursuant to this Agreement. The Buyers shall promptly provide written notice to the Company of any telephone calls, and shall promptly forward to the Company any written correspondence, e-mail or other communications, the Buyers or any of their respective Affiliates receives from any Person related to an Excluded Asset.
- 5.27 Release of Credit Support Obligations. The Buyers shall use their respective commercially reasonable efforts to cause the Company and its Affiliates (other than the Divested Companies) to be fully and irrevocably released, as of the Closing Date or as promptly as practicable after the Closing Date, in respect of the Credit Support Obligations. If the Buyers are not able to release any Credit Support Obligation, the Company shall or shall cause its applicable Affiliate to maintain such Credit Support Obligation in full force and effect until such time as such Credit Support Obligation is released, and the Buyers shall indemnify the Company or such Affiliate against any and all payments made by the Company or any of its Affiliates (other than the Divested Companies) in accordance with such Credit Support Obligation between the Closing Date and the date the Company and its Affiliates (other than the Divested Companies), as applicable, are released from such Credit Support Obligation. Without limiting the foregoing, after the Closing Date, the Buyers shall not, and shall cause their Subsidiaries (including the Divested Companies) not to, renew, extend, amend or supplement any Contract or Liability that is covered by a Credit Support Obligation without using commercially reasonable efforts to release the Credit Support Obligation applicable to the Company or its Affiliates. Any cash or other collateral posted by the Company in respect of any Credit Support Obligation that is received by the Buyers or their respective Affiliates shall be delivered to the Company. The Company shall use its commercially reasonable efforts to cooperate with the Buyers with respect to the obligations set forth in this Section 5.27. Notwithstanding anything to the contrary herein, the provisions of this Section 5.27 shall only apply to the Credit Support Obligations set forth on Schedule 3.29 and the customs bond set forth on Schedule 5.1(f).

5.28 Certain Matters.

(a) The Company shall use its commercially reasonable efforts to cause the financial statements of Lamons Spain for the financial years 2017 and 2018 to be accepted with the Spanish commercial registry. In addition, to the extent necessary to effect the transfer of the Lamons Spain Shares contemplated by this Agreement under applicable Law, the Company shall

cause the financial statements of Lamons Spain for the financial years 2017 and 2018 to be accepted with the Spanish commercial registry prior to Closing.

(b) Following the Closing, without the consent of the Buyers, (i) neither the Company nor any of its Affiliates shall amend or modify the Master Guaranty in any manner that increases the obligations of Buyer or any of its Affiliates and (ii) neither the Company nor any of its Affiliates shall breach or otherwise take any action under the Master Guaranty that would impair or otherwise negatively impact the rights of the Buyers and its Affiliates.

ARTICLE VI CONDITIONS TO THE COMPANY'S OBLIGATIONS

The obligation of the Company to effect the Closing under this Agreement is subject to the satisfaction, at or prior to the Closing, of each of the following conditions (any and all of which may be waived by the Company in whole or in part to the extent permitted by applicable Law):

- 6.1 Representations and Warranties. The representations and warranties made by the Buyers in this Agreement shall be true and correct in all respects, as of the Closing Date (except to the extent such representations and warranties are made as of a specified date, which representations and warranties shall be true and correct in all respects as of such earlier date), except to the extent the failure of any such representation or warranty to be true and correct has not had, and would not reasonably be expected to result in, a material adverse effect on the ability of the Buyers to consummate the transactions contemplated by this Agreement.
- 6.2 Performance. The Buyers shall have performed and complied in all material respects with all covenants and agreements required by this Agreement to be so performed or complied with by the Buyers at or prior to the Closing.
- 6.3 Officer's Certificates. Each Buyer shall have delivered to the Company a certificate, dated as of the Closing Date and executed by an officer of the such Buyer, certifying to the fulfillment of the conditions specified in Sections 6.1 and 6.2.
- 6.4 Competition/Foreign Investment Laws. All approvals or clearances under any applicable any Law that prohibits, restricts or regulates competition, foreign investment, antitrust, monopolization or restraint of trade shall have been obtained and all applicable waiting periods shall have expired or been terminated.
- 6.5 Governmental Orders. At the Closing there shall not be in effect any Governmental Order restraining, enjoining or otherwise prohibiting the transactions contemplated hereby.
- 6.6 Deliveries. The Buyers shall have delivered or caused to be delivered to the Company all of the deliveries listed in Section 2.9(b).

ARTICLE VII CONDITIONS TO THE BUYERS' OBLIGATIONS

The obligation of the Buyers to effect the Closing under this Agreement is subject to the satisfaction, at or prior to the Closing, of each of the following conditions (any and all of which may be waived by the Buyers in whole or in part to the extent permitted by applicable Law):

- 7.1 Representations and Warranties. The representations and warranties made by the Company in Article III of this Agreement shall be true and correct in all respects (without giving regard to any materiality qualifiers set forth therein) as of the Closing Date as though such representations and warranties were made at such date (except to the extent such representations and warranties are made as of a specified date, which representations and warranties shall be true and correct in all respects only as of such earlier date) except where the failure of such representations and warranties to be true and correct has not had, or would not reasonably be expected to have, individually or in the aggregate, a Business Material Adverse Effect or Seller Material Adverse Effect; provided, however, that the Fundamental Representations shall be true and correct in all respects, except for de minimis inaccuracies.
- 7.2 Performance. The Company and the Sellers shall have performed and complied in all material respects with all covenants and agreements required by this Agreement to be performed or complied with by the Company and the Sellers at or prior to the Closing.
- 7.3 Officer's Certificate. The Company shall have delivered to the Buyers a certificate, dated as of the Closing Date and executed by an officer of the Company, certifying to the fulfillment of the conditions specified in Sections 7.1 and 7.2.
- 7.4 Competition/Foreign Investment Law. All approvals or clearances under any applicable any Law that prohibits, restricts or regulates competition, foreign investment, antitrust, monopolization or restraint of trade, shall have been obtained and all applicable waiting periods shall have expired or been terminated.
- 7.5 Governmental Orders. At the Closing there shall not be in effect any Governmental Order restraining, enjoining or otherwise prohibiting the transactions contemplated hereby.
 - 7.6 Required Consents. The Required Consents shall have been obtained.
- 7.7 Deliveries. The Company shall have delivered or caused to be delivered to the Buyers all of the deliveries listed in Section 2.9(a).

ARTICLE VIII TERMINATION

- 8.1 Termination. This Agreement may be terminated at any time prior to the Closing:
- (a) by the mutual written consent of the Company and the Buyers;
- (b) by the Company or the Buyers if the Closing has not occurred on or before the date that is 90 days following the Execution Date (the "End Date"), unless the failure of the Closing to occur on or prior to the End Date shall be due to the failure of such Party attempting to terminate to comply in all material respects with the agreements and covenants contained herein;
- (c) by either the Company or the Buyers if any Governmental Authority of competent jurisdiction shall have issued a Governmental Order or taken any other action restraining, enjoining or otherwise prohibiting the transactions contemplated by this Agreement and such Governmental Order or other action shall have become final and non-appealable; provided, that

the right to terminate this Agreement pursuant to this Section 8.1(c) shall not be available to a Party if such Governmental Order or other action was primarily due to the failure of such Party to perform or comply in any material respect with the agreements and covenants contained herein;

- (d) by the Buyers, at any time prior to the Closing in the event that the Company or any Seller is in breach of any representation, warranty, covenant or agreement made by the Company or such Seller in this Agreement and such breach renders the conditions set forth in Article VII incapable of being satisfied absent a written waiver of such conditions by the Buyers; provided, however, that Company shall have 10 days to cure such breach following the receipt of written notice thereof by the Buyers (which cure period shall be tolled if, during such 10-day period, the Company shall have undertaken commercially reasonable efforts to cure such breach and such breach is in the process of being cured at the end of such 10-day period, provided, that such breach must in any event be actually cured prior to earlier to occur of the 30th day following such breach or the End Date);
- (e) by the Company, at any time prior to the Closing in the event that any Buyer is in breach of any representation, warranty, covenant or agreement made by such Buyer in this Agreement and such breach renders the conditions set forth in Article VI incapable of being satisfied absent a written waiver of such conditions by the Company; provided, however, that such Buyer shall have 10 days to cure such breach following receipt of written notice thereof from the Company (which cure period shall be tolled if, during such 10-day period, such Buyer shall have undertaken commercially reasonable efforts to cure such breach and such breach is in the process of being cured at the end of such 10-day period, provided, that such breach must in any event be actually cured prior to earlier to occur of the 30th day following such breach or the End Date); or
- (f) by the Company, if (i) all of the conditions set forth in Article VII have been satisfied or waived as of the date the Closing should have been consummated pursuant to Section 2.5 (other than those conditions that by their terms are to be satisfied by actions taken at the Closing but only if such conditions would be satisfied if Closing did occur), and such conditions continue to be satisfied or waived, (ii) the Company has irrevocably notified the Buyers in writing that the Company is ready, willing and able to consummate the Closing and (iii) the Buyers fail to complete the Closing in accordance with Section 2.5 within three Business Days after the delivery of such notification.
 - 8.2 Procedure and Effect of Termination.
- (a) A Party desiring to terminate this Agreement pursuant to Section 8.1 must give written notice of such termination to the other Parties in accordance with Section 10.7, specifying the provision hereof pursuant to which such termination is effective. If this Agreement is terminated as provided herein, (i) the provisions of the Confidentiality Agreement shall continue in full force and effect, (ii) no Party shall have any liability under this Agreement to any other Party, except (A) that nothing herein shall relieve the Company from any liability for any willful and material breach of any of the representations, warranties, covenants or agreements of the Company or any of its Affiliates set forth in this Agreement prior to such termination, and (B) as contemplated by Section 5.15 (Confidentiality), Section 8.2(b), Section 10.1 (Fees and Expenses) and Section 10.13 (Non-Recourse).
- (b) If this Agreement is terminated by the Company pursuant to Section 8.1(f), then the Buyers shall pay to the Company a fee in the amount of \$6,750,000 (the "Reverse Termination Fee") in cash by wire transfer of immediately available funds to an account designated

in writing by the Company. The Reverse Termination Fee shall be paid no later than five Business Days after notice of termination of this Agreement in accordance with Section 8.1(f).

(c) Each Party acknowledges and agrees that, prior to the Closing, the Company's and its Affiliates' sole and exclusive remedy against the Buyers, the Equity Investor, the Lenders, the Debt Financing Sources and anv of their respective direct or indirect, former, current or future stockholders, controlling persons, directors, officers, employees, general or limited partners, members, managers, Affiliates, agents or assignees (each, a "Buyer Related Party") for any breach of this Agreement by the Buyers (whether at Law, in equity, in Contract, in tort or otherwise), including for any failure to effect the Closing, shall be (i) the right in this Section 8.2 to terminate this Agreement and receive payment of the Reverse Termination Fee, if applicable or (ii) if available, the right to seek specific performance in accordance with Section 10.15. Upon payment of the Reverse Termination Fee or a termination of this Agreement under circumstances in which the Reverse Termination Fee is not payable, (A) no Buyer Related Party shall have any further liability or obligation relating to or arising out of this Agreement, any contract or agreement executed in connection herewith (including the Equity Commitment Letter and the Limited Guarantee) or any of the transactions contemplated hereby or thereby (or the abandonment or termination thereof), (B) none of the Company or any of its Affiliates shall be entitled to bring or maintain any action, suit or proceeding against any Buyer Related Party arising out of or in connection with this Agreement, any contract or agreement executed in connection herewith (including the Equity Commitment Letter and the Limited Guarantee) or any of the transactions contemplated hereby or thereby (or the abandonment or termination thereof) or any matters forming the basis for such termination and (C) the Company shall cause any action, suit or proceeding pending in connection with this Agreement, any contract or agreement executed in connection herewith (including the Equity Commitment Letter and the Limited Guarantee) or any of the transactions contemplated hereby or thereby (or the abandonment or termination thereof), to the extent maintained by the Company or any of its Affiliates against any Buyer Related Party, to be dismissed with prejudice promptly, and in any event within five Business Days after the payment of the Reverse Termination Fee. The Company shall not be entitled to collect the Reverse Termination Fee on more than one occasion. Under no circumstance shall the Company be permitted or entitled to receive both a grant of specific performance and any portion of the Reverse Termination Fee.

ARTICLE IX INDEMNIFICATION

- 9.1 Indemnification by the Company. From and after the Closing and subject to this Article IX, the Company shall indemnify and hold the Buyers, their respective Affiliates (including, after the Closing, the Divested Companies) and each of their respective officers, directors, general and limited partners, managers, members, equity holders, employees, agents and representatives (collectively, the "Buyer Indemnified Persons") harmless from and against (but, with respect to Section 9.1(a), other than with respect to Losses related to any breach by the Company of any Fundamental Representation or the representations and warranties set forth in Section 3.10 (Tax Matters), solely to the extent of the Cap) any and all losses, damages, costs and reasonable expenses (including reasonable fees and expenses of attorneys) (collectively, "Losses"), that any Buyer Indemnified Person actually suffers or incurs arising out of or resulting from:
- (a) any breach, inaccuracy or misrepresentation of any representation or warranty made by the Company in Article III;

- (b) the failure to perform any covenant or agreement of the Company contained in this Agreement; or
- (c) any Excluded Liability.

Notwithstanding anything to the contrary contained in this Article IX, (i) none of the Buyer Indemnified Persons shall be entitled to recover from the Company for any Losses under Section 9.1(a) unless the total of all such Losses exceeds \$607,500 (the "Deductible"), in which event the Buyer Indemnified Persons will be entitled to indemnification only for such Losses in excess of the Deductible, (ii) the Buyer Indemnified Persons shall not be entitled to recover more than \$607,500 (the "Cap") from the Company with respect to all Losses indemnifiable pursuant to Section 9.1(a), and (iii) the Company's aggregate liability for any claims for indemnification Section 9.1(b) (when combined with the amounts described in the foregoing clause (ii) of this Section 9.1) shall not exceed the aggregate amount of net proceeds actually received by the Company and its Affiliates pursuant to this Agreement. Notwithstanding anything to the contrary contained in this Article IX, the limitations set forth in this Section 9.1 (including the Deductible and the Cap) shall not apply to any claims against the Company or its Affiliates made by any Buyer Indemnified Person for Losses (A) related to any breach of any Fundamental Representation or breach of Section 3.10 (Tax Matters), (B) arising under Section 5.5 or Section 9.1(c) or (C) based upon Fraud, and such Buyer Indemnified Person shall be entitled to recover any and all Losses suffered or incurred that arise or result therefrom from the first dollar, without limitation. Any claim that may be asserted pursuant to Section 9.1(a) that may also be asserted pursuant to Section 9.1(b) solely as a result of the Company's failure to provide notice as required pursuant to Section 5.18 shall be asserted solely pursuant to Section 9.1(a). For the avoidance of doubt, the limitations provided for in this Section 9.1 shall not apply to indemnification for any Excluded Liabilities, including without limitation, Specified Excluded Liabilities.

- 9.2 Indemnification by the Buyers. From and after the Closing and subject to this Article IX, the Buyers shall, jointly and severally, indemnify and hold the Company and its Affiliates and their respective officers, directors, employees, agents and representatives (collectively, the "Seller Indemnified Persons") harmless from and against any and all Losses that any Seller Indemnified Person actually suffers or incurs arising out of or resulting from:
- (a) any breach, inaccuracy or misrepresentation of any representation or warranty of any Buyer contained in Article IV:
- (b) the failure to perform any covenant or agreement of any Buyer (or its designee(s)) contained in this Agreement; or
 - (c) any Assumed Liability.

Any claim that may be asserted pursuant to Section 9.2(a) that may also be asserted pursuant to Section 9.2(b) solely as a result of the Buyers' failure to provide notice as required pursuant to Section 5.18 shall be asserted solely pursuant to Section 9.2(a).

9.3 Exclusive Remedy. Except for Buyer's rights under the R&W Insurance Policy, the indemnification provided in Sections 2.11 and 5.5 and this Article IX, subject to the limitations, if any, set forth herein, shall be the sole and exclusive post-Closing remedy available to any Party in connection with any Losses arising out of or resulting from this Agreement, the transactions contemplated hereby, or the Buyers' ownership or operation of the Business, whether based in

contract or tort (including negligence) or otherwise predicated on common law standards, strict liability or otherwise; provided, however, the provisions of this Section 9.3 shall not prevent or limit a cause of action at Law or in equity (a) based upon Fraud or (b) under Section 10.15 to obtain an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof, nor shall such provision prevent or limit the rights of the Parties with respect to Sections 2.6(c) or 5.4. Except with respect to Losses arising from (i) any breach, inaccuracy or misrepresentation of any representation or warranty made by the Company in Article III, (ii) Fraud or (iii) any Excluded Liability (including the Specified Excluded Liabilities), the Buyer Indemnified Persons expressly waive any and all rights and remedies against the Company or its Affiliates under the Comprehensive Environmental Response, Compensation and Liability Act, 42 U.S.C. § 9601 et seq., as amended, and other Environmental Laws in connection with any Losses arising out of or resulting from this Agreement, the transactions contemplated hereby or the ownership or operation of the Business or the Sold Assets.

9.4 Indemnification Calculations. The amount of Losses recoverable by an Indemnified Party pursuant to this Article IX and pursuant to Section 5.5 with respect to an indemnity claim shall be reduced by (a) the amount of insurance proceeds or other amounts actually recovered by such Indemnified Party with respect to the Losses to which such indemnity claim relates, (b) the amount of any Tax Benefit actually realized by the Indemnified Party in the taxable year or period in which the Losses occurred and (c) any amounts included in the calculation of the Closing Working Capital in the Final Statement. For purposes of this Agreement, a "Tax Benefit" means the net amount of all net reductions, if any, in cash Taxes paid by the Indemnified Party and its Affiliates by reason of such Loss, which reductions shall be calculated as the positive excess, if any, of (1) the Indemnified Party's liability for Taxes in the year the Loss is incurred, not taking into account (i) such Loss or (ii) an indemnification payment made under this Agreement on account of such Loss, over (2) the Indemnified Party's liability for Taxes in such tax year, taking into account (i) such Loss, (ii) any adjustment to the basis of any Sold Asset, the stock of any Divested Company or the assets of any Divested Company as a result of such Loss or the receipt of any indemnification payment, and (iii) any taxable income realized by the Indemnified Party as a result of the right to receive or the receipt of an indemnification payment made under this Agreement on account of such Loss, with such Loss treated as the last item of expense or deduction realized for such tax year. If a Tax Benefit is not recognized prior to any indemnification payment during any taxable year or period, the Indemnified Party shall thereafter make payments to the Indemnifying Party within 15 days after filing the Tax Return for the taxable year or period in which the Loss occurred that reflects the Tax Benefit (which for this purposes shall not include any payment of estimated Taxes). An Indemnified Party shall use its commercially reasonable efforts to recover under insurance policies or indemnity, contribution or other similar agreements for any Losses. If an insurance recovery is made or a third party payment is received by the Buyer, the Divested Companies or any of its Affiliates with respect to any Loss for which any such Person has been indemnified hereunder, then a refund equal to the aggregate amount of the recovery (net of all direct collection costs and expenses) shall be made promptly to the Company. Notwithstanding the foregoing or anything else in this Agreement to the contrary, no Buyer Indemnified Person shall have any obligation to seek to recover under insurance policies with respect to any Specified Excluded Liability. The Parties agree that any indemnification payments made pursuant to this Agreement shall be treated for tax purposes as an adjustment to the Purchase Price, unless otherwise required by applicable Law. For purposes of determining both (i) whether there is a breach of, or any inaccuracy or misrepresentation with respect to, any representation or warranty for which indemnification would be available under this Article IX and (ii) the amount of any Losses resulting from such breach, inaccuracy or misrepresentation, all materiality, Business Material

Adverse Effect, Seller Material Adverse Effect and similar qualifiers shall be disregarded; provided, however, that (i) no Business Material Adverse Effect qualifier shall be disregarded with respect to the representations and warranties contained in the last sentence of Section 3.9 and (ii) the term "Material" in the defined term "Material Contract" shall not be disregarded for purposes of Sections 3.14(b)-(c).

Survival. The representations and warranties and covenants and agreements (to the extent such covenants and agreements relate to the performance of obligations prior to the Closing) contained in this Agreement will survive the Closing Date until and shall terminate, expire and cease to be of any force or effect at 5:00 p.m. (Eastern time) on the date that is the 15-month anniversary of the Closing Date, except for (a) the representations and warranties set forth in Sections 3.1(a) (Organization), 3.2 (Authorization; Enforceability), 3.3 (Capital Stock of Divested Companies), 3.6 (No Conflicts or Approvals) (solely as Section 3.6 relates to the Real Property Leases held by the Asset Seller), 3.17(a) (Title to Sold Assets), 3.25 (Operating Areas), 3.26 (Affiliate Transactions), and 3.27 (No Brokers' or Other Fees) (the "Fundamental Representations"), which will survive the Closing indefinitely, and (b) the representations and warranties set forth in Section 3.10 (Tax Matters), which will survive the Closing until the date that is 60 days after the expiration of the applicable statute of limitations. All other covenants and agreements contained in this Agreement to be performed following the Closing Date will survive the Closing Date in accordance with their terms. All claims for indemnification must be asserted on or prior to the date of the termination of the respective survival periods set forth in this Section 9.5, except such claims may be pursued thereafter if written notice thereof (specifying in reasonable detail the basis for such claim) was duly given within such survival period. Notwithstanding anything in this Agreement to the contrary, the Buyer Indemnified Persons' right to assert claims for indemnification pursuant to Section 9.1(c) and the Seller Indemnified Persons' right to assert claims for indemnification pursuant to Sections 9.2(c) will, in each case, survive the Closing indefinitely. Any claim for indemnification not made by any Buyer Indemnified Persons or Seller Indemnified Persons, as applicable, on or prior to the date of termination of the applicable survival period will be irrevocably and unconditionally released and waived, whether or not a longer period would be permitted by applicable Law. Notwithstanding the foregoing, the Parties acknowledge and agree that the survival periods set forth in this Section 9.5 are applicable to the Buyer Indemnified Persons' rights to indemnification against the Company in accordance with this Agreement, but are not intended to, and shall not, modify or limit the Buyer Indemnified Persons' rights to recovery under the R&W Insurance Policy in accordance with its terms.

9.6 Notice and Opportunity to Defend. If there occurs an event which a Party asserts is an indemnifiable event pursuant to Sections 9.1 or 9.2, the Party or Parties seeking indemnification (the "Indemnified Party") shall promptly notify the other Party or Parties obligated to provide indemnification (the "Indemnifying Party"), which notice shall specify the nature and basis of such claim and the amount thereof, to the extent known. If such event involves any claim or the commencement of any action or proceeding by a third Person (a "Third Party Claim"), the Indemnified Party will give such Indemnifying Party prompt written notice (the "Claim Notice") of such claim or the commencement of such action or proceeding, which notice shall specify the nature and basis of such claim and the amount thereof, to the extent known, and shall be accompanied by copies of all relevant documentation with respect to such claim, including any summons, complaint or other pleadings that may have been served, any written demand or any other relevant document or instrument; provided, however, that the failure to provide such prompt notice will not relieve the Indemnifying Party of its obligations hereunder except to the extent such failure actually and materially prejudices the Indemnifying Party hereunder. In the case of a Third Party Claim,

except for any Third Party Claim for which the Indemnified Party is seeking coverage entirely under the R&W Insurance Policy, the Indemnifying Party shall be entitled to assume the defense thereof, with counsel selected by the Indemnifying Party. After notice from the Indemnifying Party to the Indemnified Party of such election to assume the defense thereof, the Indemnifying Party shall not be liable to the Indemnified Party for any legal expenses of other counsel or any other expenses subsequently incurred by such Indemnified Party in connection with the defense thereof. The Indemnifying Party and the Indemnified Party agree to cooperate reasonably with each other and their respective counsel in connection with the defense, negotiation or settlement of any such Third Party Claim. Notwithstanding anything else set forth in this Section 9.6 to the contrary, the Indemnified Party shall at all times have the right to participate at its own expense in the defense of such action or asserted liability. If the Indemnifying Party assumes the defense of an action, no settlement or compromise thereof may be effected (a) by the Indemnifying Party without the written consent of the Indemnified Party unless the settlement (i) involves solely money damages and all such relief is paid or satisfied in full by the Indemnifying Party and (ii) provides for the unconditional release of the Indemnified Party from all liabilities and obligations in connection with such action and such settlement imposes no obligations or restrictions on such Indemnified Party or its business or assets or (b) by the Indemnified Party without the written consent of the Indemnifying Party (which consent shall not be unnecessarily withheld, conditioned or delayed). In no event shall an Indemnifying Party be liable for any settlement effected without its written consent. The terms of this Section 9.6 shall also apply to an indemnifiable event pursuant to Section 5.5; provided, however, that in the event of any conflict between the provisions of this Section 9.6 and Section 5.5, then the provisions of Section 5.5 shall control with respect to the specific matters contemplated by that Section. Notwithstanding anything in this Agreement to the contrary, the Specified Excluded Liabilities shall be subject to the procedures set forth on Schedule 9.6.

9.7 Additional Limitations.

- (a) Except for Losses resulting from an action brought by a third party against a Buyer Indemnified Person or a Seller Indemnified Person, with respect to all indemnification claims made under Section 9.1 and Section 9.2 (except for indemnification claims made under Section 9.1(a) and Section 9.2(a)), no Buyer Indemnified Person or Seller Indemnified Person shall be entitled to indemnification under Article IX for punitive damages or for any Losses to the extent such Losses are not the probable and reasonably foreseeable result of, and which are not proximately caused by, any breach by a Party of any representation, warranty, covenant or agreement in this Agreement.
- (b) Notwithstanding anything in this Article IX to the contrary, no Buyer Indemnified Person is entitled to make any claim under any provision of this Agreement for reimbursement or indemnification for any Losses pursuant to this Article IX to the extent such Losses have been reflected in the adjustment to the Purchase Price pursuant to Sections 2.5 or 2.6. In addition, no Party shall be entitled to be compensated more than once for the same Loss.
- (c) No Party shall be entitled to recover any indemnification payment or other amounts due from the other Parties under this Agreement by retaining and setting off the amounts (whether or not such amounts are liquidated or reduced to judgment) against any amounts due or to become due from such Party under this Agreement or under any document delivered pursuant hereto or in connection herewith, including any Ancillary Agreement.

- (d) The Indemnifying Party and the Indemnified Party shall use commercially reasonable efforts to cooperate with each other with respect to resolving any claim or liability with respect to which Party is obligated to indemnify the other Party or Parties under this Agreement, including by making commercially reasonable efforts to mitigate or resolve any such claim or liability. In the event that an Indemnified Party fails to make such commercially reasonable efforts to mitigate or resolve any such claim or liability, then notwithstanding anything else to the contrary contained herein, the Indemnifying Party shall not be required to indemnify the Indemnified Party for any Losses that would reasonably be expected to have been avoided if the Indemnified Party had made such efforts.
- (e) Notwithstanding anything to the contrary elsewhere in this Agreement, the rights of the Buyers to indemnification or any other remedy under this Agreement in respect of any Losses based upon or arising out of any fact or event which has caused, or is reasonably expected to cause, any inaccuracy in or breach of any of the representations or warranties of the Company contained in this Agreement, shall not be impacted or limited as a result of the Buyers having knowledge of such fact, event, inaccuracy or breach, whether prior to or after the Closing. Notwithstanding anything to the contrary elsewhere in this Agreement, Buyers shall not be entitled to indemnification with respect to Losses or Seller Taxes to the extent such Losses or Seller Taxes are: (i) Taxes attributable to the Post-Closing Tax Period or the portion of any Straddle Period beginning after the Closing Date except to the extent resulting from any breach, inaccuracy or misrepresentation of any representation or warranty contained in Sections 3.10(f), (g), (h), (j), (j), (l) or (m); (ii) Taxes that are due to the unavailability in any Post-Closing Tax Period or the portion of any Straddle Period beginning after the Closing Date of any net operating losses or other Tax attributes from a taxable period (or portion thereof) ending on or prior to the Closing Date; (iii) Taxes resulting from transactions or actions taken after the Closing Date by the Buyers, any Divested Company or any of their respective Affiliates that are not contemplated by this Agreement and are outside the ordinary course of business; or (iv) that result from the Buyer's breach of any of the covenants contained in Sections 5.4, 5.7 and 5.8.
- 9.8 Subrogation. Nothing in this Agreement shall limit or be construed to limit the right of the Company to assert any claims, demands or rights by subrogation against any Person (other than a Buyer Indemnified Person) for any amounts paid or reimbursed in respect of Losses successfully asserted by a Buyer Indemnified Person pursuant to Sections 9.1 or 5.5.

ARTICLE X MISCELLANEOUS

- 10.1 Fees and Expenses. Except as otherwise provided in this Agreement, each Party shall be liable for and shall bear its own expenses and the expenses of its Affiliates in connection with the preparation and negotiation of this Agreement and the Ancillary Agreements and the performance and consummation of the transactions contemplated by this Agreement and the Ancillary Agreements, whether or not the Closing shall have occurred.
- 10.2 Governing Law. This Agreement, and all claims and causes of action based upon, arising out of, or related to this Agreement or the transactions contemplated hereby, shall be governed by, and construed in accordance with, the Laws of the State of Delaware, without giving effect to principles or rules of conflict of laws to the extent such principles or rules would require or permit the application of Laws of another jurisdiction. The Parties expressly elect not to be bound in any way by the United Nations Convention on contracts for the International Sale of Goods.

Notwithstanding anything to the contrary contained elsewhere herein, the Parties hereby further agree (a) that no Party will bring any Proceeding against any Lender or Debt Financing Source in any way relating to this Agreement, any Ancillary Agreement, the Debt Financing or any of the transactions contemplated hereby or thereby, including any dispute arising out of or relating in any way to the Debt Commitment Letter or any other letter or agreement related to the Debt Financing, Debt Commitment Letter or the performance thereof, in any forum other than any New York State court sitting in the borough of Manhattan, or, if, under applicable Law, exclusive jurisdiction is vested in the federal courts, the United States District Court for the Southern District of New York (and appellate courts thereof) and each Party consents to the exclusive jurisdiction of such courts (and of the appropriate appellate courts) in any such action or proceeding and waives any objection to venue laid therein, and (b) that any claims brought against any Lender or Debt Financing Source will be governed by the Requirements of Law of the State of New York, without regard to the conflict of laws provisions thereof that would cause the laws of another state to apply.

- Assets and the Business, the Buyers or their Affiliates may have received, or may receive, from the Company, the Sellers, the Divested Companies or their respective representatives certain projections and other forecasts for the Business, and certain business plan and budget information. Other than in the case of Fraud and without limiting the provisions of Article III and Article IX in any respect, the Buyers acknowledge that (a) there are uncertainties inherent in attempting to make such projections, forecasts, plans and budgets, (b) the Buyers are familiar with such uncertainties, (c) the Buyers are taking full responsibility for making their own evaluation of the adequacy and accuracy of all estimates, projections, forecasts, plans and budgets so furnished to it, and (d) neither the Buyers nor any of their Affiliates will assert any claim against the Company, the Sellers or any of its or their directors, officers, employees, Affiliates or representatives, or hold the Company, the Sellers or any such Persons liable, with respect thereto. Accordingly, other than in the case of Fraud and without limiting the provisions of Article III and Article IX in any respect, the Buyers acknowledges that the Company and the Sellers make no representation or warranty with respect to such estimates, projections, forecasts, plans or budgets.
- 10.4 Amendment. This Agreement may not be amended, modified or supplemented except upon the execution and delivery of a written agreement executed by the Parties; provided, that to the extent any amendment, modification, supplement or waiver of Section 5.21, Section 8.2(c), Section 10.2, this Section 10.4, Section 10.5, Section 10.12, Section 10.13, Section 10.14 or Section 10.16 (or any amendment, modification, supplement or waiver of any other provision of this Agreement that would modify the substance of Section 5.21, Section 8.2(c), Section 10.2, this Section 10.4, Section 10.5, Section 10.12, Section 10.13, Section 10.14 or Section 10.16 in any material respect) is sought that is adverse to the rights of the Debt Financing Sources, the prior written consent of the Debt Financing Sources shall be required before such amendment, modification, supplement or waiver is rendered effective.
- 10.5 No Assignment. Neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any Party without the prior written consent of the Buyers, in the case of any assignment by the Company or any of its subsidiaries, and the Company, in the case of any assignment by a Buyer or any of its subsidiaries; provided, that (a) either Buyer may assign its rights and obligations hereunder to any Affiliate of such Buyer without obtaining the Company's consent so long as such Buyer remains liable for the obligations of such assignee hereunder, and (b) either Buyer may collaterally assign its rights (but not its obligations) under this Agreement to the Debt Financing Sources without obtaining the Company's consent. Subject to

the foregoing sentence, this Agreement shall be binding upon and shall inure to the benefit of the Parties and their respective successors and permitted assigns. Any purported assignment in contravention of this Section 10.5 shall be void and of no force or effect. No assignment of any obligations hereunder shall relieve the Parties of any such obligations.

- 10.6 Waiver. Any of the terms or conditions of this Agreement which may be lawfully waived may be waived in writing at any time by each Party which is entitled to the benefits thereof. Any waiver of any of the provisions of this Agreement by any Party shall be binding only if set forth in an instrument in writing signed on behalf of such Party. No failure to enforce any provision of this Agreement shall be deemed to or shall constitute a waiver of any of the provisions of this Agreement shall be deemed to or shall constitute a waiver of any other provision hereof (whether or not similar), nor shall such waiver constitute a continuing waiver.
- 10.7 Notices. Any notice, demand or communication required or permitted to be given by any provision of this Agreement shall be deemed to have been sufficiently given or served for all purposes if (a) personally delivered, (b) sent by a nationally recognized overnight courier service to the recipient at the address below indicated, (c) sent by electronic mail or (d) delivered by facsimile:

If to the Buyers:

LGC US Holdings, LLC and LGC US Finco, LLC c/o First Reserve XIV Advisors, L.L.C. 600 Travis Street, Suite 6000 Houston, Texas 77002 Attn: Neil Wizel

Facsimile: (713) 821-7158 Email: nwizel@firstreserve.com

With a copy to:

Vinson & Elkins LLP 1001 Fannin Street, Suite 2500 Houston, Texas 77002 Attn: Shamus Crosby Facsimile: (713) 615-5548 Email: scrosby@velaw.com

If to the Company:

TriMas Corporation 38505 Woodward Avenue, Suite 200 Bloomfield Hills, Michigan 48304

Attn: Joshua Sherbin, SVP, General Counsel and Chief Compliance Officer

Facsimile: (888) 318-8873

Email: joshsherbin@trimascorp.com

With a copy to:

Jones Day

North Point 901 Lakeside Avenue Cleveland, OH 44114-1190 Attn: Patrick J. Leddy, Esq. Erin S. de la Mare, Esq. Facsimile: (216) 579-0212 Email: pjleddy@jonesday.com

esdelamare@jonesday.com

or to such other address, e-mail address or facsimile number as any Party may, from time to time, designate in a written notice given in like manner. Except as otherwise provided herein, any notice under this Agreement will be deemed to have been duly given (i) on the date such notice is personally delivered, delivered via e-mail or delivered by facsimile or (ii) the next succeeding Business Day after the date such notice is delivered to the overnight courier service if sent by overnight courier; provided, that, in each case, notices received after 4:00 p.m. (local time of the receiving Party) shall be deemed to have been duly given on the next Business Day.

- 10.8 Complete Agreement. This Agreement (including the Disclosure Schedules and Exhibits hereto), the Confidentiality Agreement and the Ancillary Agreements contain the entire understanding of the Parties with respect to the subject matter hereof and supersede all prior agreements and understandings, both written and oral, among the Parties with respect to the subject matter hereof. The Company and the Buyers will abstain from, and will cause their respective Affiliates to abstain from, obtaining recourse under local Law under the Ancillary Agreements. In the event of any express conflict between this Agreement and any Ancillary Agreement, this Agreement will control.
- 10.9 Counterparts. This Agreement may be executed in one or more counterparts, and counterparts may be exchanged by electronic submission, all of which shall be considered one and the same agreement and each of which shall be deemed an original.
- 10.10 Publicity. The Company and the Buyers will consult with each other and will mutually agree upon any publication or press release of any nature with respect to this Agreement or the transactions contemplated hereby and shall not issue, and shall cause their respective Affiliates not to issue, any such publication or press release prior to such consultation and agreement except as may be required by applicable Law or Governmental Order or by obligations pursuant to any listing agreement with any securities exchange or any securities exchange regulation, in which case the Party proposing to issue such publication or press release shall use commercially reasonable efforts to consult in good faith with the other Party or Parties before issuing any such publication or press release and to provide a copy thereof to the other Party or Parties prior to such issuance.
- 10.11 Severability. Any provision of this Agreement which is invalid, illegal or unenforceable in any jurisdiction shall, as to that jurisdiction, be ineffective to the extent of such invalidity, illegality or unenforceability, without affecting in any way the remaining provisions hereof in such jurisdiction or rendering that or any other provision of this Agreement invalid, illegal or unenforceable in any other jurisdiction. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the Parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the Parties as closely as possible in an acceptable

manner in order that the transactions contemplated hereby are consummated as originally contemplated to the greatest extent possible.

- 10.12 Third Parties. Nothing herein expressed or implied is intended or shall be construed to confer upon or give to any Person, other than the Parties and their respective permitted successors or assigns, any rights or remedies under or by reason of this Agreement; provided, that the Debt Financing Sources shall be express third-party beneficiaries of, and shall be entitled to rely upon, Section 5.21, Section 8.2(c), Section 10.2, Section 10.4, Section 10.5, this Section 10.12, Section 10.13, Section 10.14 or Section 10.16.
- Non-Recourse. Except as otherwise expressly provided in this Agreement or in any Ancillary Agreement, no past, present or future director, officer, employee, incorporator, member, partner, stockholder, Affiliate, agent, attorney or representative of the Company or its Affiliates shall have any Liability for any obligations or Liabilities of the Company under this Agreement or the Ancillary Agreements or for any claim based on, in respect of, or by reason of, the transactions contemplated hereby and thereby. Subject to the rights of the Buyers under the Equity Commitment Letter, under the terms thereof, and notwithstanding anything to the contrary herein, the Company agrees on behalf of itself and its Affiliates that the Buyer Related Parties shall not have any liability or obligation to the Company or any of its Affiliates (whether under contract or tort, in equity or otherwise) relating to this Agreement, any Ancillary Agreement or any of the transactions contemplated hereby or thereby (including the Debt Financing). For the avoidance of doubt and notwithstanding anything herein to the contrary, neither the Company nor any of its Affiliates nor any other person acting on their behalf shall have any rights or claims (including any action, cause of action, claim, cross-claim or third-party claim of any kind or description, whether in law or in equity, whether in contract or in tort or otherwise) against any Debt Financing Source in connection with this Agreement, any Ancillary Agreement, the Debt Financing or any transaction contemplated hereby or thereby, whether in law or in equity, in contract, in tort or otherwise. In furtherance of the foregoing, each of the Company and its Affiliates agrees not to commence (and if commenced agrees to dismiss or otherwise terminate) any action or proceeding against any Debt Financing Source in connection with this Agreement, any Ancillary Agreement, the Debt Financing or any transaction contemplated hereby or thereby.

10.14 Dispute Resolution.

- (a) Informal Dispute Resolution. Except as concerns matters covered by Section 2.6 (which shall be resolved as provided therein), the Parties shall seek to resolve any disagreement, dispute, controversy or claim arising out of or in relation to or in connection with the construction, interpretation or validity of this Agreement or the other Ancillary Agreements or the alleged breach hereof or thereof or the transactions contemplated hereby or thereby amicably and in good faith through discussions between the representatives of the Company and the Buyers. If the matter is not resolved through discussion of such individuals within 30 days, each Party agrees to consider in good faith any reasonable request by the other Party or Parties to engage in mediation or any other means of alternative dispute resolution short of arbitration. Only if the Parties fail to resolve such disagreement, dispute, controversy or claim by such means within 90 days after having begun the resolution process may the aggrieved Party or Parties seek arbitration as set forth below.
- (b) Arbitration. Except as concerns matters covered by Section 2.6 and Section 5.4(c) (which shall be resolved as provided therein), any disagreement, dispute, controversy or claim arising out of or in relation to or in connection with the construction, interpretation or validity

of this Agreement or the other Ancillary Agreements or the breach hereof or thereof or the transactions contemplated hereby or thereby, and not resolved pursuant to Section 10.14(a), shall be finally settled by binding arbitration. The arbitration shall take place in Wilmington, Delaware in accordance with the Commercial Arbitration Rules of the American Arbitration Association. The arbitration shall be brought before three arbitrators, one each appointed by the Company and the Buyer, and one selected by the two appointed arbitrators. Judgment upon the award rendered by the arbitrators may be entered in any court having jurisdiction thereof. Each Party shall absorb its own costs of arbitration and the Company, on one hand, and the Buyer, on the other hand, shall split equally any arbitrators' fees, or other administrative costs and fees.

- (c) Courts. Notwithstanding any term or provision to the contrary contained in this Section 10.14, the Parties have recourse to the federal and state courts of Delaware the purpose of obtaining any provisional injunctive remedy.
- 10.15 Specific Performance. Each of the Parties acknowledges and agrees that a breach of this Agreement would cause irreparable damage to the other Parties and that such other Parties will not have an adequate remedy at Law. Therefore, the obligations of the Parties under this Agreement, including the Asset Buyer's obligation to purchase the Sold Assets and the Equity Buyer's obligation to purchase the Shares, shall be enforceable by a decree of specific performance, and appropriate injunctive relief may be applied for and granted in connection therewith. Such remedies shall, however, be cumulative and not exclusive and shall be in addition to any other remedies which any Party may have under this Agreement or otherwise. The Parties hereby waive in any such proceeding the defense of adequacy of a remedy at Law and any requirement for the securing or posting of any bond or any other security related to such equitable relief. Notwithstanding anything in this Agreement or otherwise to the contrary, the Parties hereby acknowledge and agree that the Company shall be entitled to specific performance of the Buyers' obligation to consummate the Closing in accordance with Section 2.5 only if (i) all conditions in Article VII (other than those conditions that by their nature are to be satisfied by actions taken at the Closing; provided, that such conditions would be satisfied at the Closing if the Closing were to occur) have been satisfied or waived and remain satisfied or waived, (ii) the Debt Financing has been funded in full or will be funded in full on the Closing Date in accordance with the terms of the Debt Commitment Letter, (iii) the Buyers fail to complete the Closing by the date the Closing is required to have occurred pursuant to Section 2.8 and (iv) the Company has irrevocably confirmed to the Buyers in writing that if specific performance is granted and the Equity Financing and Debt Financing are funded, then the Closing will occur.
- 10.16 Waiver of Jury Trial. THE PARTIES HEREBY KNOWINGLY, VOLUNTARILY AND INTENTIONALLY WAIVE ANY RIGHT SUCH PARTIES MAY HAVE TO A TRIAL BY JURY WITH RESPECT TO ANY SUIT OR ACTION ARISING OUT OF THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY (INCLUDING ANY LEGAL ACTION ARISING OUT OF OR RELATING IN ANY WAY TO THE DEBT FINANCING OR THE PERFORMANCE THEREOF AGAINST ANY DEBT FINANCING SOURCE). EACH PARTY HEREBY CERTIFIES THAT NEITHER IT NOR ANY OF ITS REPRESENTATIVES HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT IT WOULD NOT SEEK TO ENFORCE THIS WAIVER OF RIGHT TO JURY TRIAL. FURTHER, EACH PARTY ACKNOWLEDGES THAT THE OTHER PARTY RELIED ON THIS WAIVER OF RIGHT TO JURY TRIAL AS A MATERIAL INDUCEMENT TO ENTER INTO THIS AGREEMENT.

10.17 Attorney-Client Privilege and Conflict Waiver. Jones Day has represented one or more of the Company, the Sellers and the Divested Companies. The Parties agree that (a) the Buyers shall not, and shall not cause the Divested Companies to, seek to have Jones Day disqualified from representing the Company or the Sellers in connection with any dispute that may arise between the Company and the Sellers, on the one hand, and the Buyers and the Divested Companies (after the Closing), on the other hand, in connection with this Agreement or the transactions contemplated hereby, and (b) in connection with any dispute that may arise between the Company or the Sellers, on the one hand, and the Buyers or the Divested Companies (after the Closing), on the other hand, the Company and the Sellers (and not the Buyers or the Divested Companies) will have the right to decide whether or not to waive the attorney-client privilege that may apply to any communications between the Divested Companies and Jones Day that occurred before the Closing to the extent such communications exclusively relate to this Agreement and the transactions contemplated hereby.

10.18 Frustration of Closing Conditions. Neither the Buyer nor the Company may rely on the failure of any condition set forth in Article VI or Article VII to be satisfied if such failure was caused by such Party's breach of, or failure to comply with, any provision of this Agreement.

[Signatures are on the following page]

IN WITNESS WHEREOF, each of the Parties has caused this Agreement to be executed by its duly authorized officer, in each case as of the Execution Date.

THE COMPANY:

TRIMAS COMPANY LLC

By: <u>/s/ Thomas Amato</u>
Name: Thomas Amato
Title: Chief Executive Officer

SELLER GUARANTOR:

Solely for purposes of and Section 5.25:

TRIMAS CORPORATION

By: <u>/s/ Thomas Amato</u>
Name: Thomas Amato
Title: Chief Executive Officer

IN WITNESS WHEREOF, each of the Parties has caused this Agreement to be executed by its duly authorized officer, in each case as of the Execution Date.

THE BUYERS:

LGC US HOLDINGS, LLC

By: <u>/s/ Neil Wizel</u> Name: Neil Wizel Title: Authorized Person

LGC US FINCO, LLC

By: <u>/s/ Neil Wizel</u> Name: Neil Wizel Title: Authorized Person

Signature Page to Asset and Stock Purchase Agreement



CONTACT:

Sherry Lauderback VP, IR & Communications (248) 631-5506 sherrylauderback@trimascorp.com

TRIMAS COMPLETES THE SALE OF ITS LAMONS DIVISION

BLOOMFIELD HILLS, Michigan, December 20, 2019 - TriMas (NASDAQ: TRS) today announced that it has successfully completed the sale of its Lamons division, a transaction entered into with an investment fund sponsored by First Reserve on November 1, 2019. Lamons was sold for approximately \$135 million in cash, subject to customary post-closing adjustments.

"The funds from this transaction, combined with our strong balance sheet and robust levels of free cash flow, afford us the opportunity to continue to advance our disciplined capital allocation strategy," said Thomas Amato, President and Chief Executive Officer of TriMas. "This strategic step also unlocks TriMas' ability to reposition our portfolio by concentrating our resources and investments in our Packaging and Aerospace segments."

TriMas will report the 2019 results of its Lamons division, which was previously reported in the Specialty Products segment, as discontinued operations beginning in the fourth quarter of 2019.

About TriMas

TriMas is a global manufacturer and provider of products for customers predominantly in the consumer products, aerospace and industrial end markets, with approximately 3,400 dedicated employees in 11 countries. We provide customers with a wide range of innovative and quality product solutions through our market-leading businesses. Our TriMas family of businesses has strong brand names in the end markets served, and operates under a common set of values and strategic priorities under the TriMas Business Model. TriMas is publicly traded on the NASDAQ under the ticker symbol "TRS," and is headquartered in Bloomfield Hills, Michigan. For more information, please visit www.trimascorp.com.

About First Reserve

First Reserve is a leading global private equity investment firm exclusively focused on energy. With over 35 years of industry insight, investment expertise and operational excellence, the Firm has cultivated an enduring network of global relationships and raised approximately \$32 billion of aggregate capital since inception. First Reserve has completed over 650 transactions (including platform investments and add-on acquisitions), creating several notable energy companies throughout the Firm's history. Its portfolio companies have operated on six continents, spanning the energy spectrum from upstream oil and gas to midstream and downstream, including resources, equipment and services, and associated infrastructure. Please visit www.firstreserve.com for further information.

Notice Regarding Forward-Looking Statements

Any "forward-looking" statements, within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934, contained herein, including those relating to the Company's business, financial condition or future results, involve risks and uncertainties with respect to, including, but not limited to: general economic and currency conditions; material and energy costs; risks and uncertainties associated with intangible assets, including goodwill or other intangible asset impairment charges; competitive factors; future trends; the Company's ability to realize its business strategies; the Company's ability to identify attractive acquisition candidates, successfully integrate acquired operations or realize the intended benefits of such acquisitions; information technology and other cyber-related risks; the performance of subcontractors and suppliers; supply constraints; market demand; intellectual property factors; litigation; government and regulatory actions, including, but not limited to, the impact of tariffs, quotas and surcharges; the Company's leverage; liabilities imposed by debt instruments; labor disputes; changes

to fiscal and tax policies; contingent liabilities relating to acquisition activities; the disruption of operations from catastrophic or extraordinary events, including natural disasters; the potential impact of Brexit; tax considerations relating to the Cequent spin-off; the Company's future prospects; and other risks that are detailed in the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 2018. These risks and uncertainties may cause actual results to differ materially from those indicated by the forward-looking statements. All forward-looking statements made herein are based on information currently available, and the Company assumes no obligation to update any forward-looking statements, except as required by law.

###

TRIMAS CORPORATION UNAUDITED PRO FORMA CONSOLIDATED FINANCIAL INFORMATION

Overview

On December 20, 2019, TriMas Corporation ("TriMas" or the "Company") completed the previously announced sale of its Lamons division (the "Lamons Division") to two wholly-owned subsidiaries of an investment fund sponsored by First Reserve, pursuant to an Asset and Stock Purchase Agreement dated as of November 1, 2019 (the "Purchase Agreement"), for a purchase price of \$135 million, subject to certain adjustments as set forth in the Purchase Agreement.

The following unaudited pro forma consolidated financial information is based on historical consolidated financial information of TriMas, which is included in the Company's Annual Report on Form 10-K for the year ended December 31, 2018 and Quarterly Report on Form 10-Q for the quarterly period ended September 30, 2019, adjusted to reflect the sale of the Lamons Division. The unaudited pro forma consolidated income statements for the nine months ended September 30, 2019 and for the years ended December 31, 2018 and 2017 and the unaudited pro forma consolidated statement of operations of the Company for the year ended December 31, 2016 have been prepared to illustrate the effects of the sale as if it occurred on January 1, 2016. The unaudited pro forma consolidated balance sheet as of September 30, 2019 has been prepared to illustrate the effects of the sale as if it occurred on September 30, 2019.

The unaudited pro forma consolidated financial information is presented for informational purposes only and is not necessarily indicative of what the Company's consolidated financial position or results of operations actually would have been had the sale been completed at the dates presented. In addition, the information presented herein does not claim to project the future financial position or operating results of the Company. The unaudited pro forma consolidated financial statements should be read in conjunction with the Company's historical financial statements, including the notes thereto, as well as the accompanying notes to the unaudited pro forma financial statements.

TriMas Corporation Pro Forma Consolidated Balance Sheet As of September 30, 2019 (Unaudited - dollars in thousands)

	Α	s Reported	Lamons Division Discontinued Operations (b)			ro Forma Ijustments		c	Pro Forma TriMas orporation Continuing Operations
Assets									
Current assets:									
Cash and cash equivalents	\$	57,940	\$	(5,290)	\$	131,700 ((c)	\$	184,350
Receivables, net		152,220		(32,080)		_			120,140
Inventories		176,300		(43,040)		_			133,260
Prepaid expenses and other current assets		8,150		(1,090)		_			7,060
Total current assets		394,610		(81,500)		131,700			444,810
Property and equipment, net		226,640		(15,580)		_			211,060
Operating lease right-of-use assets		36,750		(8,180)		_			28,570
Goodwill		332,670		_		_			332,670
Other intangibles, net		171,380		(5,760)		_			165,620
Deferred income taxes		1,100		_		_			1,100
Other assets		22,060		(200)		<u> </u>			21,860
Total assets	\$	1,185,210	\$	(111,220)	\$	131,700		\$	1,205,690
Liabilities and Shareholders' Equity									
Current liabilities:									
Current maturities, long-term debt	\$	30	\$	_	\$	_			30
Accounts payable		80,180		(18,980)		20,320 ((d)		81,520
Accrued liabilities		46,560		(3,930)					42,630
Operating lease liabilities, current portion		8,500		(3,320)		_			5,180
Total current liabilities		135,270		(26,230)		20,320			129,360
Long-term debt, net		294,410		_		_			294,410
Operating lease liabilities		28,640		(4,900)		_			23,740
Deferred income taxes		21,700		(2,230)		_			19,470
Other long-term liabilities		43,130		_		_			43,130
Total liabilities		523,150		(33,360)		20,320			510,110
Preferred stock		_		_		_			_
Common stock		450		_		_			450
Paid-in capital		796,310		_		_			796,310
Accumulated deficit		(118,250)		(76,160)		111,380 ((c) (d)		(83,030)
Accumulated other comprehensive income		(16,450)		(1,700)		_			(18,150)
Total shareholders' equity		662,060		(77,860)		111,380			695,580
Total liabilities and shareholders' equity	\$	1,185,210	\$	(111,220)	\$	131,700		\$	1,205,690

TriMas Corporation Pro Forma Consolidated Statement of Income Nine Months Ended September 30, 2019 (Unaudited - dollars in thousands, except per share amounts)

	A	s Reported (a)	amons Division Discontinued Operations (b)	 Forma TriMas Corporation Continuing Operations
Net sales	\$	697,490	\$ (144,880)	\$ 552,610
Cost of sales		(511,080)	108,040	(403,040)
Gross profit		186,410	(36,840)	149,570
Selling, general and administrative expenses		(100,760)	21,620	(79,140)
Operating profit		85,650	(15,220)	70,430
Other expense, net:				
Interest expense		(10,450)	_	(10,450)
Other income, net		1,280	(30)	1,250
Other expense, net		(9,170)	(30)	(9,200)
Income before income tax expense		76,480	(15,250)	61,230
Income tax expense		(16,260)	3,540	(12,720)
Net income	\$	60,220	\$ (11,710)	\$ 48,510
Basic earnings per share:				
Net income per share	\$	1.33		\$ 1.07
Weighted average common shares—basic		45,448,711		45,448,711
Diluted earnings per share:				
Net income per share	\$	1.32		\$ 1.06
Weighted average common shares—diluted		45,745,421		45,745,421

TriMas Corporation Pro Forma Consolidated Statement of Income Twelve Months Ended December 31, 2018 (Unaudited - dollars in thousands, except per share amounts)

	As	s Reported (a)	ī	amons Division Discontinued Operations (b)		Corma TriMas Corporation Continuing Operations
Net sales	\$	877,140	\$	(172,110)	\$	705,030
Cost of sales		(633,020)		128,110		(504,910)
Gross profit		244,120		(44,000)		200,120
Selling, general and administrative expenses		(121,800)		30,920		(90,880)
Net loss on dispositions of assets		(250)		160		(90)
Operating profit		122,070		(12,920)		109,150
Other expense, net:						
Interest expense		(13,910)		_		(13,910)
Other expense, net		(2,180)		(360)		(2,540)
Other expense, net		(16,090)		(360)		(16,450)
Income before income tax expense		105,980		(13,280)		92,700
Income tax expense		(22,680)		3,950		(18,730)
Net income	\$	83,300	\$	(9,330)	\$	73,970
Basic earnings per share:						
Net income per share	\$	1.82			\$	1.61
Weighted average common shares—basic		45,824,555				45,824,555
Diluted earnings per share:						
Net income per share	\$	1.80			\$	1.60
Weighted average common shares—diluted		46,170,464			_	46,170,464
	_				_	

TriMas Corporation Pro Forma Consolidated Statement of Income Twelve Months Ended December 31, 2017 (Unaudited - dollars in thousands, except per share amounts)

	As	s Reported (a)	amons Division Discontinued Operations (b)	 Forma TriMas Corporation Continuing Operations
Net sales	\$	817,740	\$ (161,580)	\$ 656,160
Cost of sales		(598,350)	131,470	(466,880)
Gross profit		219,390	(30,110)	189,280
Selling, general and administrative expenses		(129,140)	33,310	(95,830)
Net loss on dispositions of property and equipment		(1,080)	880	(200)
Operating profit		89,170	4,080	93,250
Other expense, net:				
Interest expense		(14,400)	10	(14,390)
Debt financing and related expenses		(6,640)	_	(6,640)
Other expense, net		(1,920)	110	(1,810)
Other expense, net		(22,960)	120	(22,840)
Income before income tax expense		66,210	4,200	70,410
Income tax expense		(35,250)	1,130	(34,120)
Net income	\$	30,960	\$ 5,330	\$ 36,290
Basic earnings per share:				
Net income per share	\$	0.68		\$ 0.79
Weighted average common shares—basic		45,682,627		45,682,627
Diluted earnings per share:				
Net income per share	\$	0.67		\$ 0.79
Weighted average common shares—diluted		45,990,252		45,990,252

TriMas Corporation Pro Forma Consolidated Statement of Operations Twelve Months Ended December 31, 2016 (Unaudited - dollars in thousands, except per share amounts)

	As	s Reported (a)	Ī	amons Division Discontinued Operations (b)	Pr	o Forma TriMas Corporation Continuing Operations
Net sales	\$	794,020	\$	(158,990)	\$	635,030
Cost of sales		(583,220)		129,300		(453,920)
Gross profit		210,800		(29,690)		181,110
Selling, general and administrative expenses		(151,960)		41,390		(110,570)
Net loss on dispositions of property and equipment		(1,870)		1,110		(760)
Impairment of goodwill and indefinite-lived intangible assets		(98,900)		_		(98,900)
Operating loss		(41,930)		12,810		(29,120)
Other expense, net:		_				
Interest expense		(13,720)		50		(13,670)
Other expense, net		(2,580)		440		(2,140)
Other expense, net		(16,300)		490		(15,810)
Loss before income tax expense		(58,230)		13,300		(44,930)
Income tax benefit		18,430		(910)		17,520
Net loss	\$	(39,800)	\$	12,390	\$	(27,410)
Basic loss per share:						
Net loss per share	\$	(0.88)			\$	(0.60)
Weighted average common shares—basic		45,407,316				45,407,316
Diluted loss per share:						
Net loss per share	\$	(0.88)			\$	(0.60)
Weighted average common shares—diluted		45,407,316				45,407,316

TRIMAS CORPORATION NOTES TO UNAUDITED PRO FORMA CONSOLIDATED FINANCIAL INFORMATION

1. Basis of Presentation

On December 20, 2019, TriMas completed the previously announced sale of its Lamons Division to two wholly-owned subsidiaries of an investment fund sponsored by First Reserve, pursuant to an Asset and Stock Purchase Agreement dated as of November 1, 2019 (the "Purchase Agreement"), for a purchase price of \$135 million, subject to certain adjustments as set forth in the Purchase Agreement.

The unaudited pro forma consolidated financial information is based on historical consolidated financial information of TriMas, which is included in the Company's Annual Report on Form 10-K for the year ended December 31, 2018 and Quarterly Report on Form 10-Q for the quarterly period ended September 30, 2019, adjusted to reflect the sale of the Lamons Division. The unaudited pro forma consolidated income statements for the nine months ended September 30, 2019 and for the years ended December 31, 2018 and 2017 and the unaudited pro forma consolidated statement of operations of the Company for the year ended December 31, 2016 have been prepared to illustrate the effects of the sale as if it occurred on January 1, 2016. The unaudited pro forma consolidated balance sheet as of September 30, 2019 has been prepared to illustrate the effects of the sale as if it occurred on September 30, 2019.

The unaudited pro forma consolidated financial information is presented for informational purposes only and is not necessarily indicative of what the Company's consolidated financial position or results of operations actually would have been had the sale been completed at the dates presented. In addition, the information presented herein does not claim to project the future financial position or operating results of the Company. The unaudited pro forma consolidated financial statements should be read in conjunction with the Company's historical financial statements, including the notes thereto, as well as the accompanying notes to the unaudited pro forma financial statements.

2. Pro Forma Adjustments

The following pro forma adjustments are included in the unaudited Pro Forma Financial Statements:

- (a) The as reported column in the unaudited pro forma consolidated financial information reflects TriMas' historical financial statements from continuing operations for the periods presented and do not reflect any adjustments related to the sale of the Lamons Division.
- (b) The information in the Lamons Division Discontinued Operations column in the unaudited pro forma consolidated financial information was derived from the Company's accounting records.
- (c) Reflects the estimated cash proceeds at closing in connection with the sale of the Lamons Division at an adjusted purchase price of \$135.0 million, less estimated transaction costs of \$3.3 million related primarily to investment banking, legal and other professional fees, which were unpaid as of December 20, 2019. Such proceeds have been shown as an addition to the recorded cash balance given the Company's intent to reinvest in its businesses in the future.

The final purchase price is subject to finalization of net working capital to be settled within 60 days of the closing date.

The pro forma cash proceeds is calculated as follows (in thousands):

Base purchase price	\$ 135,000
Plus: cash sold	5,900
Less: net working capital adjustment	 (5,900)
Adjusted purchase price	135,000
Less: transaction expenses	 (3,300)
Pro forma cash proceeds	\$ 131,700

(d) The unaudited pro forma consolidated balance sheet contains tax adjustments associated with TriMas' income tax effects from the estimated taxable gain on divestiture of the Lamons Division, based on the applicable federal U.S. statutory rate of 21% and blended state tax rate of 3.4%, net of federal benefit, resulting in an increase to current liabilities for income taxes payable of \$20.3 million.